

Nicholas H. Fonseca, Chairman Shingle Springs Band of Miwok Indians P.O. Box 1340 Shingle Springs, CA 95682-1340 Fax (530) 676-8033

Timothy Cope, President Lakes Entertainment, Inc. 130 Cheshire Lane Minnetonka, MN 55305 Fax (952) 449-9353

Dear Chairman Fonseca and Mr. Cope:

The National Indian Gaming Commission (the "NIGC") received for review and approval the First Amended and Restated Memorandum of Agreement regarding the Development and Management Agreement dated October 13, 2003 (the "Contract"), between the Shingle Springs Band of Miwok Indians and Lakes KAR-Shingle Springs, LLC ("Lakes"), as well as the Amendment, dated June 16, 2004 (collectively the "Amended Contract"). I am pleased to inform you that I have approved the Amended Contract, a class II and III gaming management contract.

The Indian Gaming Regulatory Act and the regulations of the NIGC require that the Chairman of the NIGC approve management contracts for gaming operations on Indian lands. Accordingly, you submitted the Amended Contract as required by 25 U.S.C. § 2711 and 25 C.F.R. Parts 533 and 537. We have reviewed the Amended Contract and other information submitted and have determined that the standards of 25 C.F.R. Parts 531, 533, and 537 have been met.

In approving this Amended Contract, we are relying on the parties' compliance with their commitments and applicable law. Among other things, we have received assurances that construction and operation of the casino will not start until the interchange linking U.S. 50 to the Tribe's Rancheria is operational; the parties have agreed to submit to NIGC all executed loan documents; the Manager will submit the General Manager's background investigation application, making sure that NIGC has enough time to complete its investigation prior to the General Manager assuming any management responsibilities under the Contract; and Lakes has committed to continuing to make changes to the way in which it conducts business as stated in its letter dated February 18, 2004.

This letter and my signature on the Amended Contract constitute approval of the Amended Contract. Please note that if we learn of any actions or conditions that violate the standards contained in 25 C.F.R. Parts 531, 533, 535, or 537, or learn that there has been a change in the scope of the project described in the Amended Contract, we may require modifications of, or may void, the Amended Contract, after providing the parties with an opportunity for a hearing and a subsequent appeal to the NIGC as set forth in 25 C.F.R. Part 577.

Should you have any questions concerning this approval, please call us at (202) 632-7003.

Sincerel Philip N. Hogen

Philip N. Hogen Chairman

cc: Anthony Cohen, Attorney for Tribe Fax Only (707) 546-1360

1

Thomas Foley, Attorney for Lakes Fax Only (651) 602-9976

OCT 1 4 2003

¥

# JUN 18 2004

1

# FIRST AMENDED AND RESTATED MEMORANDUM OF AGREEMENT

## REGARDING

## GAMING DEVELOPMENT

## AND

## MANAGEMENT

## AGREEMENT

### between

# SHINGLE SPRINGS BAND OF MIWOK INDIANS a federally recognized tribe

and

# LAKES KAR-SHINGLE SPRINGS, LLC a Delaware limited liability company

Dated: October 13,2003

Shingle Springs Development/Management Contract 10/13/03 revision

THIS FIRST AMENDED AND RESTATED MEMORANDUM OF AGREEMENT, is made and entered into this 13 day of October, 2003 by and between the Shingle Springs Band of Miwok Indians, a federally recognized Indian tribe (hereinafter referred to as ("the Tribe"), located in the State of California with tribal offices located at P.O. Box 1340, Shingle Springs, California 95682 and Lakes KAR-Shingle Springs, LLC, a Delaware limited liability company (hereinafter referred to as "LKAR"), whose business office is located at 130 Cheshire Lane, Minnetonka, MN 55305.

### RECITALS

A. The Tribe is a federally recognized Indian tribe eligible for the special programs and services provided by the United States to Indians because of their status as Indians and is recognized as possessing powers of self-government.

B. The United States government holds lands in the State of California in trust for the benefit of the Tribe ("Tribal Lands") over which the Tribe possesses sovereign governmental powers, and the Tribe intends to acquire other lands to be held also in trust for the Tribe by the federal government and over which the Tribe will possess sovereign governmental powers.

C. In compliance with the Indian Gaming Regulatory Act of 1988, P.L. 100-497, 25 U.S.C. 2701 <u>et seq.</u> as it may from time to time be amended, the Tribal Council of the Tribe has enacted a tribal ordinance regulating the operation of gaming activities on Tribal Lands (hereinafter referred to as the "Tribal Gaming Ordinance"), creating the Shingle Springs Tribal Gaming Commission, and authorizing Class II Gaming and Class III Gaming on its Tribal Lands subject to the provisions of the Tribal Gaming Ordinance and a Tribal-State Compact.

D. The Tribe is committed to the use of gaming activities to provide employment and improve the social, economic, education, and health needs of its members; to increase the revenues of the Tribe; and to enhance the Tribe's economic self-sufficiency and self-determination.

E. The Tribe presently lacks the resources to develop and operate a gaming facility and enterprise on its own and desires to retain the services of a developer and manager with knowledge and experience in the gaming industry to secure financing, develop, manage and operate a Class II Gaming and Class III Gaming facility and related resort facilities located on its Indian lands in accordance with the Indian Gaming Regulatory Act of 1988, as amended.

F. LKAR has represented to the Tribe that it has the managerial and financial capacity to provide and secure financing for the funds necessary to develop and construct the Facility, as defined herein, and to commence operation of the Enterprise, as defined herein; and LKAR agrees to assist the Tribe in obtaining the capital investment necessary to the development of gaming facilities, and provide the management expertise necessary to the conducting of successful tribal gaming operations.

)

G. Kean Argovitz Resorts-Shingle Springs, L.L.C. ("KARSS") entered into Development and Management Agreements with the Shingle Springs Band of Miwok Indians (the "Tribe") dated June 11, 1999 (the "Development and Management Agreements"), pursuant to which KARSS was to develop and manage a gaming facility and certain related facilities to be owned by the Tribe. Lakes Gaming, Inc. and KARSS entered into an agreement under which they agreed to form LKAR to assume the rights and obligations of KARSS in connection with the Development and Management Agreements. KARSS has assigned its rights and obligations under the Development and Management Agreements to LKAR pursuant to the terms of an Assignment and Assumption Agreement.

)

H. In connection with the development and construction of the Facility, as defined herein, and the operation and management of the Enterprise, as defined herein, LKAR and the Tribe entered into a Memorandum of Agreement Regarding Gaming Development and Management Agreement dated May 5, 2000 ("May 5, 2000 Memorandum of Agreement"). Pursuant to the May 5, 2000 Memorandum of Agreement of the Development and Management Agreements to LKAR by KARSS and LKAR's assumption of KARSS's rights and responsibilities under the Development and Management Agreements.

I. In accordance with LKAR's assumption of KARSS's rights and responsibilities under the Development and Management Agreements, the Tribe granted to LKAR the exclusive right and obligation to develop, manage, operate and maintain the Facility and Enterprise as described in the May 5, 2000 Memorandum of Agreement, which was intended by the parties thereto to supercede and replace the Development and Management Agreements.

J. The May 5, 2000 Memorandum of Agreement (and related exhibit documents) between the Tribe and LKAR has been submitted for approval to the National Indian Gaming Commission ("NIGC"); NIGC staff have reviewed the May 5, 2000 Memorandum of Agreement (and related exhibit documents) and have requested certain modifications to the documents prior to issuing NIGC approval; the Tribe and LKAR have made such modifications to the documents as they have deemed necessary; such modifications are incorporated herein and, accordingly, the Tribe and LKAR agree to enter into this First Amended and Restated Memorandum of Agreement Regarding Gaming Development and Management Agreement ("Amended Memorandum Agreement") and related Transaction Documents in connection with the Project. The parties hereto intend that this Amended Memorandum Agreement is to supercede and replace the May 5, 2000 Memorandum of Agreement.

K. This Amended Memorandum Agreement shall become effective when all the necessary approvals listed in Section 3.19 of this Amended Memorandum Agreement are received (the "Effective Date") and shall continue for a term of from the Commencement Date, or as otherwise provided in this Amended Memorandum Agreement.

L. The Tribe and LKAR desire that the preliminary Facility design and development work will be done (but not the Facility construction or Enterprise operation) so the Project may proceed prior to receipt of necessary regulatory approvals.

ţ

M. The Tribe and LKAR desire to take all lawful steps reasonably possible prior to the receipt of the necessary regulatory approvals: (i) to obtain a preliminary commitment for financing of the Facility, (ii) to select and develop the site for the Facility, (iii) to design the Facility, and (iv) to enter into contracts to construct and equip the Facility so that the Facility can be opened to the public as soon as possible after the receipt of all necessary regulatory approvals.

N. [

O. The Tribe has selected LKAR, and the LKAR has agreed, to assist the Tribe in obtaining permanent financing for the Project, subject to the terms and conditions of the Facility Loan described herein, and to furnish technical experience and expertise for the development and design of the Project, and for contracting for the construction, furnishing and equipping of the Project.

ゟイ

P. This Amended Memorandum Agreement is entered into pursuant to the Indian Gaming Regulatory Act of 1988, P.L. 100-497, 25 U.S.C. 2701 <u>et seq.</u> (the "IGRA") as that statute may be amended. All gaming conducted at the Facility will at all times comply with the IGRA, applicable tribal law and the Tribal-State Compact.

Q. Any dispute regarding this Amended Memorandum Agreement between the parties or any other Transaction Documents is to be subject to the dispute resolution and governing law provisions contained herein, as well as the Resolution of Limited Waiver attached hereto.

NOW, THEREFORE, in consideration of the hereinafter mutual promises and covenants, and for other good and valuable consideration as set forth herein, the receipt and sufficiency of which are expressly acknowledged, the Tribe and LKAR agree as follows:

### ARTICLE 1. Definitions

)

As they are used in this Amended Memorandum Agreement, the terms listed below shall have the meaning assigned to them in this Article:

1.1 "Access Land Transfer Agreement" shall mean the Access Land Transfer Agreement described in Section 2.1(b) to be executed simultaneously herewith by the parties hereto in connection with providing access to Tribal Lands, together with all amendments, substitutions and renewals thereof.

1.2 "Amended Memorandum Agreement" means the two agreements contained in this document, the Development Agreement and the Management Agreement, and all amendments hereto.

1.3 "Approved Construction Budget" means the budget prepared in the manner set forth in Section 2.2(b) in connection with the development and construction of the Facility, which has been approved by Developer and the Tribe.

1.4 "Acquired Tribal Lands" shall have the meaning set forth in Section 2.1(b).

1.5 "BIA" means the United States Department of Interior Bureau of Indian Affairs.

1.6 "Chairperson" means the Tribal Chairperson as defined by the Tribe's governing documents.

1.7 "Class II Gaming" means games as defined in 25 U.S.C. §2703 (7)(A), as such law may be amended and as defined by the National Indian Gaming Commission in 25 C.F.R. §502.3 and amendments thereto, but only to the extent such games are authorized by tribal ordinance and licensed by the Tribal Gaming Commission.

1.8 "Class III Gaming" means all gaming that is not Class I or Class II Gaming as defined in the IGRA, including, but not limited to, the forms of gaming listed as Class III games by the National Indian Gaming Commission in 25 C.F.R. §502.4 and amendments thereto, but only to the extent such gaming is allowed by the Tribal-State Compact, tribal ordinance, and licensed by the Tribal Gaming Commission.

1.9 "Commencement Date" means the first day upon which the Facility is open to the public to engage in gaming activities.

1.10 "Completion" means the completion of the Facility, or portions thereof, in substantial accordance with the Plans and Specifications, as evidenced by a completion certificate from the Architect that the Facility, or portions thereof, have been substantially completed in accordance with

the Plans and Specifications, and by the issuance of a certificate of occupancy by the tribal governmental agency having jurisdiction, and certificates of such professional designers, inspectors or consultants or opinions of counsel, as LKAR may reasonably determine to be appropriate, verifying construction and furnishing of the Facility is in compliance with all Legal Requirements.

1.11 "Construction Contract" means the contract between the Tribe and the General Contractor described in Section 2.4(b), together with all amendments, substitutions and renewals thereof.

1.12 "Costs of Construction" means all costs incurred by the Tribe or LKAR pursuant to this Amended Memorandum Agreement in the aggregate to develop, construct and complete the Facility, including, without limitation,

\$4

] but excluding Initial Costs of Operation. The final amount of costs to be included in the Costs of Construction shall be determined by mutual agreement of the parties and shall be documented in the Approved Construction Budget.

1.13 "Costs of Gaming Operations" means the total of all expenses for the operation of the Enterprise's Class II Gaming and III Gaming activities pursuant to Generally Accepted Accounting Principles ("GAAP"), including but not limited to the following:

Shingle Springs Development/Management Contract 10/13/03 revision

1.14 "Costs of Incidental Operations" means all expenses and costs pursuant to Generally Accepted Accounting Principles incurred in operating the hotel, restaurants, food and beverage service, office space, swimming pool, fitness center, childcare, kids arcade, golf course and other commercial business areas comprising the Facility in which the Enterprise conducts neither Class II Gaming nor Class III G aming, i ncluding, without limitation: 64

б४

64

1.15 "Design Agreement" means the contract between the Tribe and the Architect described in Section 2.2(a).

1.16 "Developer" means Lakes KAR-Shingle Springs, LLC, a Delaware limited liability company with its business office located at 130 Cheshire Lane, Minnetonka, MN 55305.

1.17 "Development Agreement" shall mean those provisions of this Amended Memorandum Agreement that deals with the development and construction of the Facility, as the same may be amended or modified. The Development Agreement shall continue until the earlier of either the Commencement Date or provided however, that

]

1.18 "Dominion Account" shall have the meaning described in Section 3.8(a) herein.

1.19 "Dominion Account Agreement" shall mean the Dominion Account Agreement to be executed by the Tribe in favor of the LKAR in the form agreed to by the parties thereto, together with all amendments, substitutions and renewals thereof.

Shingle Springs Development/Management Contract 10/13/03 revision

1.20 "Effective Date" means the effective date of this Amended Memorandum Agreement as determined pursuant to Section 3.19.

1.21 "Enterprise" means the business enterprise of the Tribe created to engage in Class II Gaming and Class III Gaming at the Facility, and which shall include any other lawful commercial activity allowed in or near the Facility including,<sup>1</sup>

1.22 "Enterprise Accounts" shall have the meaning described in Section 3.8 herein.

54

んチ

1.23 "Facility" means the permanent buildings, structures and improvements used by the Enterprise for its gaming and incidental operations located on the Gaming Site and all Furnishings and Equipment.

1.25 "Facility Note" means the promissory note evidencing the Facility Loan in a form to be agreed to by the parties to the Facility Loan, together with all amendments, substitutions and renewals thereof.

1.26 "Fiscal Year" means the accounting year used for the operation of the Enterprise.

1.27 "Furnishings and Equipment" shall mean all furniture, furnishings and equipment required for the operation of the Enterprise in accordance with the standards set forth in this Amended Memorandum Agreement, including, without limitation:

(i) cashier, money sorting and money counting equipment, surveillance and communication equipment, and security equipment;

(ii) slot machines, video games of chance, table games, keno equipment and other gaming equipment;

(iii) office furnishings and equipment;

(iv) specialized equipment necessary for the operation of any portion of the Enterprise for accessory purposes, including equipment for kitchens, laundries, dry cleaning,

cocktail lounges, restaurants, public rooms, commercial and parking spaces, and recreational facilities; and

(v) hotel equipment (to the extent a hotel is included in the Enterprise);

(vi) all other furnishings and equipment hereafter located and installed in or about the Facility which are used in the operation of the Enterprise in accordance with the standards set forth in this Amended Memorandum Agreement.

1.28 "Gaming Site" shall mean the Tribal Lands located within the exterior boundaries of the Tribe's reservation (whose legal description is described in Exhibit A attached hereto), which are held in trust by the United States government for the benefit of the Tribe, and upon which Class II Gaming and Class III Gaming may legally be conducted under IGRA and the Tribal-State Compact.

1.29 "General Contractor" shall mean the person or entity selected by the Tribe and approved by Developer pursuant to Section 2.4 to construct the Facility.

1.30 "Generally A ccepted A ccounting P rinciples" or "GAAP" means those principles defined by the Financial Accounting Standards Board consistently applied to the gaming industry practice.

1.31 "Governmental Authorities" means the United States federal government, the BIA, the State, the State Gaming Agency, the Tribal Council, the National Indian Gaming Commission, the Tribal G aming Commission, and any court, agency, department, commission, board, bureau or instrumentality, or any of them to the extent each has legal jurisdiction over the Class II and Class III Gaming activities, Tribal Lands, the construction and operation of the Facility and Enterprise thereon, or LKAR's performance under this Amended Memorandum Agreement.

1.32 "Gross Gaming Revenues" means the Enterprise's total revenue from Class II Gaming and Class III Gaming activities (excluding any insurance proceeds received other than business interruption insurance proceeds and insurance proceeds received to reimburse the Enterprise for any claims included, or to be included, as Costs of Gaming Operations).

1.33 "Gross Incidental Revenues" means the Enterprise's total receipts from the sale of food, beverages, souvenirs and any other goods and services supplied for non-Class II Gaming and non-Class III Gaming activities that are incidental to the operation of the Enterprise (excluding any insurance proceeds received other than business interruption insurance proceeds and insurance proceeds received to reimburse the Enterprise for any claims included, or to be included, as Costs of Incidental Operations).

Shingle Springs Development/Management Contract 10/13/03 revision

)

1.34 "Gross Total Revenues" means the total of Gross Gaming Revenues and Gross Incidental Revenues of the Enterprise.

1.35 "IGRA" means the Indian Gaming Regulatory Act of 1988, P.L. 100-497, as codified at 25 U.S.C. §2701 *et. seq.*, as such may be amended from time to time.

1.36 "Initial Costs of Operation" means all Costs of Operation advanced to the Tribe pursuant to Section 2.8 and 6.2, prior to the opening of the Facility to the public, including, but not limited to,

Ś۶

1.37 "Interim Promissory Note" means the promissory note evidencing the Transition Loan in the form agreed to by the parties hereto, together with all amendments, substitutions and renewals thereof.

1.38 "Land Acquisition Loan" means the loan or advances made to the Tribe directly by LKAR pursuant to Section 2.3(a)(ii) and evidenced by the Land Acquisition Note.

1.39 "Land Acquisition Note" means the promissory note of even date herewith evidencing the Land Acquisition Loan in the form agreed to by the parties hereto, together with all amendments, substitutions and renewals thereof.

1.40 "Legal Requirements" means any and all present and future judicial, administrative, and tribal rulings or decisions, and any and all present and future federal, state, local and tribal laws, ordinances, rules, regulations, permits, licenses and certificates, in any way applicable to the Tribe, LKAR, the Tribal Lands, t he Gaming Site, the Facility, and the Enterprise, including without limitation, the IGRA, the Tribal-State Compact, and the Tribal Gaming Ordinance.

1.41 "Lender or Other Lender" means any third party who makes the Facility Loan to the Tribe under Section 2.5 herein.

1.42 "Limited Recourse" means

]

1.43 "Management Agreement" shall mean those provisions of this Amended Memorandum Agreement that deals with the management of the Facility and the Enterprise, as the same may be amended or modified. The Effective Date of the Management Agreement shall be as set out in Section 3.19.

1.44 "Manager" means Lakes KAR-Shingle Springs, LLC, a Delaware limited liability company with its business office is located at 130 Cheshire Lane, Minnetonka, MN 55305.

1.45 "Minimum Guaranteed Monthly Payments" means the minimum monthly amount payable to the Tribe, which amount shall be determined pursuant to Section 6.3 hereof.

1.46 "National Indian Gaming Commission" or "NIGC" means the commission established pursuant to the IGRA.

1.47 "Net Gaming Revenues" means Gross Gaming Revenues less (1) amounts paid out as, or paid for, prizes; and (2) Costs of Gaming Operation (excluding management compensation as set forth in Section 6.5 herein).

1.48 "Net Incidental Revenues" means Gross Incidental Revenues less Costs of Incidental Operations.

)

1.49 "Net Total Revenues" means the sum of Net Gaming Revenues plus Net Incidental Revenues.

1.50 "NIGC Approval" means written approval by the NIGC Chairman of the Management Agreement provisions herein.

1.51 "Non-Gaming Lands Transfer Agreement" shall mean the Non-Gaming Lands Transfer Agreement described in Section 2.1(b) to be executed simultaneously herewith by the parties hereto, together with all amendments, substitutions and renewals thereof.

1.52 "Notes and Security Provisions" shall have the meaning set forth in Section 7.4(a) hereof.

1.53 "Operating Note" means the promissory note evidencing the Minimum Guaranteed Payment Advances under Section 6.3(b) herein and the Working Capital Advances under Section 6.7 herein made by Manager, substantially in the form agreed to by the parties hereto, together with all amendments, substitutions and renewals thereof.

1.54 "Plans and Specifications" means the approved plans, drawings, and specifications for the Facility pursuant to Section 2.2(b).

1.55 "Project" means the scope of the development project contemplated by this Amended Memorandum Agreement, established in the Design Agreement and approved by the parties pursuant to Section 2.2(a).

1.56 "Property" means the Tribal Lands, agreed to by the parties, upon which the Tribe will build the Facility, and which land is held by the United States in trust for the Tribe and upon which Class II Gaming and Class III Gaming may legally be conducted by the Tribe, and any other land or rights-of-way acquired for development of the Project.

1.57 "Replacement" shall have the meaning described in Section 7.4(b) herein.

1.58 "Request for Advance" means any request by the Tribe for funds to pay for Project expenses incurred in connection with either approved Costs of Construction or Initial Costs of Operation pursuant to either Sections 2.5(a) or 2.5(b).

1.59 "Resolution of Limited Waiver" refers to the limited waiver of sovereign immunity to be adopted by the Tribe in the form attached hereto as Exhibit B and evidencing all approvals required pursuant to the Tribe's governing documents and applicable law.

1.60 "Scope of Project Description Sheet" shall mean the Scope of Project Description Sheet dated July 15, 2003, which substantially describes the scope of Project currently contemplated by the Tribe and LKAR.

1.61 "Secretary" means the Secretary of the Interior of the United States, or her appropriately designated representative/agent.

1.62 "Security Agreement" shall mean the Security Agreement to be executed by the Tribe in favor of LKAR in the form a greed to by the parties thereto, together with all a mendments, substitutions and renewals thereof.

1.63 "State" means the State of California.

)

1.64 "Transaction Documents" shall have the meaning described in Section 10.12(b) herein.

1.65 "Transition Loan" means the loan or advances made to the Tribe directly by LKAR pursuant to Section 2.3(a)(i) and evidenced by the Interim Promissory Note.

1.66 "Tribal Council" means the governing body of the Tribe.

1.67 "Tribal Gaming Commission" means the Shingle Springs Tribal Gaming Commission created by Shingle Springs Tribal Gaming Ordinance, No. 96-4, as such ordinance now exists or may in the future be amended, with authority to license and regulate gaming activities on Tribal Lands and which is a subordinate governmental entity of the Tribe and is entitled to all sovereign governmental immunity of the Tribe; which ordinance was approved by the NIGC on August 5, 1996.

1.68 "Tribal Gaming Enterprise" shall mean any gaming operations conducted by the Tribe which are not a part of the Enterprise.

1.69 "Tribal Lands" means all lands presently and in the future held in trust by the United States for the Tribe and all lands within the confines of the Shingle Springs Reservation and such lands as may be hereafter added thereto.

1.70 "Tribal-State Compact" means the agreement between the Tribe and the State approved on May 16, 2000, as published in the Federal Register, concerning Class III Gaming and any amendments or other modifications thereto.

1.71 "Tribe" means the Shingle Springs Band of Miwok Indians.

1.72 "UCC Financing Statements" means UCC-1 financing statements naming Tribe as debtor and naming the Lender and LKAR as secured parties, in the form approved by the parties.

1.73 "Working Capital Advances" shall have the meaning described in Section 6.7 herein.

)

Shingle Springs Development/Management Contract 10/13/03 revision

#### **DEVELOPMENT AGREEMENT PROVISIONS**

)

## ARTICLE 2. Gaming Site Selection; Construction and Financing

2.1 Gaming Site Selection.

)

(a) The Tribal Council and LKAR have mutually agreed that the site to be used for constructing the Facility ("Gaming Site") shall be those Tribal Lands located within the exterior boundaries of the Tribe's reservation (whose legal description is described in <u>Exhibit A</u> attached hereto), which are held in trust by the United States government for the benefit of the Tribe, and upon which Class II Gaming and Class III Gaming may legally be conducted under IGRA and the Tribal-State compact.

(b) It is understood that the parties intend that LKAR will transfer its interests in certain lands held by its designees, nominees or members to the Tribe pursuant to the terms and conditions of, respectively, the parties' Access Land Transfer Agreement and their Non-Gaming Lands Transfer Agreement, each of which is to be executed simultaneously herewith. Lands transferred by LKAR to the Tribe under the parties' Access Land Transfer Agreement and their Non-Gaming Lands Transfer Agreement shall be known as "Acquired Tribal Lands".

(c) Any costs incurred by the Tribe in connection with the acquisition of the Acquired Tribal Lands described in Section 2.1(b) above may be financed in advance of the Tribe obtaining any permanent financing of the Project by advances from LKAR to the Tribe, repayable under the Land Acquisition Loan on the terms and conditions described in Sections 2.3(a)(ii), (b)(ii) and (c), and shall be evidenced by the Land Acquisition Note.

2.2 Architects, Studies, Plans and Specifications.

(a) As soon as reasonably possible after signing this Amended Memorandum Agreement, the Tribe, based upon the recommendation and subject to the approval of LKAR, shall select an architect (the "Architect") for the purpose of performing certain services in connection with the design and construction of the Facility, including site development. The Tribe's agreement with the Architect shall be in the form of a contract (the "Design Agreement") approved by LKAR and the Tribal Council. The scope of the project contemplated by this Amended Memorandum Agreement (the "Project"), shall be stated and established in the Design Agreement, and shall be subject to the mutual approval of the parties. It is contemplated the scope of the Project will be substantially as described on the Scope of Project Description Sheet, subject to such changes as may be necessary or appropriate taking into account competitive conditions, financing and other circumstances. The parties understand that market, Tribal-State Compact, governmental or other conditions may change and it may be necessary to expand or decrease the scope of the Project before construction is

commenced. The Design Agreement shall also provide for and establish appropriate design packages, each pertaining to a discrete portion or phase of the Project. The Design Agreement shall allow LKAR the right and responsibility to supervise, direct, control and administer the duties, activities and functions of the Architect and to efficiently carry out its covenants and obligations under this Amended Memorandum Agreement.

(b) The Architect shall be responsible for creating the plans and specifications for the Facility ("the Plans and Specifications") and a budget for all Costs of Construction ("Approved Construction Budget"), both of which shall be subject to the mutual approval of the Tribe and LKAR prior to the commencement of construction of the Facility. The Costs of Construction budget shall not be exceeded unless mutually agreed otherwise in writing by the Tribe and LKAR, except LKAR may in its reasonable discretion reallocate part or all of the amount budgeted with respect to any line item to another line item. The Architect shall also supervise the completion of all construction, development and related activities undertaken pursuant to the terms and conditions of the Construction Contract with the General Contractor.

(c) The Tribe, using funds advanced to it under the Facility Loan or Transition Loan, shall provide funds necessary for the design, construction and development of the Facility and Enterprise, including architectural and engineering costs. The fee for the Architect's services shall: (i) be agreed to by the Tribe and LKAR; (ii) be advanced by LKAR to the Tribe; and (iii) be repaid by the Tribe to LKAR as part of and according to the terms of the Transition Loan. Following Completion or in the event of a termination of this Amended Memorandum Agreement, it is agreed between the parties hereto that the Plans and Specifications and all other design documents shall be owned by the Tribe.

(d) The Facility shall be designed and constructed so as to adequately protect the environment and the public health and safety. The design, construction and maintenance of the Facility shall, except to the extent a particular requirement or requirements may be waived in writing by the Tribal Council, meet or exceed all reasonable minimum standards pertaining to the Tribe and national, State and local building codes, fire codes and safety and traffic requirements (but excluding planning, zoning and Gaming Site use laws, ordinances, regulations and requirements), which would be imposed on the Enterprise by existing State or federal statutes or regulations which would be applicable if the Facility were located outside of the jurisdictional boundaries. To the extent that the Tribe has adopted or may in the future adopt more stringent requirements, those requirements shall govern. Nothing in this subsection shall grant to the State or any political subdivision thereof any jurisdiction (including but not limited to, jurisdiction regarding zoning or Gaming Site use) over the Facility or Enterprise or its development, management and operation.

(e) Any costs incurred by the Tribe in connection with the activities described in this Section 2.2 may be financed in advance of the Tribe obtaining any permanent financing of the Project by advances from LKAR to the Tribe, repayable under the Transition Loan on the terms and conditions described in Section 2.3 and evidenced by the Interim Promissory Note.

2.3 Pre-Construction Advances and Terms of Land Acquisition Loan and Transition Loan.

(a)(i) LKAR agrees to make the following pre-construction advances to the Tribe: (1)

The Tribe and LKAR agree that all sums previously advanced to the Tribe by KARSS under the superceded Development and Management Agreements, and by LKAR under the superceded May 5, 2000 Memorandum of Agreement (excluding advances for acquisition of the Acquired Tribal Lands), shall constitute advances by LKAR to the Tribe hereunder, be credited to LKAR's obligations hereunder, and shall be subject to the terms of the Transition Loan herein.

(a)(ii) in accordance with the terms of the Access Land Transfer Agreement, the Non-Gaming Lands Transfer Agreement and Section 2.1 hereof, LKAR agrees to make advances to the Tribe to finance the acquisition costs associated with the Acquired Tribal Lands.

(b)(i) The total amount of funds advanced to the Tribe directly from LKAR pursuant to Section 2.3(a)(i) shall equal

コ

**\$**\$

64

( J

Principal and interest due under the Transition Loan shall be paid as provided under Section 2.3(c) below.

(b)(ii) The total amount of funds advanced to the Tribe by LKAR pursuant to Section 2.3(a)(ii) shall equal

Shingle Springs Development/Management Contract 10/13/03 revision

64

৬५

.....

P rincipal and interest due under the Land A equisition Loan shall be p aid as provided under Section 2.3(c) below.

(c) The Land Acquisition Loan and the Transition Loan shall each (i) be subject to all the terms and conditions of this Amended Memorandum Agreement; (ii) subject to the priorities set forth below,  $l_{int}$ 

Shingle Springs Development/Management Contract 10/13/03 revision

)

64

The Tribe agrees to enter into a limited, transactional waiver of sovereign immunity and consent to jurisdiction and arbitration as to LKAR and in connection with the Land Acquisition Loan and the Transition Loan, as provided in the Resolution of Limited Waiver.

}

(d) The Tribe shall retain the right to prepay each of the Land Acquisition Loan and the Transition Loan, in whole or in part, without imposition of any prepayment penalty.

(e) It is the understanding of the parties that each of the Land Acquisition Loan and the Transition Loan will be the sole responsibility of Tribe, will be a Limited Recourse obligation of the Tribe, and will not be subject to any other guarantee or obligation on the part of the Tribe except the security interests and liens described herein.

LKAR shall arrange financing for the Costs of Construction pursuant to Section

2.4 Construction.

(a)

2.5.

(b) As soon as reasonably possible after the approval of the Plans and Specifications and the budget for the Costs of Construction, the Tribe, with the assistance of the Architect, shall enter into a contract with a General Contractor pertaining to the construction of the Facility (the "Construction Contract"), subject to the approval of such contract by LKAR. The General Contractor must (i) exhibit the financial capability to complete the work, (ii) have the ability to obtain adequate payment and performance bonds and builder's risk insurance in amounts requested by LKAR and Tribe, (iii) provide an acceptable bid, as mutually agreed upon by Tribe and LKAR, (iv) be capable of meeting the construction schedule and (v) construct quality facilities. The General Contractor shall be responsible for providing, including through subcontractors, all material, equipment and labor to construct and initially equip the Facility as necessary in conformance with the Plans and Specifications, including site development.

(c) The Construction Contract shall contain such provisions for the protection of the Tribe and LKAR as the parties deem appropriate, and shall provide that construction of the Facility shall commence within thirty days of the parties receiving NIGC Approval, and any necessary Tribal Gaming Commission approvals (not to be unreasonably withheld), following and subject to the granting of all approvals necessary to commence construction and obtaining the Facility Loan; and shall also provide that the General Contractor, and all its subcontractors, shall exert its best efforts to complete construction within such time as the Tribe and LKAR agree, but which shall not exceed one

year following NIGC Approval. The Construction Contract shall provide that LKAR shall be responsible and obligated for all construction administration during the construction phases of the Project. LKAR shall act as the Tribe's designated representative and shall have full power and complete authority to act on behalf of the Tribe in connection with the Construction Contract. To the extent allowed by the Construction Contract, LKAR shall have control and charge of any persons performing work on the Project site, and shall interpret and decide on matters concerning the performance of any requirements of the Construction Contract. LKAR shall have the authority and obligation to reject work which does not conform to the Construction Contract. LKAR shall conduct inspections to determine the date or dates of substantial completion and the date of Completion. LKAR shall observe and evaluate or authorize the observation and evaluation of Project work performed, review or authorize review of applications for payment for submission to the tribe and review or authorize review and certification of the amounts due the contractors and/or the General Contractor. The General Contractor shall, at a minimum, warrant its work to be performed free of defects and unworkmanlike labor pursuant to industry standards for not less than the maximum period subsequent to Completion customary in the industry or as required by State law or the Tribal-State Compact. The General Contractor shall also be required to obtain before construction commences and maintain until Completion a policy of insurance of at least equal to the estimated Cost of Construction of the Facility naming the Tribe as an additional insured; said policy to be issued by an insurance company licensed by the State and having an AM Best rating of A7 or better. General Contractor shall also provide comprehensive general liability insurance in the amount of not less than ten million dollars (\$10,000,000), naming the Tribe as an additional insured.

}

(d) Construction change orders to the Plans and Specifications or to the Construction Contract shall require written approval of the Architect, Tribe and LKAR and a representative of the Tribal Gaming Commission if required by applicable law. It is agreed that if completion of the construction, equipping and furnishing of the Facility cannot be reasonably accomplished within the budget for Costs of Construction to be agreed upon pursuant to Section 2.2(b) above, then, Tribe, upon receiving advice from LKAR and Architect, shall promptly determine which components of the Facility shall be deleted or reduced in size, such that the total expenditures for Costs of Construction shall not exceed the approved budget.

(e) LKAR and the Tribe each reserves to itself the right, as a Cost of Construction, to inspect the Facility prior to the disbursement of each requested advance of funds, and (i) approve the progress and the workmanship of the construction; (ii) verify compliance with the Plans and Specifications; (iii) verify the percentage of the Completion as set forth in requests for advance; and (iv) satisfy itself that all work for which such advance is requested has been performed and all materials for which such advance is requested are in place or, as to stored materials, are owned by the Tribe and suitably safeguarded. Such inspection will be performed in a timely manner and not unreasonably delay the disbursement of any advance.

)

)

\$6

(f) Final acceptance of construction of the Facility shall not occur until (i) evidence of Completion has been received and approved by Tribe, (ii) a fully executed indemnity or release from liens is received from the General Contractor and all subcontractors, (iii) any other documentation reasonably requested by the Tribe, the Tribal Gaming Commission or LKAR is received from the General Contractor and (iv) all approvals have been received from all Governmental Authorities from which approvals are required.

(g) LKAR, with the assistance of the Architect, shall submit to the Tribal Council, for its review and approval, the specifications for Furnishings and Equipment . Thereafter, LKAR shall select and procure vendors for purchase by the Tribe of Furnishings and Equipment required to operate the Enterprise in conformity with such specifications. The cost of Furnishings and Equipment shall be financed through the Facility Loan. Alternatively, in the sole discretion of the Tribe, LKAR may arrange for the procurement of Furnishings and Equipment on lease terms consistent with the terms provided as to the Facility Loan.

(h) The Tribe shall keep the Facility and Gaming Site free and clear of all mechanic's and other liens resulting from the construction of the Facility, which shall at all times remain the property of the Tribe. If such lien is claimed or filed, it shall be the responsibility of the Enterprise to discharge or take reasonable steps to otherwise cause the removal of the lien within thirty days after receiving written notice of such claim. The Enterprise shall indemnify and hold the Tribe and LKAR harmless for any pre-existing conditions on the Gaming Site. Tribe shall indemnify and hold LKAR harmless from any and all liability alleged to arise from any prior agreements entered into by the Tribe with any persons or entities in connection with development of the Facility and the Enterprise, including but not limited to  $\int_{1}^{1}$  Sharp Image Gaming and Excelsior Gaming; provided, however, that in the event that Lakes Gaming, Inc. is sued in its individual capacity based upon any alleged relationship with Sharp Image prior to the formation of LKAR, and not as a member of LKAR, this indemnification and hold harmless obligation shall not apply to Lakes Gaming.

2.5 Financing Obligation and Terms of Facility Loan.

(a) Commencing on the Effective Date of this Amended Memorandum Agreement, and subject to satisfaction of each of the conditions set forth in Section 2.6(a) below, LKAR will, upon LKAR's approval of requests to advance funds ("Requests for Advance"), arrange for a Lender to loan to the Tribe under the Facility Loan or, in its discretion, LKAR may advance directly to the Tribe for the benefit of the Tribe funds for the actual Costs of Construction up to an aggregate of the Approved Construction Budget approved by the parties. Such amounts loaned or advanced for Costs of Construction shall be payable and accrue interest on terms as set forth in the Facility Note, in the form agreed to by the Tribe and the Lender or by the Tribe and LKAR, as applicable. All advances of Costs of Construction made by LKAR shall be recorded by LKAR on a schedule to be attached to the Facility Note.

Shingle Springs Development/Management Contract 10/13/03 revision

(b) Commencing on the Effective Date of this Amended Memorandum Agreement, and subject to satisfaction of each of the conditions set forth in Section 2.6(b) below, LKAR will arrange to have a Lender loan the Tribe under the Facility Loan or, in its discretion, LKAR may advance directly to the Tribe funds up to the amount of the approved budget to finance the Initial Costs of Operation. All amounts loaned or advanced for Initial Costs of Operation shall be payable and accrue interest on terms as set forth in the Facility Note. All advances of Initial Costs of Operations made by LKAR shall be recorded by LKAR on a schedule to be attached to the Facility Note.

(c) The Costs of Construction and Initial Costs of Operation shall equal

64

64

(d) The Facility Loan, or total advances if made directly by LKAR, shall (i) be subject to all the terms and conditions of this Amended Memorandum Agreement; (ii)

Shingle Springs Development/Management Contract 10/13/03 revision

The Tribe agrees to enter into a limited, transactional waiver of sovereign immunity and consent to jurisdiction and arbitration as to the holder of the Facility Note, or to LKAR to the extent LKAR makes advances directly to the Tribe, as provided in the Resolution of Limited Waiver.

64

(e) The Tribe shall retain the right to prepay the Facility Loan, or total advances if made directly by LKAR, in whole or in part, without imposition of any prepayment penalty.

(f) It is the understanding of the parties that the Facility Loan, or total advances if made directly by LKAR, will be the sole responsibility of Tribe, will be a Limited Recourse obligation of the Tribe, and will not be subject to any other guarantee or obligation on the part of the Tribe except the security interests and liens described herein.

2.6 Conditions Precedent to Facility Loan.

5

(a) The obligation of LKAR to arrange for the issuance of the Facility Loan to the Tribe, or to make direct advances to the Tribe, for the initial or any subsequent advance of Costs of Construction pursuant to Section 2.5(a) above is subject to the following conditions:

(i) The Facility Loan and related Facility Note, UCC Financing Statements or other related documentation required shall be dated and duly executed and delivered by the Tribe and shall have been approved by the BIA or National Indian Gaming Commission if required by applicable law.

(ii) This Amended Memorandum Agreement shall have become effective pursuant to Section 3.19 and LKAR shall have received an opinion of counsel for the Tribe concerning the enforceability of this Amended Memorandum Agreement and the other Transaction Documents against the Tribe and the authority of the Tribe to execute each of the same.

(iii) LKAR shall have received and approved the Plans and Specifications, the budget for the Costs of Construction, and the executed Construction Contract in accordance with Sections 2.2 and 2.4.

(iv) LKAR shall have received evidence that the Gaming Site is held in trust by the United States of America, as trustee for the Tribe.

(v) LKAR shall have received and approved evidence of the bonds and insurance required of the General Contractor pursuant to Section 2.4(b).

(vi) Tribe shall have furnished to LKAR an ALTA/ACSM survey of the Gaming Site and any other Tribal Lands on which the Enterprise will be located in a form acceptable to LKAR, which survey shall locate all property lines, existing access ways, building s etback lines and e asements a ffecting the Gaming Site and other applicable Tribal Lands identified by book and page of recording, where applicable, water, electric and sewer lines, and other physical matters, including encroachments, if any, affecting the title and use of such property. The survey shall set forth the exact legal description of the Gaming Site and any other applicable Tribal Lands used with respect to the Enterprise. The Tribe further agrees to furnish to LKAR a copy of the recorded plat, if any, applicable to the Gaming Site and any other applicable Tribal Lands. All surveys required hereunder shall contain a certificate in favor of, and in form and substance satisfactory to, LKAR.

(vii) LKAR shall have received satisfactory evidence that all permits or other authorizations, including, and without limitation, the building permit(s), required by any applicable Governmental Authority to authorize construction of the Facility have been issued and are in full force and effect. If all permits are not available prior to the closing of the Facility Loan, it shall be within LKAR's discretion to arrange for the advance by a lender of such sums under the Facility Loan for work for which all applicable permits have been received. At LKAR's option, the Tribe shall furnish LKAR reasonable evidence that all other permits required in order to construct the Facility in accordance with the Plans and Specifications, and within the Approved Construction Budget, will be available when necessary.

(viii) LKAR shall have received satisfactory evidence of the availability of adequate water, electricity, telephone, sanitary sewer, and, if applicable, storm sewer service to the Facility to be provided as part of the Costs of Construction or otherwise as agreed to by the parties.

(ix) All representations and warranties of the Tribe shall be true and correct, this Amended Memorandum Agreement shall remain in effect, and the Tribe shall not be in default under this Amended Memorandum Agreement on the date of each advance.

(x) There shall be no pending or threatened litigation, claim or dispute which, in LKAR's good faith judgment, might materially adversely affect the ability of the Tribe to timely perform its obligations under this Amended Memorandum Agreement, including, without limitation, a ruling that the Tribal--State Compact is void for purposes of the IGRA. Further, the Tribe shall not be the subject of any pending or threatened bankruptcy, insolvency, reorganization or similar proceedings which, in LKAR's good faith judgment, would materially adversely affect the security for the Facility Loan or the Tribe's ability to perform its obligations under this Amended Memorandum Agreement or the Facility Note.

(xi) LKAR shall have received satisfactory evidence that the Gaming Site is free from environmental contamination of any nature whatsoever or any other environmental condition that would require any remediation pursuant to any applicable Legal Requirement.

(b) The obligation of the LKAR to arrange for a Lender to make the initial or any subsequent advances for Initial Costs of Operation, or to make such advances directly to the Tribe, is subject to the conditions precedent set forth above in subparagraphs 2.6(a) (i), (ii), (ix), (x) and (xi).

2.7 Advances for Costs of Construction. Nothing herein contained shall obligate LKAR to arrange for a Lender to advance the Costs of Construction, or make advances directly to the Tribe, for payment of any item not included in or in an amount in excess of the Approved Construction Budget.

(a) Subject to the provisions of Section 2.7 (c) relating to retainage, LKAR shall arrange for a Lender to make advances, or make advances directly, to the Tribe for materials purchased by the Tribe and stored on or off the Gaming Site but not yet incorporated into the Facility only if the Tribe provides evidence satisfactory to LKAR that such stored materials are protected against theft and damage.

(b) Unless it otherwise agrees, LKAR shall not be required to arrange for a Lender to make advances, or make advances directly to the Tribe, for Costs of Construction under the Facility Note more often than once monthly. Advances for Costs of Construction will be made based upon the progress of construction as verified by Requests for Advance approved and certified by the Architect.

(c) Subject to the provisions of the Construction Contract, LKAR may

Shingle Springs Development/Management Contract 10/13/03 revision

(d) LKAR shall not be obligated to arrange for a Lender to make the final advance, or make the final advance directly to the Tribe, for Costs of Construction until the following conditions have been satisfied: (i) all conditions stated in this Section 2.7 and Section 2.4(f) shall have been satisfied; and (ii) LKAR shall have received the following: (1) evidence that all work requiring inspection by any Governmental Authorities having jurisdiction over the Facility has been inspected and approved by such authorities and that all other required certificates and approvals have been issued; (2) an as-built survey showing the Gaming Site, the Facility, including, without limitation, the building, parking areas (including parking spaces designated as regular, compact or handicapped spaces), walkways, driveways, access ways to public streets, signs, and any encroachments; and (3) a certificate from the Architect to the effect that the Facility (including landscaping and on-site and any off-site improvements) have been completed substantially in accordance with the Plans and Specifications and that direct connection has been made to all appropriate utility facilities.

)

2.8 Advances for Initial Costs of Operation. Advances under the Facility Note for Initial Costs of Operation will be arranged by LKAR with a Lender, or made directly by LKAR to the Tribe, upon written request by the Tribe within the budget for Initial Costs of Operation approved pursuant to Section 6.1(b) and if supported by invoices or other documentation as LKAR may reasonably require. In addition, LKAR is hereby authorized to make direct payments for Initial Costs of Operation incurred by LKAR in its role as agent for the Tribe, subject to the accounting and record keeping provisions of Section 3.4(d). LKAR shall provide the Tribe with monthly reports of all advances for Initial Costs of Operation that shall compare actual advances with the budget for Initial Costs of Operation prepared pursuant to Section 6.1(b).

2.9 <u>Title to Facility</u>. The Facility, the Enterprise and all related improvements and assets shall be the sole and exclusive property of the Tribe, subject to no liens or encumbrances except for any UCC and other liens permitted in favor of the Lender and LKAR to be granted pursuant to the provisions of this Amended Memorandum Agreement or any loan or other financing agreements between the Lender or LKAR and the Tribe entered into after the date the parties execute this Amended Memorandum Agreement.

2.10 <u>No Liens</u>. During the term of this Amended Memorandum Agreement, neither the Tribe nor LKAR shall act in any way whatsoever, either directly or indirectly, to cause any other party to lease or to become a lienholder of the Gaming Site, the Facility, the Enterprise, or any related assets except as expressly agreed to by the parties or permitted herein.

2.11 <u>Limited Waiver of Sovereign Immunity</u>. By this Amended Memorandum Agreement, the Tribe does not waive, limit, or modify its sovereign immunity from unconsented suit except as provided in the Resolution of Limited Waiver as simultaneously executed, delivered and attached hereto as <u>Exhibit B</u>. The Tribe understands that its agreement to adopt an enforceable Resolution of

Limited Waiver is a material inducement to the LKAR's execution of this Amended Memorandum Agreement and is a condition precedent to any of the respective obligations of the parties under this Amended Memorandum Agreement. The Tribe further agrees that it will not amend or alter or in any way lessen the rights of the Lender or LKAR as set forth in the Resolution of Limited Waiver, which is attached hereto as <u>Exhibit B</u> and incorporated here by reference. This Section 2.11 shall survive the termination of this Amended Memorandum Agreement, regardless of the reason for the termination.

2.12 <u>Exclusivity</u>. During the term of this Amended Memorandum Agreement, the Tribe shall have an exclusive relationship with LKAR regarding the development of the Facility and the Enterprise; provided however, that LKAR acknowledges that it is aware of the Tribe's pre-existing relationship with Sharp Image and waives the application of this section and Section 2.6(a)(x) with respect to that relationship to the extent the Tribe may have existing enforceable legal obligations to Sharp Image but only on the following conditions: (i)

64

<sup>1</sup>/Subject to all Legal requirements, the Tribe also agrees that, during the term of this Amended Memorandum Agreement, the tribe may pursue other commercial, gaming and economic development opportunities on Tribal Lands; provided it

Shingle Springs Development/Management Contract 10/13/03 revision

## 2.13 Intentionally Omitted.

2.14 <u>Term of Development Agreement</u>. Unless sooner terminated as provided in this Amended Memorandum Agreement, the term of the Development Agreement shall run until the earlier of either (i) the Commencement Date; or (ii) provided however, that the Notes and Security Provisions shall continue until all amounts owing to LKAR with respect thereto have been paid in full.

64

64

2.15 <u>Tribal Representatives</u>. The Tribe hereby acknowledges and agrees that to the extent any authorization, consent or other approval of the Tribe or the Tribal Council is required under this Amended Memorandum Agreement or any related Transaction Documents and the Tribe shall provide to LKAR a tribal resolution naming any individual or individuals authorized to represent the Tribe and the Tribal Council for purposes or for the purpose of the operation and performance of this Amended Memorandum Agreement and related Transaction Documents, then LKAR shall be entitled to rely on all decisions, authorizations, consents, and approvals provided by such individual or individuals, as applicable, until such time as the Tribe shall deliver to LKAR an additional tribal resolution revoking or otherwise modifying such authority.

Shingle Springs Development/Management Contract 10/13/03 revision

#### MANAGEMENT AGREEMENT PROVISIONS

)

## ARTICLE 3. Authority and Duty of LKAR

Appointment as Agent. Subject to the terms and conditions of this Amended 3.1 Memorandum Agreement, the Tribe hereby appoints LKAR to act as the exclusive agent for the Tribe for all non-governmental matters related to the Facility and the Enterprise during the term of the Management Agreement. LKAR's agency responsibilities shall include, among other things, maintenance and improvement of the Facility, management and operation of the Enterprise's Class II Gaming and Class III Gaming activities within the Facility, and all other revenue producing activities that are conducted by the Enterprise, such as the sale of food and beverages in the Facility. LKAR accepts such appointment as the Tribe's exclusive agent for the term of this Management Agreement. Subject to the provisions of this Amended Memorandum Agreement and specifically the restrictions in this Article 3 and the budget provisions of Article 6 hereof, LKAR shall have, and the Tribe does hereby grant to LKAR, the power and authority as agent for the Tribe, to exercise the nongovernmental rights of the Tribe under and to execute, modify, or amend any contracts associated with the operations of the Facility and Enterprise, including, without limitation, purchase orders, equipment and retail leases, contracts for services, including utilities, and maintenance and repair services, relating to the operation of the Facility and the Enterprise (except for this Amended Memorandum Agreement and contracts or compacts between the Tribe and the State or its political subdivisions, which shall remain the sole authority of the Tribe). The duties and authorities of LKAR shall be subject in all events to receipt of all necessary licenses, consents or approvals from the Tribal Gaming Commission.

3.2 Limitations. LKAR shall have no authority to waive or impair the Tribe's sovereign immunity or any other attribute of its sovereign governmental powers. Except as stated herein, LKAR shall have no authority as the Tribe's agent under this Amended Memorandum Agreement without the prior written approval of the Tribe (not to be unreasonably withheld): (a) to incur costs which are materially in excess of the expenditures to be agreed upon in the operating budget or capital expenditure budget to be developed pursuant to Section 6.1(c) herein; (b) to sell, encumber or otherwise dispose of any personal property or equipment located in the Facility, except for inventory sold in the regular course of business and other items which must be replaced due to age, obsolescence, or wear and tear; (c) to purchase any goods or services from LKAR or any of LKAR's affiliated companies as a Costs of Gaming Operations, Costs of Incidental Operations or Costs of Construction unless such arrangement is specifically approved in writing by the Tribal Council. Except as specifically authorized in this Article 3, LKAR shall not hold itself out to any third party as the agent or representative of the Tribe.

Shingle Springs Development/Management Contract 10/13/03 revision

## 3.3 LKAR's Authority and Responsibility.

(a) LKAR shall conduct and direct all business and affairs in connection with the day-to-day operation, management and maintenance of the Enterprise and the Facility, including the establishment of operating days and hours, in accordance with the requirements of this Amended Memorandum Agreement. It is the parties' intention that the Enterprise be open 24 hours daily, seven days a week. LKAR is hereby granted the necessary power and authority to act, through the Enterprise's general manager, in order to fulfill all of its responsibilities under this Amended Memorandum Agreement. Nothing herein grants or is intended to grant LKAR any title or ownership interest to the Facility or to the Enterprise. LKAR hereby accepts such retention and engagement. The Tribe shall have the sole proprietary interest in and ultimate responsibility for the conduct of all Class II Gaming and Class III Gaming conducted by the Enterprise, subject to the rights and responsibilities of LKAR under this Amended Memorandum Agreement.

(b) In managing, o perating, m aintaining and r epairing the Enterprise and the Facility, LKAR's duties shall include, without limitation, the following: (i) using reasonable measures for the orderly physical administration, management, and operation of the Enterprise and the Facility, including without limitation cleaning, painting, decorating, plumbing, carpeting, grounds care and such other maintenance and repair work as is reasonably necessary; (ii) complying with all duly enacted statutes, regulations and ordinances of the Tribe; and (iii) complying with all applicable provisions of the Internal Revenue Code including, but not limited to, the prompt filing of any cash transaction reports and W-2G reports that may be required by the Internal Revenue Service of the United States or under the Tribal-State Compact.

#### 3.4 <u>Compliance with Laws</u>.

(a) LKAR shall assist the Tribe in compliance with all terms and conditions of the Tribal-State Compact, the Tribal Gaming Ordinance, IGRA and any gaming regulations (collectively, the "Governing Laws"), the violation of which would materially impair the conduct of gaming permitted to be conducted under IGRA by the Enterprise. Without limiting the foregoing, LKAR, as agent for the Tribe, shall supply the NIGC with all information necessary to comply with the National Environmental Policy Act, as it may be amended from time to time, and comply with the NIGC's regulations relating thereto. LKAR shall ensure compliance with requirements c oncerning the reporting and withholding of taxes with respect to the winnings from gaming operations pursuant to this Amended Memorandum Agreement. The Tribe agrees to cooperate with LKAR and to aid LKAR in ensuring compliance with the foregoing laws, regulations and requirements.

(b) The parties shall use their best efforts to obtain all necessary approvals of Governmental Authorities of this Amended Memorandum Agreement.

ł

)

3.5 <u>Security</u>. LKAR shall provide for appropriate security for the operation of the Enterprise. All aspects of Facility security shall be the responsibility of LKAR. Upon agreement of the Tribe and LKAR, any security officer may be bonded and insured in an amount commensurate with his or her enforcement duties and obligations. The cost of any charge for security and increased public safety services will be a Costs of Gaming Operations or Costs of Incidental Operations, as appropriate.

## 3.6 Accounting, Financial Records, and Audits.

(a) LKAR shall maintain full and accurate records and books of account for operations of gaming activities and related incidental operations managed hereunder. Such records shall be maintained at the Enterprise's general manager's office located within the Facility and shall be made available for immediate inspection and verification at all times. Inspection or verification by the Governmental Authorities shall be coordinated through the Tribal Gaming Commission.

(b) At least three months prior to the Commencement Date, and subject to the approval of the Tribal Council and the Tribal Gaming Commission, which approvals shall not be unreasonably withheld and which shall occur prior to the Commencement Date, LKAR shall establish and maintain such approved accounting systems and procedures that shall: (i) include procedures for internal accounting controls; (ii) permit the preparation of financial statements in accordance with generally accepted accounting principles; (iii) be susceptible to audit; (iv) allow the Enterprise, the Tribe and NIGC to calculate the annual fee under 25 CFR §514.1; (v) permit the calculation of LKAR's compensation under Section 6.5(b) herein; and (vi) provide for the allocation of operating expenses or overhead expenses among the Tribe, the Enterprise and LKAR, or any other user of shared facilities or services. Supporting records and the agreed upon accounting system shall be sufficiently detailed to permit the calculation and payment of any fee or contribution computations required under IGRA, the Tribal-State Compact and other applicable laws or regulations.

(c) Net Gaming Revenues, Net Incidental Revenues, and Net Total Revenues will be calculated by LKAR for purposes of distribution monthly in accordance with Section 6.5 and copies of such calculations shall be promptly supplied to the Tribal Council as required by Section 6.4 herein.

(d) All records shall be maintained so as to permit the preparation of financial statements in accordance with generally accepted accounting principles consistently applied and in accordance with procedures to be mutually agreed upon by the parties. LKAR shall, as a Costs of Gaming Operations, furnish to the Tribe and the Tribal Gaming Commission, monthly financial reports in accordance with Section 6.4. Such reports shall provide reasonable detail as requested by the Tribe and the Tribal Gaming Commission with respect to revenues and expenses of each profit center of the Enterprise. In addition, all gaming operations conducted within the Facility shall be

)

)

∖≮

subject to special outside annual audits, which the Tribal Gaming Commission may cause to be conducted, and all contracts or subcontracts for supplies, services or concessions for a contract amount in excess of \$25,000 annually relating to gaming activities within Facility shall be subject to audits which the Tribal Gaming Commission may cause to conducted by an independent certified public accountant selected and approved by the Tribal Gaming Commission. The cost of such audits and audit reports (including the annual audit under Section 6.6) shall constitute Costs of Gaming Operations. LKAR, shall make any reports or presentations to the Tribal Council as are requested by the Tribe.

3.7 <u>Cash Monitoring</u>. As a Cost of Construction, LKAR shall install a video surveillance system and computerized systems for monitoring the Gross Gaming Revenues on a daily basis. LKAR through the enterprise's general manager will promulgate, and all parties and their respective employees, agents, and representatives will obey operational policies consistent with the Gaming Control Ordinance respecting the handling of cash, security systems, and access to cash cage, counting rooms, and other places where cash is kept and handled. The Tribe and the Tribal Gaming Commission and their authorized representatives shall have the right to monitor and investigate systems for cash management implemented by the Enterprise's general manager in order to prevent any skimming of receipts or losses of the proceeds and to verify daily Gross Gaming Revenues, Gross Incidental Revenues, and Gross Total Revenues.

### 3.8 Bank Accounts, Reserve Funds and Permitted Investments.

(a) On or prior to the Commencement Date, the Tribe and LKAR shall execute the Dominion Account Agreement and create the dominion Account ("Dominion Account") described therein. Gross Gaming Revenues and Gross Incidental Revenues shall be deposited daily into the Dominion Account, which shall be subject to the lien of the Dominion Account Agreement and established at a commercial bank, of the Tribe's choice, organized under the laws of the United States of America or any state thereof provided such bank is a member of the Federal Deposit Insurance Corporation and has combined capital, undivided profits and surplus of a t least \$ 500,000,000. LKAR, through the Enterprise's general manager, shall also establish other segregated bank accounts with the approval of the Tribe for the operation of the Enterprise (the "Enterprise Accounts"), which accounts must indicate the custodial nature of the accounts.

Shingle Springs Development/Management Contract 10/13/03 revision

(b) LKAR agrees that subject to the terms of Sections 6.3(b) and 7.4(a) hereof and the Dominion Account Agreement, LKAR shall make or permit timely transfers from the Dominion Account to the Enterprise Accounts of all funds needed to pay (i) Costs of Gaming Operations; (ii) Costs of Incidental Operations; (iii) required debt service on the Land Acquisition Loan, the Transition Loan and the Facility Loan, as well as any other third party loans to which LKAR has consented in writing pursuant to the terms of this Amended Memorandum Agreement or other agreement; (iv) the Minimum Guaranteed Monthly Payment; (v) Minimum Guaranty Payment Advances; and (vi) disbursements required pursuant to Section 6.5 hereof. Upon the termination of this Amended Memorandum Agreement and so long as: (a) any amounts remain owing to LKAR hereunder or with respect to any related Transaction Documents, and (b) the Enterprise shall continue in operation pursuant to the terms of Section 7.4 hereof, then LKAR shall continue to permit transfers from the Dominion Account to the Enterprise Accounts for payment of the amounts described above, including a ny Costs of Gaming Operations or Costs of Incidental Operations (but specifically excluding any such costs otherwise payable to the Tribe or any political subdivision or other affiliate of the Tribe which is not a reasonable gaming license fees, and any costs or expense which is not associated with the provision of reasonable supplies and/or services provided by the Tribe to the Enterprise).

## (c)

2

## 3.9 Enforcement of Rights.

(a) During the term of this Amended Memorandum Agreement, except as otherwise provided in Section 3.9 (b), the Tribe and LKAR shall mutually agree with respect to the handling of the defense, prosecution or settlement of civil disputes with third parties relating to gaming and other management activities conducted or contracts executed by LKAR, as agent for the Tribe. The parties will assist and cooperate with each other with respect to such third-party claims and disputes. All uninsured liabilities incurred or expenses incurred by the Tribe and LKAR or any of the employees, officers or directors of any party in defending such claims by third parties or prosecuting claims against third parties shall be considered either Costs of Gaming Operation or Costs of Incidental Operations, depending upon the circumstances and nature of the claim, except with respect to claims and liabilities resulting from criminal misconduct, which shall be governed by Article 8 herein. 64

3.10 <u>Fire and Safety Services</u>. LKAR shall be responsible for obtaining adequate coverage for fire and safety services and may, with the consent of the Tribe, which consent shall not be unreasonably withheld, have such services provided on a contractual basis by the local Fire and Police Departments. The costs of any fire and safety protection services shall be appropriately allocated between Costs of Gaming Operation and Costs of Incidental Operations, and, if provided by a Department of the Tribe, shall not exceed the actual cost to the Tribe of providing such services.

3.11 <u>Timely Payment of Costs of Gaming Operations and Costs of Incidental Operations</u>. LKAR shall be responsible for paying Costs of Gaming Operation and Costs of Incidental Operations from the bank account(s) established pursuant to Section 3.8 so as to avoid any late-payment penalties, except those incurred as a result of good faith payment disputes.

3.12 Acquisition of Gaming and Other Equipment.

(a) All gaming equipment shall, subject to the approval of the Tribe (not to be unreasonably withheld), be acquired by the Enterprise from Tribal Gaming Commission licensed distributors and manufacturers.

(b) All acquisitions of new equipment after the public opening of the Facility shall, subject to the approval of the Tribe, which approval shall not be unreasonably withheld, Junless otherwise agreed by the Tribe.

64

3.13 <u>Hours of Operation</u>. Unless otherwise agreed by the parties, the Facility and Enterprise shall be operated for the maximum days per week and hours per day, subject to any restrictions in the IGRA and the Tribal-State Compact.

3.14 <u>Access to Operations</u>. LKAR, through the Enterprise's general manager, shall provide immediate access by appropriate officials of the Tribal Gaming Commission and the Tribe's designated representative to the gaming operation, including all books and records in addition to those listed in the access requirements set forth in Sections 3.6 and 3.7.

3.15 <u>Increased Public Safety Services</u>. The parties agree that increased actual costs of law enforcement and police protection services required as a result of the Class II Gaming and Class III Gaming in the Facility shall be paid as Costs of Gaming Operation.
3.16 <u>Advertising</u>. LKAR, through the Enterprise's general manager, shall contract for and place advertising, subject to prior approval of the general concepts of the advertising by the Tribe. Advertising costs will be included in the operating budgets prepared in accordance with Article 6.

3.17 <u>Certain Meetings</u>. The parties a gree that, to facilitate oversight of the activities conducted pursuant to this Management Agreement and to maintain communication generally between the individuals who will be involved in supervising those activities, the Tribal Council or its designated representative and the LKAR's principal individuals will meet at least monthly to review operations of the Facility and Enterprise and any current issues pertaining thereto.

3.18 <u>Maintenance</u>. LKAR will cause the Facility to be repaired and maintained and operated in a clean, good and orderly condition. Repairs and maintenance will be paid as Costs of Gaming Operation if related to the gaming operations of the Enterprise, or as Costs of Incidental Operation if related to the other operations of the Enterprise.

3.19 <u>Term</u>. Notwithstanding the date of signature of the parties hereto, this Amended Memorandum Agreement shall become effective upon the last of the following events to occur (the "Effective Date"): (i) written approval of the Tribal Gaming Ordinance and this Amended Memorandum Agreement by the Chairman of the National Indian Gaming Commission; (ii) approval by the Secretary and publication in the federal register of the Tribal-State Compact; or (iii) issuance by the Tribal Gaming Commission to LKAR of all applicable license(s) required by IGRA, the Tribal Gaming Ordinance or the Tribal-State Compact. The Commencement Date shall be the first day upon which the Facility is open to the public to engage in Class II Gaming and Class III Gaming activities. Unless sooner terminated as provided in this Amended Memorandum Agreement, this Amended Memorandum Agreement shall continue for a period of from the Commencement Date  $\sqrt{2}$ 

64

, provided that the Tribe shall have the right to terminate this Amended Memorandum Agreement after

Shingle Springs Development/Management Contract 10/13/03 revision

## ARTICLE 4. <u>Personnel Matters</u>

64

.7

Ē

Employees. All employees involved with operation of the Enterprise's Class II 4.1 Gaming and Class III Gaming activities and related activities throughout the Facility subject to management under this Management Agreement shall be employees of the Tribe. Subject to the applicable requirements in the Tribal-State Compact, the employment relationship shall be governed by Tribe substantive law, and any applicable federal law, subject to the Tribe's reasonable Indian preference policies, and all matters will be subject to dispute resolution procedures in the manner described in this Amended Memorandum Agreement. LKAR, through the Enterprise's general manager, shall be solely responsible for the hiring, training, promoting, and firing of all such employees (except for the general manager as agreed to by the Tribe and LKAR, whose employment, advancement and termination shall be subject to approval of the Tribe, such approval not to be unreasonably withheld). LKAR, through the Enterprise's general manager, shall develop a policy and procedure in conjunction with the Tribe, to implement an executive development program for employees who are members of the Tribe whereby Tribe members will be prepared through training and education to assume key management positions within the gaming and non-gaming operations of the Enterprise. All salaries, wages, employee insurance, worker compensation premiums, employment taxes, government exactions of any kind related to employment, benefits, and overhead related to the hiring, supervising, and discharge of employees, will be Costs of Gaming Operations or Costs of Incidental Operations, as appropriate.

4.2 <u>Enterprise Employee Policies.</u> LKAR, through the Enterprise's general manager, shall prepare a draft of personnel policies and procedures (the "Enterprise Employee Policies"), including a job classification system with salary levels and scales, which policies and procedures shall be subject to approval by the Tribal Council. The Enterprise Employee Policies shall include a grievance procedure in order to establish fair and uniform standards for the Enterprise employees, which will include procedures for the resolution of disputes between the Enterprise and Enterprise employees. At a minimum, the Enterprise Employee Policies shall provide for an employee grievance process which provides the following:

A written "Board of Review" process will be created by the Enterprise's general manager to provide Enterprise employees with a procedure for bringing work related issues to the attention of Enterprise management so they may be promptly and permanently resolved in a fair and equitable manner. The Board of Review process will be available to all Enterprise

Shingle Springs Development/Management Contract 10/13/03 revision

)

employees except: (1) employees at the director level and above, and (2) employees discharged for actions involving violations of tribal gaming regulations or law, or federal, state, or local law. Enterprise employees will be eligible to use the Board of Review process if they have: (a) completed thirty (30) work shifts, and progressed through the chain of command in their home department in the Enterprise, (b) brought the work related issue to the attention of the Enterprise's human resources department, and (c) completed a Board of Review hearing request form within the allotted time frame. The Enterprise's human resources department shall be responsible for selecting hearing panel members as outlined in written procedures to be adopted. The Board of Review will be empowered to make the full range of decisions available and appropriate (i.e. providing back pay and an apology to the employee, or upholding the employee's discharge). The Board of Review's decision on the work related issue will final and binding, and there will be no appeal beyond the Board of Review.

LKAR, through the Enterprise's general manager, shall be responsible for administering the Enterprise Employee Policies. Any revisions to the Enterprise Employee Policies shall not be effective unless they are approved by the Tribal Council. All such actions shall comply with applicable tribal law, subject to the applicable requirements in the Tribal-State Compact.

4.3 <u>Employee Background Checks.</u> A background investigation shall be conducted by the Tribal Gaming Commission in compliance with all Legal Requirements, to the extent applicable, on each applicant for employment as soon as reasonably practicable. No individual whose prior activities, criminal record, if any, or reputation, habits and associations are known to pose a threat to the public interest, the effective regulation of Class II Gaming or Class III Gaming, or to the gaming licenses of LKAR or the Enterprise, or to create or enhance the dangers of unsuitable, unfair, or illegal practices and methods and activities in the conduct of Enterprise gaming activities, shall knowingly be employed by Enterprise or the Tribe. The background investigation procedures employed by the Tribal Gaming Commission shall be formulated in consultation with LKAR and shall satisfy all regulatory requirements independently applicable to LKAR; provided, however, that this provision shall not be deemed to limit or impair the exclusive authority of the Tribal Gaming Commission pursuant to the Tribal Gaming Ordinance or the exercise of its discretion thereunder. Any cost associated with obtaining such background investigations shall constitute an Costs of Gaming Operations.

4.4 <u>Indian Preference</u>. LKAR, through the Enterprise's general manager, shall adhere in regard to recruitment, employment, reduction in force, promotion, training and related employment actions to a publicly announced policy and practice of Indian Preference as approved by the Tribe.

#### 4.5 Conflict of Interest.

(a) LKAR covenants that it will not unduly interfere with, or attempt to influence the internal affairs or government decisions of the Tribe for its gain or advantage.

(b) LKAR hereby certifies that no payments have been made or will be made in the future by LKAR to any Tribe official, member of the Tribal Council, relative of any tribal official or tribal government employee for the purpose of obtaining any special privilege, gain, advantage or consideration for LKAR, except for the fees payable to the Tribal Gaming Commission and amounts payable to the Tribe pursuant to this Amended Memorandum Agreement. However, nothing in this provision shall prohibit LKAR from making contributions to community organizations within the Tribe or to the Tribe for the purpose of funding community activities.

(c) No member of the Tribal Gaming Commission, or any tribal court official may be employed by the Enterprise or LKAR, or be a "Party in Interest" as defined in Section 9.1 (a) herein with respect to this Amended Memorandum Agreement or a gaming equipment agreement or have any direct or indirect financial interest in the gaming to be operated pursuant to this Amended Memorandum Agreement. Members of the Tribal Council and their relatives shall not be eligible for employment at the Facility and Enterprise but shall be eligible to enter into contracts for the provision of goods or services for the Facility and Enterprise.

(d) LKAR further agrees to comply with all conflict of interest rules set forth in regulations or ordinances of the Tribe.

4.6 <u>Participation in Tribe Functions</u>. LKAR acknowledges that personnel who are members of the Tribe have cultural and religious responsibilities to perform in regard to Tribe rituals and similar activities. LKAR, through the Enterprise's general manager, will schedule working hours and take other actions, with the assistance and advice of the Tribe, to accommodate Tribe members in performing these responsibilities without affecting their employment status or position.

## ARTICLE 5. <u>Insurance</u>

5.1 <u>Duty to Maintain</u>. LKAR, acting as agent for the Tribe, shall maintain during the course of this Amended Memorandum Agreement, appropriately allocated as a Cost of Gaming Operation or a Cost of Incidental Operations, insurance coverages in forms and amounts that will adequately protect the Tribe and LKAR, but in no case less than the amounts set forth in this Article, or as required by the Tribal-State Compact.

5.2 <u>Workers' Compensation</u>. L KAR, acting as a gent for the Tribe, shall maintain adequate workers' compensation insurance in accordance with all applicable laws, including

Shingle Springs Development/Management Contract 10/13/03 revision

.

Sψ

employer's liability insurance, in the amounts agreed to by the LKAR and Tribe, and as required by the Tribal-State Compact.

5.3 <u>Commercial General Liability</u>. LKAR, acting as agent for the Tribe, shall maintain commercial general liability insurance covering operations of the Enterprise, including b lanket contractual liability coverage, broad form property liability coverage, and personal injury coverage in the amount of \$1,000,000 per person/\$3,000,000 per occurrence for bodily injury and \$1,000,000 per person/\$3,000,000 per occurrence for bodily injury and \$1,000,000 per person/\$3,000,000 per occurrence for bodily injury and \$1,000,000 per person/\$3,000,000 per occurrence for bodily injury and \$1,000,000 per person/\$3,000,000 per occurrence for bodily injury and \$1,000,000 per person/\$3,000,000 per occurrence for bodily injury and \$1,000,000 per person/\$3,000,000 per occurrence for bodily injury and \$1,000,000 per person/\$3,000,000 per occurrence for bodily injury and \$1,000,000 per person/\$3,000,000 per occurrence for bodily injury and \$1,000,000 per person/\$3,000,000 pe

5.4 <u>Automobile</u>. LKAR, acting as agent for the Tribe, shall maintain comprehensive automobile liability insurance covering operations of the Enterprise, including all owned, hired and non-owned automobiles, trucks, buses, trailers, motorcycles or other equipment licensed for highway use with limits and coverage approved by the LKAR and Tribe.

5.5 <u>Tribe and LKAR to be Insured</u>. Insurance set forth in Sections 5.3 and 5.4 shall name the Tribe and LKAR as insureds, and such policies shall be endorsed to prohibit the insurer from raising tribal sovereign immunity as a defense to the payment of claim by the insurer.

5.6 <u>Property Insurance</u>. LKAR, acting as agent for the Tribe, shall procure replacement value all-risk casualty and extended hazard insurance in appropriate coverage amounts which shall insure the Facility and any fixtures, improvements and contents located therein against lost or damage by fire, theft and vandalism. Such casualty insurance policy or policies shall name the Tribe and LKAR, and the Facility Loan lenders as insureds. All such casualty insurance proceeds shall be applied to the immediate replacement of the applicable Facility part or fixture, improvements or contents therein unless the parties agree otherwise.

5.7 <u>Fidelity Bond</u>. LKAR, as agent for the Tribe, shall maintain fidelity bonds on such employees and in such amounts as LKAR and Tribe shall deem reasonable.

5.8 <u>Unemployment Insurance</u>. LKAR, as agent for the Tribe, shall maintain adequate unemployment compensation/disability insurance with respect to the Enterprise employees in compliance with the Tribal-State Compact.

5.9 <u>Evidence of Insurance</u>. Prior to the opening of the Facility to the public and commencing operations of the Enterprise and from time to time as reasonably requested by the Tribe, LKAR, as agent for the Tribe, shall supply to the Tribe and any necessary Governmental Authorities copies of the insurance policies applicable to the Facility or Enterprise operations as required by this Article.

5.10 Insurance Proceeds. Subject to the terms of Sections 7.4 and 7.6 hereof, any insurance proceeds received with respect to the Enterprise, except as provided in Section 5.6 hereof, shall be deposited into the Dominion Account and disbursed in accordance with the same terms and provisions applicable to Gross Total Revenues, provided, however, that if there is any insurance recovery for a claim related to the operation of the Enterprise for which either the Tribe or LKAR has previously paid from its own separate funds, then, to the extent of amounts paid by either of such parties, the insurance proceeds will be paid over to them and the balance shall be deposited into the Dominion Account as above.

#### ARTICLE 6. Budgets, Compensation and Reimbursement

6.1 <u>Projections and Budgets</u>.

( )

(a) The p arties h ave u sed their b est efforts to project expected revenues and expenses for the first three (3) years of operation of the Enterprise, and the projections, as described in the business plan to be submitted to the NIGC with this Amended Memorandum Agreement, represent the parties' mutual expectations.

(b) LKAR shall prepare a budget for the Initial Costs of Operation and submit such budget to the Tribe, for approval by the Tribal Council or its designated representative, within one hundred twenty (120) days of execution of this Amended Memorandum Agreement. LKAR shall also prepare an initial operating budget for the first Fiscal year and submit the same to the Tribe for approval by the Tribal Council at least ninety (90) days prior to the Commencement Date. Annual operating budgets shall be submitted by the Enterprise's general manager to the Tribe thereafter by no later than thirty (30) days prior to the commencement of each succeeding Fiscal Year. The proposed initial operating budget and each subsequent annual operating budget shall be subject to approval or disapproval within thirty (30) days of submission to the Tribe, such approval not to be unreasonably withheld. The parties recognize that mutually agreeable adjustments may be made to previously approved o perating budgets from time to time during any Fiscal Year, to reflect the impact of unforeseen circumstances, financial constraints, or other events. LKAR, shall keep the Tribe informed regarding any items of revenue or expense that are reasonably anticipated to cause a material change to the operating budget previously approved by the Tribe.

)

(c) LKAR shall prepare an annual capital expenditure budget and submit such budget to the Tribe, for approval by the Tribal Council or its designated representative, at least ninety (90) days prior to the Commencement Date. Annual capital expenditure budgets shall be submitted by the Enterprise's general manager to the Tribe thereafter by no later than thirty (30) days prior to the commencement of teach succeeding Fiscal Year. The proposed capital expenditure budgets shall be subject to approval or disapproval within thirty (30) days of submission to the Tribe for approval, such a pproval n ot to be unreasonably withheld. The parties recognize that mutually agreeable adjustments may be made to previously approved capital expenditure budgets from time to time during any budget year, to reflect the impact of unforeseen circumstances, financial constraints, or other events. LKAR, through the Enterprise's general manager, shall keep the Tribe informed and obtain the Tribe's approval regarding any projects or expenditures that are reasonably anticipated to cause a material change to the capital expenditure budget previously approved by the Tribe.

6.2 <u>Initial Costs of Operation</u>. The budget for the Enterprise's Initial Costs of Operations shall contain amounts approved by LKAR and the Tribe, which amounts shall be included as pre-opening costs and not as Costs of Gaming Operations or Costs of Incidental Operations.

6.3 <u>Minimum Guaranteed Monthly Payments</u>.

(a) During the term of this Management Agreement, the Enterprise shall, subject to the provisions of Section 6.3(b) below, pay the Tribe

beginning on the Commencement Date and continuing for the remainder of the term of this Management Agreement. The Minimum Guaranteed Monthly Payment shall be payable to the Tribe in arrears on the twentieth (20th) day of each calendar month following the month in which the Commencement Date occurs, F

Commencement Date is a date other than the first day of a calendar month, the first payment will be prorated from the Commencement Date to the end of the month. The Minimum Guaranteed Monthly Payment shall be prorated if gaming is conducted at the Facility for any other partial months.

(b) Minimum Guaranteed Monthly Payments shall be deducted from any disbursements of Net Total Revenues received by the Tribe under Section 6.5 hereof in any given month; provided, however,

Shingle Springs Development/Management Contract 10/13/03 revision

64

.7 Minimum

Guaranteed Monthly Payments shall be prorated with respect to any months (or portions thereof) that Class II Gaming or Class III Gaming is suspended or terminated at the Facility, and no Minimum Guaranteed Monthly Payments shall be required with respect to any months that no Class II Gaming or Class III Gaming is conducted at the Facility or accrue subsequent to termination of this Management Agreement.

Any obligations owing by the Tribe under the Operating Note shall be repaid (c) solely as a Limited Recourse obligation of the Tribe without any cross collateralization from other projects of Tribe and without any other liability or guarantee on the part of the Tribe except for the security interests described herein. Except for the Minimum Guaranteed Monthly Payment to the Tribe, repayment of the Operating Note obligations shall have first priority on any Net Gaming Revenues and Net Incidental Revenues generated by the Enterprise or any other Tribal Gaming Enterprise. Subject to the foregoing, the Tribe agrees to grant to LKAR a first priority and perfected security interest, including a Dominion Account arrangement pursuant to the Dominion Account Agreement (in a form consistent with the terms of this Amended Memorandum Agreement), on any Net Gaming Revenues and Net Incidental Revenues of the Enterprise or any other Tribal Gaming Enterprise in order to secure repayment of the Operating Note, and such Operating Note shall also be secured on a first priority and perfected basis by any Furnishings and Equipment financed by proceeds of the Facility Loan and Transition Loan pursuant to the Security Agreement. The Tribe agrees not to encumber any of the assets of the Facility or the Enterprise without the written consent of LKAR, which consent will not be unreasonably withheld; except that the Tribe shall have the right without the consent of LKAR to grant security interests in the Enterprise's revenues which are subordinate to LKAR's interests under this Amended Memorandum Agreement and all related Transaction Documents pursuant to a subordination agreement in form and substance acceptable to LKAR. The Tribe agrees to enter into a limited, transactional waiver of sovereign immunity and consent to jurisdiction and arbitration as to LKAR and in connection with the Operating Note, as provided in the Resolution of Limited Waiver.

6.4 <u>Daily and Monthly Statements</u>. LKAR shall furnish to the Tribe's designated representative statements identifying for each day the Gross Gaming Revenues attributable to the Enterprise's Class II Gaming and Class III Gaming on each day that such reports are normally available. Within fifteen (15) days after the end of each calendar month, LKAR, shall provide a verifiable financial statements in accordance with GAAP to the Tribe and the Tribal Gaming Commission covering the preceding month's operations of the Enterprise, including operating

6.5 <u>Distribution of Net Total Revenues</u>.

(a) All Net Total Revenues shall be disbursed on a monthly basis as set forth below, paid on the twentieth day of each calendar month for the preceding month. Such Net Total Revenues shall be disbursed from the Enterprise Bank Account(s) to the extent available for payment of the following accounts in the following order of priority:

(i) The Minimum Guaranteed Monthly Payment described in Section 6.3 hereof;

(ii) All outstanding Minimum Guaranteed Payment Advances and Working Capital Advances (and accrued interest thereon) or any other amounts owing to LKAR under the Operating Note or to the Tribe under Section 6.7 hereof;

(iii) Current principal, accrued interest and any other payments due on Facility Loan;

(iv) Current principal, accrued interest and any other payments due on the Land Acquisition Loan and the Transition Loan;

(v) Management compensation due LKAR under Section 6.5(b) below; provide that if the distribution under this subsection in any month is insufficient to fund such payment in full, the unpaid amount shall be deferred and paid under subsection (vi) below;

(vi) Any a mounts d eferred (including

Junder subsections (ii), (iii),

64

(iv) and (v) above;

(vii) Any monthly capital replacement or other reserve contributions which have been created with the written approval of LKAR and the Tribe;

(viii) Any indemnification or other obligations then owing by the Tribe to LKAR under any Transaction Document and not paid as Costs of Gaming Operations or Costs of Incidental Operations (provided LKAR has provided written notice to the

৫২

Tribe that above amounts are owed under the Transaction Documents, and the Tribe has not disputed the same or such amounts have been determined to be owing through an arbitration proceeding under Article 11 hereof); and

(ix) All remaining Net Total Revenues shall be disbursed to the Tribe on the same date the Management compensation is paid, subject to the rights of the LKAR under the Dominion Account Agreement upon the occurrence of a Material Breach by the Tribe.

(b) As compensation for LKAR's services, LKAR shall receive Jof Net Total Revenues for the prior calendar month, for so long as this Amended Memorandum Agreement shall remain in effect during the term hereof, as provided in this Amended Memorandum Agreement. Any amounts owing to LKAR hereunder shall be Limited Recourse obligations of the Tribe and shall be subject to the security provisions described in Section 6.3(c) hereof, including the Dominion Account Agreement and Security Agreement.

(c) LKAR, as agent for the Tribe, shall ensure that the Enterprise's general manager is responsible for making the Net Total Revenues disbursements to the appropriate party.

6.6 Annual Audit. With respect to each Fiscal Year, the Tribe shall cause an audit to be conducted by an independent certified public accountant from a Big Five accounting firm, who is licensed in California, has more than five (5) years experience in audits of gaming enterprise operations, and is to be selected and approved by the Tribe. On or before one hundred twenty (120) days after the end of such year, such accounting firm shall issue a report with financial statements in accordance with GAAP with respect to the preceding Fiscal Year (or portion of the year in the case of the first year) operations of the Enterprise, including operating statements, balance sheets, income statements and statements reflecting the amounts computed to be distributed in accordance with Section 6.5 hereof. In addition, upon termination of this Management Agreement portion of this Amended Memorandum Agreement in accordance with its terms, such accounting firm shall conduct an audit, and on or before ninety (90) days after the termination date, shall issue a report setting forth the same information as is required in the annual report, in each case with respect to the portion of the Fiscal Year ending on the termination date. If the Net Total Revenues or other amounts paid to the Tribe or LKAR in accordance with Section 6.5 (b) above for relevant period are different from the amount which should have been paid to such party based on the report prepared by the accounting firm and based upon the provisions of this Management Agreement, then to the extent either party received an overpayment, it shall repay and deposit the amount of such overpayment into the bank account referenced in Section 3.8 (a) hereof within twenty-five (25) days of the receipt by such party of the accountant's report, and to the extent either party received an underpayment, it shall receive a distribution from the bank account referenced in Section 3.8 (a) hereof of the amount of such underpayment within ten (10) days of the receipt by such party of the accountant's report. LKAR, as

agent for the Tribe, shall ensure that the Enterprise's general manager may make adjustment to future payments to correct a discrepancy if required distributions are not made.

6.7 Advances for Working Capital. Where amounts in bank accounts established pursuant to Section 3.8 are insufficient to meet Costs of Gaming Operations or Costs of Incidental Operation, during the first twelve (12) months after the Commencement Date, the LKAR shall advance monies to the Enterprise sufficient to meet Costs of Gaming Operations and Costs of Incidental Operations. Thereafter, the Tribe shall advance such monies to the Enterprise sufficient to meet Costs of Gaming Operations and Costs of Incidental Operations. If LKAR makes any advances hereunder ("Working Capital Advances", which advances shall be evidenced by an Operating Note in a form agreed to by the Tribe and LKAR and shall accrue interest rate[

64

くく

LKAR shall be repaid as provided in Section 6.5 hereof (and any amounts outstanding on account of Working Capital Advances at the end of the term of this Management Agreement shall be immediately due and payable by the Tribe). Any Working Capital Advances shall be Limited Recourse obligations of the Tribe and shall be subject to the security provisions described in Section 6.3(c) hereof, including the Dominion Account Agreement and Security Agreement. Any advances made by the Tribe hereunder shall accrue interest at the same rate as applies to the Transition Loan as described in Section 2.3(c) hereof and shall be repaid pursuant to Section 6.5 hereof.

6.8 <u>Development and Construction Cost Repayment</u>. The maximum dollar amount for repayment of development and construction costs for the Facility and Enterprise shall be Subject to any applicable Legal Requirements, the parties may increase the maximum repayment amount by mutual written agreement.

## PROVISIONS APPLICABLE TO BOTH THE DEVELOPMENT AND MANAGEMENT AGREEMENTS

# ARTICLE 7. <u>Termination</u>

#### 7.1 <u>Termination for Cause</u>.

Subject to the provisions of Section 9.2, either party may terminate this (a) Amended Memorandum Agreement if the other party commits or allows to be committed a Material Breach (as hereinafter defined) of this Amended Memorandum Agreement and fails to cure or to take steps to substantially cure such breach within thirty (30) calendar days after receipt of a written notice from the non-breaching party identifying the nature of the Material Breach in specific detail and its intention to terminate this Amended Memorandum Agreement. Termination is not an exclusive remedy for breach, and the non-breaching party shall be entitled to other rights and remedies as may be available. For purposes of this Amended Memorandum Agreement, a "Material Breach" is any of the following circumstances: (i) failure of LKAR to provide the Tribe with the Minimum Guaranteed Monthly Payments pursuant to Section 6.3, (ii) material failure of either party to perform in accordance with this Amended Memorandum Agreement for reasons not excused under Section 10.6 (Force Majeure), (iii) if any of LKAR's employees commits theft, embezzlement or crime of moral turpitude and if, after knowledge of such act or, if disputed, after determination by arbitration under Article 11, LKAR does not remove such employee from connection with Class II Gaming or Class III Gaming operations of the Enterprise within thirty (30) days after receipt of written notice, (iv) default under the Facility Note, the Land Acquisition Note, the Interim Promissory Note, the Operating Note, any other Transaction Document or any document or agreement related thereto by the Tribe, or (v) any representation or warranty made pursuant to Section 10.11 or 10.12 proves to be knowingly false or erroneous in any material way when made. Any final notice of termination hereunder shall be in writing detailing the reason the party considers the Material Breach not to be cured and must be delivered to the other party before such termination becomes effective.

(b) Notwithstanding any provision to the contrary herein, the parties agree that, for so long as the Tribe owes any amounts under the Facility Note, the Land Acquisition Note, the Interim Promissory Note or the Operating Note, the Tribe agrees that it will not terminate this Amended Memorandum Agreement without cause.

7.2 <u>Mutual Consent</u>. This Amended Memorandum Agreement may be terminated at any time upon the mutual written consent and approval of the parties.

7.3 <u>Involuntary Termination Due to Changes in Law or Tribal-State Compact</u>. The parties hereby agree to use their best efforts to conduct Class II Gaming and Class III Gaming activities in

accordance with this Amended Memorandum Agreement and to ensure that such activities and this Amended Memorandum Agreement conform to and comply with all applicable laws and the Tribal-State Compact. The Tribe agrees that, except as may be required by federal law, the Tribe will not enact or p ass any n ew ordinances subsequent to the execution of this A mended Memorandum Agreement that would materially impair the rights of LKAR under this Amended Memorandum Agreement. The Tribe will not enact any tax ordinance that will put the Facility or the Enterprise, or any portion thereof, at a competitive disadvantage with businesses in the same or like industries. In the event of any change in state or federal laws that results in a final determination by the Secretary, the National Indian Gaming Commission, or a court of competent jurisdiction that this Amended Memorandum Agreement is unlawful, the Tribe and LKAR shall use their best efforts to amend this Amended Memorandum Agreement in a mutually satisfactory manner which will comply with the change in applicable laws and not materially change the rights, duties and obligations of the parties hereunder. In the event such amendment is not practical, performance of this Amended Memorandum Agreement shall be automatically suspended effective upon the date that performance of this Amended Memorandum A greement becomes unlawful, and either party shall have the right to terminate such suspended Amended Memorandum Agreement (except the Notes and Security Provisions, as defined in Section 7.4 (a)), upon thirty (30) days written notice to the other party.

#### 7.4 Ownership of Assets and Repayment of Notes on Termination.

(a) Upon termination, except in connection with LKAR's and Lender's security interests in the Net Total Revenues of the Enterprise pursuant to the Tribe's Limited Recourse obligations under the Facility Note, the Land Acquisition Note, Interim Promissory Note and Operating Note (if not yet satisfied), or any other security interests or liens in any Furnishings and Equipment purchased with Facility Loan and Transition Loan proceeds or other purchase money agreements, the Tribe will retain full ownership of the Facility, Plans and Specifications therefor, and the Enterprise and its assets; and LKAR will have no rights to the Enterprise and its assets or the Facility (or any equipment, books and records, materials or furnishings therein that were purchased with Costs of Gaming Operations, Costs of Incidental Operations or Costs of Construction) except as to the security interests and liens recited above or as may be established otherwise by a proceeding pursuant to Article 11 hereof. In the event of any termination (whether voluntary or involuntary), the Tribe shall continue to have the obligation to pay unpaid principal and interest and other amounts due under either the Facility Note, the Land Acquisition Note, Interim Promissory Note or Operating Note executed in connection herewith, together with any unpaid compensation owed to LKAR under Section 6.5(b) hereof (if not yet satisfied), each of which shall become due and payable on such termination date. Any and all obligations and provisions contained in this Amended Memorandum Agreement concerning indemnity obligations or repayment of the Facility Note, Land Acquisition Note, Interim Promissory Note or Operating Note, and the security therefor, including the Security Agreement and Dominion Account Agreement, together with any unpaid compensation owed to LKAR under Section 6.5(b) hereof and the terms and provisions set forth in Articles 10 and 11 hereof

Shingle Springs Development/Management Contract 10/13/03 revision

64

\_7

excluding Sections 10.21 and 10.22 (collectively, the "Notes and Security Provisions"), shall survive termination of this Amended Memorandum Agreement.

(b) Subject to the provisions of Section 7.1 (b), in the event of termination of this Amended Memorandum Agreement for any reason prior to the full repayment to LKAR of any amounts owed to it by the Tribe under the Transaction Documents, including without limitation, the Facility Note, Land Acquisition Note, Interim Promissory Note or Operating Note, the Tribe shall,

7.5 <u>Notice of Termination</u>. In the event of termination pursuant to this Article, the Tribe shall provide notice of the termination to the Secretary or other appropriate Governmental Authorities within ten (10) days after the termination.

7.6 Cessation of Either Class II Gaming or Class III Gaming at the Facility.

(a) If, during the term of this Amended Memorandum Agreement, either Class II Gaming or Class III Gaming cannot be lawfully conducted at the Facility by reason of the application of any legislation or court or administrative agency order or decree adopted or issued by a governmental entity having the authority to do so, LKAR shall, within sixty (60) days after such legislation, order or decree becomes effective, elect to:

> (i) retain LKAR's interest in this Amended Memorandum Agreement and suspend both Class II and Class III Gaming operations until such date on which both Class II Gaming and Class III Gaming at the Facility becomes lawful (during which period the term of the Management Agreement will be tolled until both Class II Gaming and Class III G aming at the Facility becomes lawful, and the period of cessation shall not be deemed to have been part of the term of the Management Agreement and the term shall be extended by the length of time of the cessation); or

> (ii) retain LKAR's interest in this Amended Memorandum Agreement, suspend both Class II Gaming and Class III Gaming operations until such date on which both Class II Gaming and Class III Gaming at the Facility becomes lawful

(during which period the term of the Management Agreement will be tolled until Class III Gaming at the Facility becomes lawful, and the period of cessation shall not be deemed to have been part of the term of the Management Agreement and the term shall be extended by the length of time of the cessation), and with the prior approval of the Tribe, which approval shall not be unreasonably withheld, use the Facility for any other lawful purpose pursuant to a use agreement containing terms reasonably acceptable to LKAR and the Tribe; or

(iii) terminate both Class II Gaming and Class III Gaming operations and terminate this Amended Memorandum Agreement.

LKAR shall give the Tribe written notice of LKAR's election within such sixty (60) day period.

(b) If LKAR elects to retain its interest in this Amended Memorandum Agreement under Section 7.6 (a)(i) or (ii) above, LKAR shall have the right (but not the obligation) to commence either Class II Gaming and Class III Gaming operations within sixty (60) days after the date on which both Class II Gaming and Class III Gaming becomes lawful. LKAR may exercise such right by giving the Tribe written notice of such exercise within thirty (30) days after the date on which both Class II Gaming and Class III Gaming becomes lawful.

(c) If LKAR elects to terminate this Amended Memorandum Agreement under this Section 7.6, the provisions of Section 7.4 above shall apply.

(d) If, during the term of this Amended Memorandum Agreement, the Facility is damaged by casualty or other occurrence to the extent, as reasonably determined by LKAR, that either Class II or Class III Gaming cannot be conducted at the Facility, LKAR shall, within sixty (60) days after such casualty or occurrence, elect to:

(i) retain LKAR's interest in this Amended Memorandum Agreement pending repair or reconstruction of the Facility, suspend both Class II Gaming and Class III Gaming operations pending the repair or reconstruction of the Facility (during which period the term of the Management Agreement will be tolled until Class III Gaming is recommenced at the Facility, and the period of cessation shall not be deemed to have been part of the term of the Management Agreement and the term

Shingle Springs Development/Management Contract 10/13/03 revision

shall be extended by the length of time of the cessation), and arrange for such repair or reconstruction in the manner described in this Section 7.6; or

(ii) terminate both Class II Gaming and Class III Gaming operations and terminate this Amended Memorandum Agreement, such termination to be effective on the thirtieth (30<sup>th</sup>) day after written notice of termination has been delivered to the Tribe.

LKAR shall give the Tribe written notice of LKAR's election within such sixty (60) day period.

(e) If LKAR elects to retain its interest in this Amended Memorandum Agreement under Section 7.6(d)(i) above, LKAR shall promptly verify the amount of insurance or other proceeds available to pay the cost of repair, replacement or reconstruction. LKAR is hereby granted the authority to submit, adjust and settle, on behalf of the Tribe, all insurance claims associated with the casualty or occurrence; provided, however, that LKAR shall obtain the Tribe's prior written consent (which consent shall not be unreasonably withheld) to any settlement. LKAR shall provide copies of all settlement documents to the Tribe.

7.7 <u>Renewal Option</u>. Subject to Section 3.19, the parties by mutual agreement may decide to renew or extend the Management Agreement of this Amended Memorandum Agreement. Any such renewal or extension shall only become effective upon approval by the NIGC and appropriate licensing by the Tribal Gaming Commission.

7.8 <u>Buyout Option</u>.

Shingle Springs Development/Management Contract 10/13/03 revision

Ĺ

### ARTICLE 8. Release and Indemnity

8.1 <u>Third-Party Claims</u>. Except for Section 2.4(h) and except for claims resulting from the other party's own gross negligence or willful or criminal misconduct, neither party shall be entitled to recover from, and expressly releases, the other party, its agents, officers and employees, from or for any third-party damages, claims, causes of action, losses and/or expenses of whatever kind or nature, including attorneys' fees and expenses incurred in defending such claims in connection with the lawful operation of the Facility and Enterprise in accordance with the terms of this Amended Memorandum Agreement, and such claims, damages, losses or expenses shall be considered either Costs of Gaming Operations or Costs of Incidental Operations, depending on the circumstances and nature of the claim, payable from the bank account established pursuant to Section 3.8(a).

8.2 Indemnity from LKAR. Notwithstanding Section 8.1, LKAR shall indemnify and hold the Tribe harmless against any and all damages, claims, losses or expenses of whatever kind or nature, including reasonable attorneys' fees resulting from the criminal misconduct and intentional torts of LKAR, its officers and directors in connection with LKAR's performance of this Amended Memorandum Agreement, and no such damages, losses or expenses shall be paid from the bank accounts established pursuant to Section 3.8(a), nor shall such losses or expenses be considered Costs of Gaming Operations or Costs of Incidental Operations.

8.3 <u>Indemnity from Tribe</u>. Notwithstanding Section 8.1, Tribe shall indemnify and hold LKAR harmless against any and all damages, claims, losses or expenses of whatever kind or nature, including reasonable attorneys' fees resulting from the criminal misconduct and intentional torts of the Tribe, its officers, directors, or tribal government employees, in connection with Tribe's performance of this Amended Memorandum Agreement, and no such damages, losses or expenses shall be paid from the bank accounts established pursuant to Section 3.8(a), nor shall such losses or expenses be considered Costs of Gaming Operations or Costs of Incidental Operations.

8.4 Indemnity Against Unauthorized Debt and Liabilities. The parties expressly agree that neither this Amended Memorandum Agreement nor its performance creates or implies a partnership between the parties or authorizes either party to act as agent for the other except to the extent expressly provided herein. LKAR hereby agrees to indemnify and hold the Tribe harmless from any third-party claims, actions and liabilities, including reasonable attorneys' fees on account of obligations or debts of LKAR that LKAR is not authorized to undertake as agent for the Tribe pursuant to the terms of this Amended Memorandum Agreement. The Tribe likewise agrees to indemnify and hold LKAR harmless from any third-party claims, actions and liabilities on account of any of the separate obligations or debts of the Tribe that are not authorized Costs of Gaming Operations, Costs of Incidental Operations or Costs of Construction pursuant to this Amended Memorandum Agreement.

#### ARTICLE 9. Parties in Interest

64

(a) As used in this Section 9.1, the term "Parties in Interest" includes any person or entity with a financial interest in, or having management responsibility for, this Amended Memorandum Agreement or for which background investigations are required by 25 C.F.R. Part 537, or by tribal or federal law, or the Tribal-State Compact and any amendments thereto.

(b) LKAR shall require sufficient information and identification from each "Party in Interest" to perform a background investigation for the purpose of determining the suitability of such persons for employment in a Class II Gaming and Class III Gaming operation, including, at a minimum, the information required by the National Indian Gaming Commission as set forth in 25 C.F.R. Part 537, by federal and tribal law, and by the Tribal-State Compact, and any amendments thereto.

(c) Without limiting the foregoing, LKAR shall obtain a current set of fingerprints on each person for whom background investigations are required by the Tribal Gaming Commission and the National Indian Gaming Commission, using forms supplied by the National Indian Gaming Commission and/or the Tribal Gaming Commission, which shall be referred to the Federal Bureau of Investigation (FBI) Fingerprint Identification Division or other law enforcement agency designated by the Tribal Gaming Commission.

(d) The parties hereby agree that a listing of all "Parties in Interest" as defined in Section 9.1 (a) above shall be submitted to the NIGC. All such "Parties in Interest", as such listing shall be supplemented from time to time, shall be required to furnish the information required by this Section 9.1 prior to obtaining such interest. All necessary Governmental Authorities must approve any change in the "Parties in Interest". Any deletion of a person listed as one of the "Parties in Interest" shall not constitute a change in persons with a financial interest in or management responsibility for a management contract.

9.2 Removal: Divestiture. Should the Tribal Gaming Commission or the National Indian Gaming Commission, by agency action, find that any person or entity with either a "direct or indirect financial interest" in this Amended Memorandum Agreement (as defined in 25 C.F.R. §502.17, and any amendments thereto) or with management authority hereunder, whose prior activities, criminal record, if any, or reputation, habits, and associations pose a threat to the public interest, or the tribal interest, or the effective regulation of gaming, or create or enhance the dangers of unsuitable, unfair, or illegal practices and methods and activities in the conduct of gaming or the carrying on of related business and financial arrangements, or should such agency revoke the license of such person or entity, and should either agency notify LKAR or the Tribe of such finding or revocation, then LKAR shall immediately take all necessary steps to require such person or entity to divest their interest in this Amended Memorandum Agreement and shall immediately remove such person or entity from all association with operations under this Amended Memorandum Agreement upon receipt of such notice; provided that any person or entity subject to such removal/divestiture shall be permitted to be revested and able to associate with operations in the event the agency action is reversed upon agency administrative or judicial appeal. In addition, if any person or entity with "direct or indirect financial interest" in this Amended Memorandum Agreement (as defined in 25 C.F.R. §502.17, and any amendments thereto) or with management authority hereunder: (a) has been or is subsequently convicted of a felony relating to gaming, (b) knowingly or willfully provided materially false statements to the Tribe, the Tribal Gaming Commission or the National Indian Gaming Commission, or refused to respond to questions from either of such agencies, or (c) attempts to unduly interfere or unduly influence for their gain or advantage any decision or process of tribal government relating to Class II Gaming or Class III Gaming, and if LKAR becomes aware of such conflicts or prohibited actions, then LKAR shall notify Tribe of such event and immediately take all necessary steps to cause such person or entity to divest their interest in LKAR. No person or entity required under this provision to divest itself of its interest in LKAR or this Amended Memorandum Agreement, or subject to removal under this provision, shall as a result thereof have any legal or other recourse against the Tribe or any officer, employee, or agent of the Tribe as a result of such divestiture except as provided in Article 11 or pursuant to applicable law or regulations.

Shingle Springs Development/Management Contract 10/13/03 revision

#### ARTICLE 10.

#### **Miscellaneous**

10.1 <u>Assignment and Subcontractors</u>. The rights and obligations under this Amended Memorandum Agreement shall not be assigned or subcontracted by any party without the prior written consent of the other party and without first obtaining prior approval by the National Indian Gaming Commission or the BIA, if applicable, and any other necessary regulatory approvals. However, the Tribe reserves the right to assign its rights and obligations under this Amended Memorandum Agreement to a tribally chartered entity or an IRA section 17 corporation that it wholly owns and controls and the LKAR reserves the right to assign its rights and obligations under this Amended Memorandum Agreement to a wholly owned subsidiary, provided that LKAR shall have received prior approval from the NIGC and any other necessary regulatory approvals. Other than as expressly provided herein or in Section 10.2 below, any attempted assignment or subcontracting without such consent and approval shall be void. Approval of any assignment or subcontract to any new party must be preceded by a complete background investigation of the new party as required by Section 9.1. Subject to the preceding requirements, this Amended Memorandum Agreement shall be binding upon and shall inure to the benefit of the parties and their respective successors and assigns.

Change of Control in Ownership Interest; Severability. LKAR is a Delaware limited 10.2 liability company whose members are Lakes Shingle Springs, Inc. and Kean Argovitz Resorts-Shingle Springs, LLC. In the event that either of LKAR's members, as an entity, either: (i) has its gaming license withdrawn or fails to obtain a gaming license, each after exhaustion of all available administrative and other legal due process, and fails, despite good faith efforts to do so, to cure the condition causing the license withdrawal or failure to obtain a license within the time frames contained in Section 7.1; or (ii) for any reason is rendered legally incapable of continuing its participation in this A mended Memorandum Agreement, the Tribe agrees and consents to the cessation of all participation by that member of LKAR in this Amended Memorandum Agreement and to the assumption by the remaining member of LKAR of all rights and obligations pursuant to this Amended Memorandum Agreement. In the event that either of LKAR's members, as an entity, with the consent of the other member requests permission from the Tribe to withdraw from participation in this Amended Memorandum Agreement, the Tribe shall consider in good faith granting its consent, which consent shall not be unreasonably withheld. In granting such consent, the Tribe agrees and consents to the cessation of all participation by that member of LKAR in this Amended Memorandum Agreement and to the assumption by the remaining member of LKAR of all rights and obligations pursuant to this Amended Memorandum Agreement. Any Change of Control (as defined herein) in LKAR shall require prior written consent of the Tribe and be subject to Legal Requirements, or this Amended Memorandum Agreement shall be terminated. For purposes of this Amended Memorandum Agreement, a "Change of Control" means the acquisition by any legal entity,

person or affiliated group of persons not presently members of LKAR of beneficial ownership of fifty one percent (51%) or more of membership interest in LKAR.

10.3 <u>Notices</u>. Any notice, consent or any other communication permitted or required by this Amended Memorandum Agreement shall be in writing and shall be effective on the date sent and shall be delivered by personal service, via telecopier with reasonable evidence of transmission, express delivery or by certified or registered mail, postage prepaid, return receipt requested, and, until written notice of a new address or addresses is given, shall be addressed as follows:

If to the Tribe:	Shingle Springs Band of Miwok Indians P.O. Box 1340
	Shingle Springs, CA 95682
	Attention: Chairman
With a copy to:	Anthony Cohen, Esq.
	Clement, Fitzpatrick & Kenworthy
	3333 Mendocino Ave.
	Suite 200
	Santa Rosa, CA 95403
If to the LKAR:	Lakes KAR-Shingle Springs, LLC
	130 Cheshire Lane
	Minnetonka, MN 55305
	Attention: Timothy J. Cope
With a copy to:	Kevin C. Quigley, Esq.
	Hamilton Quigley Twait & Foley PLC
	W1450 First National bank Building
	332 Minnesota Street
	St. Paul, MN 55101-1314
and	Brian Klein, Esq.
	Maslon, Edelman, Borman & Brand, LLP
	3300 Wells Fargo Center
	90 South Seventh Street
	Minneapolis, MN 55402-4140

Copies of any notices shall be given to the Tribal Gaming Commission.

10.4 <u>Amendments</u>. This Amended Memorandum Agreement may be amended only by written instrument duly executed by all of the parties hereto and with any and all necessary regulatory approvals previously obtained.

10.5 <u>Counterparts</u>. This Amended Memorandum Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same agreement.

10.6 <u>Force Majeure</u>. No party shall be in default in performance due hereunder if such failure or performance is due to causes beyond its reasonable control, including acts of God, war, fires, floods, or accidents causing damage to or destruction of the Facility or property necessary to operate the Facility, or any other causes, contingencies, or circumstances not subject to its reasonable control which prevent or hinder performance of this Amended Memorandum Agreement.

10.7 <u>Time is Material</u>. The Parties agree that time is of the essence and the time and schedule requirements set forth in this Amended Memorandum Agreement are material terms of this Amended Memorandum Agreement.

10.8 <u>Further Assurances</u>. The parties hereto agree to do all acts and deliver necessary documents as shall from time to time be reasonably required to carry out the terms and provisions of this Amended Memorandum Agreement.

10.9 <u>Severability</u>. In the event that any provision of this Amended Memorandum Agreement is, by final order of a court of competent jurisdiction or Government Authority, held to be illegal or void, the validity of the remaining provisions of the Amended Memorandum Agreement shall be enforced as if the Amended Memorandum Agreement did not contain such illegal or void clauses or provisions, and the parties shall use their best efforts to negotiate an amendment to this Amended Memorandum Agreement which will comply with the judicial order and maintain the originally contemplated rights, duties and obligations of the parties hereunder.

10.10 <u>Sovereign Immunity</u>. Except as described in the Resolution of Limited W aiver attached hereto as <u>Exhibit B</u> and incorporated herein by reference, nothing in this Amended Memorandum Agreement shall be deemed or construed to constitute a waiver of sovereign immunity of the Tribe and the only applicable waivers of sovereign immunity shall be those expressly provided and executed by the Tribe's duly authorized representative and substantially conforming to the form as approved by the parties. The parties agree that they will not amend or alter the Resolution of Limited Waiver in any way which will lessen the rights of any party as set forth in the Resolution of Limited Waiver. This Section 10.10 shall survive termination of this Amended Memorandum Agreement, regardless of the reason for the termination.

10.11 <u>Representations and Warranties of LKAR</u>. The LKAR hereby represents and warrants as follows:

(a) This Amended Memorandum Agreement has been duly executed and delivered by LKAR and, when approved by necessary Governmental Authorities as set forth (where applicable), will constitute a valid and binding obligation, enforceable against LKAR in accordance with its terms.

(b) The execution and delivery of this Amended Memorandum Agreement, the performance by LKAR of its obligations hereunder and the consummation by LKAR of the transactions contemplated hereby will not violate any contract or agreement to which LKAR or any of its affiliated companies is a party or any law, regulation, rule or ordinance or any order, judgment or decree of any federal, state, tribal or local court or require any regulatory approval beyond those contemplated herein.

(c) LKAR has the full legal right, power and authority and has taken all action necessary to enter into this Amended Memorandum Agreement, to perform its obligations hereunder, and to consummate all other transactions contemplated by this Amended Memorandum Agreement.

(d) LKAR has conducted due diligence investigations and has satisfied itself as to the present and past relationships between Tribe and all other entities in connection with the Tribe's economic development activities, including but not limited to Sharp Image Gaming, Inc., and, subject to the provisions in Section 2.12, waives any claim that such relationships and the Tribe's legally required performance thereunder, if any, constitute a breach of this contract or an actionable tort.

10.12 <u>Representations and Warranties of Tribe</u>. The Tribe hereby represents and warrants as follows:

(a) The Tribe is duly organized Indian tribe under the laws of the Tribe and the United States.

(b) The Tribe has full legal right, power and authority under the laws for the Tribe and has taken all official Tribal Council action necessary (i) to enter into this Amended Memorandum A greement and a uthorize the Tribe to e xecute and d eliver this Amended Memorandum Agreement, the Operating Note, Dominion A ccount A greement, S ecurity Agreement, the F acility Loan d ocumentation, F acility Note, Land Acquisition Note and Interim Promissory Note, and any and all other documents and agreements related thereto (collectively and as amended, renewed, or extended from time to time, the "Transaction

Documents"), (ii) to perform its obligations hereunder, and (iii) to consummate all other transactions contemplated by this Amended Memorandum Agreement and other Transaction Documents.

(c) This Amended Memorandum Agreement (and the other Transaction Documents), when executed and delivered by Tribe and approved by necessary Governmental Authorities, if required, including the Tribe, will constitute a valid and binding obligation, enforceable against Tribe in accordance with their terms.

(d) Subject to the provisions of Section 10.11(d), the execution and delivery of this Amended Memorandum Agreement (and other Transaction Documents), the performance by Tribe of its obligations hereunder and thereunder, and the consummation by Tribe of the transactions contemplated hereby will not violate any contract or agreement to which Tribe is a party, law, regulation, rule or ordinance or any order judgment or decree of any federal, state, tribal or local court, or require any approval by Governmental Authorities beyond those contemplated herein.

10.13 <u>Governing Law</u>. This Amended Memorandum Agreement has been negotiated, made and executed at the Tribe's office located in the State of California and shall be construed in accordance with the applicable laws of the State of California, without regard to its conflict of laws provisions, and applicable Tribe and federal laws.

10.14 <u>Entire Agreement</u>. This Amended Memorandum Agreement, including all exhibits, represents the entire agreement between the parties and supersedes all prior agreements relating to the subject matter of Class II Gaming and Class III Gaming to be developed and conducted by the Tribe at the Facility and operations of the Enterprise. The parties hereto intend that this Amended Memorandum Agreement (and related Transaction Documents) is to supercede and replace the May 5, 2000 Memorandum of Agreement and any other prior agreements between the parties in connection with the Project.

10.15 <u>Representatives of Tribe</u>. The Tribal Council shall furnish to LKAR a list of the authorized representatives who are empowered to act on behalf of the Tribe for the purposes of this Amended Memorandum Agreement and the Tribe shall keep such list current. The Tribe hereby acknowledges and agrees that to the extent any authorization, consent or other approval of the Tribe or the Tribal Council is required under this Amended Memorandum Agreement or any related Transaction Documents and the Tribe shall provide to LKAR a tribal resolution naming any individual or individuals authorized to represent the Tribe and the Tribal Council for purposes or for the purpose of the operation and performance of this Amended Memorandum Agreement and related Transaction Documents, then LKAR shall be entitled to rely on all decisions, authorizations, consents,

and approvals provided by such individual or individuals, as applicable, until such time as the Tribe shall deliver to LKAR an additional tribal resolution revoking or otherwise modifying such authority.

)

10.16 <u>Limitations of Liability</u>. LKAR expressly agrees that the Tribe's total aggregate liability for damages for breach of the Amended Memorandum Agreement shall be limited in accordance with the Resolution of Limited Waiver attached hereto as <u>Exhibit B</u> and incorporated herein by reference. The Tribe shall bear no liability for further damages.

10.17 <u>Approvals</u>. Unless otherwise provided herein, all approvals or consents required by either party hereunder shall not be unreasonably withheld or delayed, unless otherwise provided herein. Approval by the Tribal Council or its duly authorized representative shall be deemed to constitute approval by the Tribe and approval by the duly authorized President of LKAR shall be deemed to constitute approval by the LKAR.

10.18 <u>Best Efforts</u>. LKAR and the Tribe shall use their best efforts to perform and fulfill their obligations under this Amended Memorandum Agreement in the manner required by this Amended Memorandum Agreement.

10.19 <u>Request for NIGC Approval</u>. The parties specifically request that the NIGC, or the Secretary where appropriate, approve this Amended Memorandum Agreement and other Transaction Documents, if required, or declare that such approval is not required.

10.20 Non-disclosure. The parties agree not to divulge to third parties the terms of this Amended Memorandum Agreement or any other proprietary or confidential information exchanged between the parties pursuant to this Amended Memorandum Agreement, unless (i) the information is required to be disclosed pursuant to judicial or Legal Requirements, (ii) the information is at the time of disclosure already in the public domain, or (iii) to the extent required in order to obtain financing. This prohibition shall not apply to disclosures by either party to their attorneys, accountants, or other professional advisers. In situations where disclosure of the terms of this Amended Memorandum Agreement to regulatory, governmental or judicial entities is required by law or regulations, the parties will make reasonable efforts to secure confidential treatment of the terms of this Amended Memorandum Agreement by such entities; provided, however, this disclosure restriction shall not prohibit LKAR making any SEC filings it deems legally necessary. The parties agree to consult with each other and cooperate regarding any press releases regarding this Amended Memorandum Agreement and the relationships described herein. LKAR understands that as a tribal government, the Tribe is obligated to fully inform, consult with, and obtain the consent of its membership to its entry into and performance of this Amended Memorandum Agreement. LKAR also understands that the Tribe has no control over the distribution by its members of this Amended Memorandum Agreement or their dissemination of other confidential information. LKAR therefore agrees that this section is intended only to bind the Tribe as a legal entity, and its officers and agents, and the Tribe shall have no liability for failure of any other person to comply with this provision.

)

64 66

10.21 <u>Non-Competition</u>. LKAR agrees that, during the term of this Amended Memorandum Agreement, it will not finance, manage, or consult in connection with any facility where Class II Gaming or Class III Gaming is or will be conducted within a radius of \_\_\_\_\_?miles from the Gaming Site or \_\_\_\_\_\_area without the prior written consent of the Tribe (excluding \_\_\_\_\_\_). The Tribe agrees that, during the term of this Amended Memorandum Agreement, it will not solicit or enter into any negotiations or agreements with any person or company with respect to any Class II Gaming or Class III Gaming conducted within \_\_\_\_\_?miles of the Gaming Site without the prior written consent of LKAR.

10.22 <u>Other Business Opportunities</u>. LKAR agrees that, in an effort to develop a long-lasting business relationship with the Tribe, LKAR will make its best efforts to present appropriate business and investment opportunities to the Tribe.

10.23 <u>Use of Trade Marks and Trade Names</u>. To assure that the Tribe can continue operation of the Facility and Enterprise without disruption in the event that this Amended Memorandum Agreement is terminated or not renewed, LKAR agrees that it will not use any trade mark or trade name to identify any portion of the Facility or Enterprise or services offered within the Facility or Enterprise unless such trade mark or trade name is registered in the name of the Tribe.

## ARTICLE 11. Dispute Resolution

11.1 <u>Disputes Between the Enterprise and Patrons</u>. Disputes that arise between the Enterprise and any p atron of the F acility s hall be r esolved in a ccordance with the Tribal-State Compact and tribal ordinances, if applicable.

11.2 <u>Disputes Between the Enterprise and Enterprise Employees</u>. The Tribe and LKAR shall jointly develop an employee dispute resolution policy and LKAR, as agent of the Tribe, shall ensure that the enterprise's general manager shall implement and administer the employee dispute resolution policy after its adoption.

11.3 <u>Disputes Between the Tribe and LKAR</u>. Disputes between the Tribe and LKAR with respect to this Amended Memorandum Agreement, the Land Acquisition Note, the Interim Promissory Note, the Facility Note, Operating Note, or any other Transaction Document or a party's performance hereunder or thereunder, shall be resolved by the following dispute resolution process herein and pursuant to the Resolution of Limited Waiver attached hereto.

(a) The parties shall first meet and confer in a good faith attempt to resolve the dispute through negotiations not later than ten (10) calendar days after receipt of written notice of the dispute, unless both parties agree in writing to an extension of time.

(b) If the dispute is not resolved to the satisfaction of the parties within thirty (30) calendar days after the first meeting in Section 11.3(a) above, then any claim, controversy or dispute arising out of or relating to this Amended Memorandum Agreement, Facility Note, Land Acquisition Note, the Interim Promissory Note, the Operating Note, or any other Transaction Document, or any alleged default hereunder or breach of any provisions thereof, shall be submitted to binding arbitration in accordance with the Commercial Arbitration Rules of the American Arbitration Association in effect at the time of submission; except that: (a) the question whether or not a dispute is arbitrable under this Amended Memorandum Agreement or any other Transaction Document shall be a matter for binding arbitration by the arbitrators, such question shall not be determined by any court and, in determining any such question, all doubts shall be resolved in favor of arbitrability; and (b) discovery shall be permitted in accordance with the Federal Rules of Civil Procedure, subject to supervision as to scope and appropriateness by the arbitrators. Judgment on any arbitration award may be entered in any court having jurisdiction over the parties pursuant to the Resolution of Limited Waiver attached hereto as <u>Exhibit B</u> and incorporated herein by reference.

(c) Unless the parties hereto otherwise agree in writing prior of the submission of such claim, controversy or dispute to arbitration, arbitration proceedings under this Article 11 shall be held in Sacramento, California.

(d) Either party may, at any time prior to the selection of an arbitrator or arbitrators, require that the arbitrator or arbitrators selected be an attorney or attorneys licensed to practice law in California and that the attorneys have experience in Indian gaming regulatory and development issues.

(e) Unless the parties hereto otherwise agree in writing, any matter to be arbitrated shall be submitted to a panel of three arbitrators. One arbitrator shall be selected by the Tribe, one arbitrator shall be selected by LKAR and the third arbitrator shall be selected by mutual agreement of the two arbitrators selected by the parties hereto.

(f) The arbitration award shall be in writing signed by each of the arbitrators, and shall state the basis for the award. The arbitration award shall be set forth in reasonable detail as to its findings of fact and law, and basis of determination of award form and amount. In connection with any arbitration award, the arbitrators shall be empowered to take the actions and enforce the judicial remedies described in Paragraph 5 of the Resolution of Limited Waiver; provided however, that although the arbitrators may award damages in the event the Tribe or the Tribal Gaming Commission do not to comply with the award, the arbitrators may not require the Tribe or the Tribal Gaming Commission to take or modify any governmental legislative decision or action which the arbitrators have determined has resulted in the dispute between the parties and is contrary to the parties rights, liabilities or obligations under this Amended Memorandum Agreement, the Facility Note, the Land Acquisition Note, the Interim Promissory Note, the Operating Note, or any other Transaction

Document ("Specific Performance Restriction"). Provided further, that: (a) should the arbitrators determine that there has been an intentional bad faith violation of a party's rights under this Amended Memorandum Agreement or any other transaction Document by the Tribe or Tribal Gaming Commission, and if the Tribe or the Tribal Gaming Commission do not reverse such intentional bad faith violation through governmental legislative decision or action within thirty (30) days after the being notified by the arbitrators of such determination, then the arbitrators shall

64

74

(g) Except to the extent such enforcement will be inconsistent with a specific provision of this Amended Memorandum Agreement, arbitration awards made pursuant to this Article 11 shall be enforceable in federal court under Title 9 of the United States Code and any applicable tribal, federal or state law governing the enforcement of arbitration awards.

(h) In addition to any basis for appeal of an arbitration award stated in Title 9 of the United States Code or any applicable law governing the enforcement of arbitration awards, either party hereto may appeal an arbitration award on the basis that the arbitrator or arbitrators incorrectly decided a question of law in making the award, or the award was made in an arbitrary or capricious manner or in manifest disregard of the factual evidence.

(i) Either party hereto, without having to exhaust any tribal remedies first, shall have the right to seek and obtain a court order from a court having jurisdiction over the parties requiring that the circumstances specified in the order be maintained pending completion of the arbitration proceeding, to the extent permitted by applicable law.

(j) The parties agree that: (1) in the event that a dispute submitted to arbitration under this section involves the right of LKAR to continue to receive compensation under Section 6.5 hereof, or (2) in the event that LKAR appeals an agency decision under Section 9.2 for removal/divestiture, then any compensation which is asserted to be due to LKAR during such dispute shall.

О с ј .

64

[Rest of page left blank intentionally; signature page to follow]

Shingle Springs Development/Management Contract 10/13/03 revision

IN WITNESS WHEREOF, the parties hereto have executed this Amended Memorandum Agreement, subject to separate written approval of all exhibits referred to herein and related Transaction Documents.

Shingle Springs Band of Miwok Indians By: Its: Tribal Chairman

By: E Its: Wczechatrman.

Approved pursuant to 25 U.S.C. \$2711 JUL 19 2004 National Indian Gaming Comprission

By:

( )

Print Name: Philip N. Hogen Its Chairman Lakes KAR-Shingle Springs, LLC

Bv: Its: \_

Shingle Springs Development/Management Contract 10/13/03 revision

## LIST OF EXHIBITS

Exhibit A Legal Description of Gaming Site

()

()

Exhibit B Resolution of Limited Waiver of Immunity from Suit

#### <u>Exhibit A</u> <u>Legal Description of Gaming Site</u> [tribal lands located within the exterior boundaries of the Tribe's Rancheria]

The Gaming Site is legally described as follows:

The NW ¼ of Section 29, Township 10 North, Range 10 East, El Dorado County, California.

See attachment

Bk,03 和我的 139 In and that executed the within instrument, and also known to me to be the persons who execu ted the same on bahalf of the corporation therein named, and they acknowledged to me that such corporation executed the same. IN WITHERS WHEREOF, I have herewanto set my hand and affixed my official seal, et my office in the City and County of San Francisco, the day and year in this Certificate first abovs written. (SEAL) O.A. ECGERS .... Notary Public in and for the city and county of San Francisco, State of California. Filed for record at the reducat of Inter-County Title Co. Rovember 7, 1925, et 51 minutes past 9 O'clock A.M. Ð NO. 1627. CHAS E. MARSH Recorder of El Dorado County. ⑪  $\prod^{e}$ \*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\* CLAUDE E. COOPER, as admr. to UNITED STATES OF AMERICA. \*\*\*\*\*\*\*\*\*\*\* THIS INDENTURE, made this lithday of March, 1920, between Claude E.Cooper, as Administrator of the estate of Walter J.Keldrum, also known as W.J.Neldrum, deceased, the merty of the first mart and UNITED STATES OF ANTRICA, the perty of the second pert, WITNESSETH: THAT WHEREAS, the Superior Court of the State of California, in and for the County of Sacramento, did on the 6th day of October, 1919, make an order decreeing that a conveyance be erecuted to the party of the second nart of the property heraingfter des-"Cribed, a certified copy of said order having been recorded in the office of the County Recorder of the County of El Dorado, State of California, within which county the said lands are situated, on the 30th day of October, 1919, which said order is now on file and of record in the office of the County Recorder of the County of El Dorado, State of Califor-R gia, which said record thereof in said Recorder's office, is hereby referred to for greater weitainty. ŭ 1501 NOW, THEREFORE, said Claude E.Cooper, as administrator of the estate of Walter J. Hold a, also known as W.J.Meldrum, deceased, marty of the first part, pursuant to said order aforesaid, of said Superior Court for end in consideration of the sum of \$1400.00, lawful money of the United States of America, to him in hand haid by the said n-rty of the second part, the receipt of which is hereby acknowledged, has greated, bargained, sold and conveyed, and by these presents, does grant, bargain, sell and convey unto the said party of the second mert all right, title, interest and estate of the said Walter J.Meldrum, also known as W.J. Meldrum, at the time of his death, and all the right, title and interast that the said estate or said administrator, by operation of law or otherwise way have "secured, in addition to that of said deceased, at the time of his death. In and to all that certain real property situated in the County of El Doredo, Stat. of California, and bounded and particularly described as follows, to-wit:

140 ten(10) N The Northwest one-quarter(N. 72) of section twenty-nine(29) in township Range Ten(10) East, M.D.BdM. فليوش TOGETHER with all and singular the tenements, hareditaments and appurtenan whatsoever to the same belonging or in any way appertaining. 194-1 TO HAVE AND TO HOLD all and singular the above mentioned and described premise together with the appurtenances, unto the said United States of America, for the use and occupancy of the Sacremento Verona-Band of Homeless Indians. IN WITNESS WHEREOF, the seid party of the first part, as administrator of the estate of sold deceased, as aforesaid, has executed these presents, the day and year fir above written. CLAUDE E. COOPER .. Administrator of the estate of Walter J.Meldrum, also known as W.J.Meldrum, deceased. State of California, 89. County of Sacramento. CN THIS 11th day of March, 1920, before me, C.H.S.Bidwell, a Notary Public in and for the County of Sacremento, State of California, personally appeared Claude K. Ccoper, known to me to be the person whose name is subscribed to the foregoing instrument as administrator of the estate of Walter J.Meldrum, also known as W.J.Meldrum, deceased, end he acknowledged to me that he, as said administrator, executed the same. IN WITNESS WHEREOF, I have hereunto set my hand and affixed my official seal, at the County of Sacramento, the day and year in this certificate first above written. (SEAL) C.H. S. BIDWELL Notary Public in and for the County of Sacramento, State of California. Filed for record at the request of La F.A. Dorrington(Supt. Nov .9,1925 at 5 min past 9 U'clock A.W. NO. 1630. CHAS E. MARSH Recorder. of El Dorado County. AROLINE TARD to RUTH CAROLINE CORDER THIS INDENTURE, made the twenty fourth day of October, A.D. one thousand nine hundred and twenty-five, BETWEEN Caroline Ward of Georgetown, El Dorado County, California, the party of the first pert, end Ruth Caroline Corker of Georgetown, El Domedo County, California, the party of the second part, WITNESSETH: That the said party of the first part, in consideration of the sum of one dollar and other good and valuable consideration ---dollars of the United States of Americ to her in hand peid by the said party of the second part, the receipt whereof is hereby acknowledged, does by these presents, grant, bargain, and sell unto the seld party of th

Done in open Court this 27th day of October 1919. (ERBORSED: P1104 Oct 27, 1919 Saunders-Clerk

Stateor California. County of Yolo: SE

W.A.Anderson..Judge of the Superior Court.

BEgi

County.

No. 1163

7

×.

1

.-

I. B.R.Saunders, County Clerk of the County of Yolo, State of California and ex-officio Clark of the Superior Court thereof do hereby certify that I have compared the foregoing copy with the original Decree Settling Pinel Account and Making Distribution in the estate of Susan Gilmore, elso known as Suele N. Pierce, deceased, filed in my office on the 27th day of October 1919 and that the same is a full, true and correct copy of the original and the whole thereof, as the same remains of record in my office. WITHESS my hand and the seal of the Superior Court this 27th day of October 1919. ((SEAL)) H.R.Saunders...Clerk.

1 Filed for record at the request of Chas. A. Swisler, Oct 28th 1919 at 5 min. rest 1-o'clock 0 P. M. Ches.E.Mar Becorder of Elbora

> IN THE SUPERIOR COURT OF THE STATE OF CALIFORNIA IN AND FOR THE COUNTY OF SACRAMEDITO.

In the Motter of the Estate - 01-Walter J. Meldrum, also known as H.J.Meldrum,

Decessed.

No. 8363. Dept. 2 ORDER DIRECTING ADMINISTRATOR TO EXECUTE DEED.

The varified petition of Claude E.Cooper, the coministrator of the estate of the above named decedent proginginat he be authorized and directed to convey to the United States of America certain property described in said petition coming on regularly this day to be heard and it appears, from the proofs adduced that due and legal notice of this hear ing has been given; that said decedent in his lifetime duly executed a contract in writing of America. The convey the real property hereinefter described to the United States and that said United States of America is entitled to a conveyance of the said property hereinafter described upon payment of the sum of fourteen hundred and 00/100 dollars (\$1400.00)

THEREFORE it is hereby ordered and decread that said Claude C.Cooper as administrator of seld estate be and he hereby is authorized and directed upon payments to him soministrator by the United States of America of the sum of fouriern hundred D 1 1075 (\$1400 to execute a proper deed to said United States of America of the following described real property to.wit; Real property situated in the County of ElDoredo, State of California and bounded and particularly described as follows to-wit:

The Northwest one-quarter (NW1) of Section Twenty-nine (29) in Township Ten (10) North of Renge Ten (10) East M.DP & M.

> Dated-October 6th 1919. Judge of the Superior Court.

ENDORSED: Filed Oct 17, 1919 Herry W.Hall..Clerk Bys, P.Dunning--Deputy.

12 March 1. 1. 18


JUN 18 2004

DEPARTMENT OF THE INTERIOR JUL 2 9 1918 4572 The Commissioner, Office of Indian Affairs. Dear Mr. Counissioners I transmit herewith a copy of the opinion of the Bolioitor, on the abstract of title to the MW. 2 Sec. 29, . T. 10 H. ; R. 10 E., M.D.M., which land it is proposed to sequire for the use of the Sacramento-Verona Band of Hor Indiana, Onlifornia. This opinion T be rahm FILED BY S. C. accordance with your request of the 23d instant. Cordially 3 He pelies Inc Logia

Ble 82 4 CONCEPTION OF ÷ في ا έ. M2 632 100 A Walter Cam Maild day of Manch. THIS UNDENTURE, Made the y The the part y of the first part, and Walter James . Weldim ... the part 'f of the second part, WITNESSETH, that the said part y of the first part for and in consideration of the sum \_\_\_ DOLLARS, gold \_ Corres \_\_\_\_\_ of the United States of America, to row \_\_\_\_\_ in hand paid by the soid part T of the focond part, the roceips whereas is hereby acknowledged, here arren is and for the soid part T of the brief of these foresaction describer is a release and for write guild have a fore the soid part to the source of the second purt, and to \_\_\_\_\_\_ for write and basigns forewar, all that certain lot\_\_\_\_\_\_, pice\_\_\_\_ or purced (L of land situate, lying and being in the \_\_\_\_\_\_ County of El Dorado, State of the soil of the second purt, and to \_\_\_\_\_\_ the second basigns forewar, all that certain lot\_\_\_\_\_\_. pice - or pared & of land eiluste, lying and being in the Style of California, and bounded and particularly described, as folloos, to wil: The Northwest quarter & N. W. 47 g. Section Insate - nin Lag Detrois hip Sten Meth North g Mangel Sen Nor Cart. No. 2. 13. 50 2 e signa Together with all and singular the tenements, hereditaments and appurtenances thereunto belonging or in anywise appartaining, and the reversion and reversions, remainder and remainders, rents, issues and profits thereof. and elso the cohile, right, title, realized, property, preservation, claim and down, and whateou as sould not for the sould set property for war for and and and seal the day and sould set for the the sould be first part of the first part of the sould be the sould be for the sould be Sidney Thomas Signed, sealed and delivered in presence of (SBAL) .(PBAL) (SEAL) STATE OF CALIFORNIA. lacia, The second s

18 86 · · สถ -1 . 1500 Se -A LA the states with the states 1.1 . . . 15. thereto incident, appendant and appurtement or therewith usually had and enjoyed, and saled all and singular the tenements, hereditsments, and appurtenances thereun to belongrging, or in anywise appertaining, and the reversion and reversions, and red remainder s mainders, rents rissues and profits thered is the state of the TO. HAVE AND TO HOLD all and singular the said presides together with the appuriant funto the said party of the second part and to his beirs and essigns forever. IN WITNESS THEREOF the said parties of the first part have bereunto set their hands and and the second second 1.1 27 16 seals the Jay and year tirst above written. ::: Louis Caffaro, .... (Seal) 「「「「「「「「」」」」  $2 \sim 10^{-10}$ Francesca Caffaro. . (Seal ) 5. B: Galdo ..... (Seal) W. P. Tanguary..... (Seal) A. Fracinetti.....(Seal) - <u>1</u> т. 1.00 State of California, and the . . . . County of Secremento :: SS On this Fifth day of July in the year one thousand nine hundred and 16, before me, A. G. Zallio, a Robery Public in and for the County of Sacramenic, personally appeared Louis Caffaro, Francesca Caffaro, B. Gaido, W. P. Tanquary A. Fracinetti, known to me to be the persons whose names are bubacribed to the within. instrument and they duly acknowledged to me that they executed the same. 3:520 IN WITNESS WHEREOF, I have bereunto set my hand and affired my official seal the day and year in this certificate first above written. . (| S B A L )) A. G. Zallio....Notary Public, In and for the County of Sacramento, State of California. Commission expires May 19, 1918. Filed for record at the request of Hugh O. Pierce, Sept 18th 1916 at 10 min. past 9o'olock A.N. Char Emarch Recorder of El Dorado County. ÷. ANB RUNCY No.1232 Contraction of the Rachel E. B. Doyle et. vir . - WALTER JAMES HELDRUN State State of September in the year 1916, between Rachel E. B. Doyle ( formerly Rachel E. B. Meldrum) and Fred Doyle her husband, both of the County of Sonoma, State of California, the parter of of the first period Walter James Meldrum, of the County of Sacramento, State of Callfornia the party of the second part, WINTSSETHESE THESE That the said parties of the first A PERSONAL PROPERTY OF A PARTY part, for and in consideration of the sum of ten dollars, gold coin of the United States of America, to them in hand paid by the said party of the second part, the receipt whereof is hereby acknowledged; do by these presents grant, bargain and sell, convey and confirming -關 · · · · · · and the second second 

30 Min Links 議会 -6 A. **建制**。如此 ः । unto the said party offthe second part, and to blathelrs and assigns forever, an individed Tourth (1) for all that certain lot and parcelief land, alluate in the County of Elborate ado, State of Callornia, and bounded and described as follows to site. The Northwest quarte (NW1) of Section Twenty-Hine (29) in Township Tent (10) North of Range Ten (10) East M.D.B.E. TOUETRER (sith all and singular the tenements infertil annuls and appurtenances thereinto belonging or in any issappertaining and the reversionland reversions, remainder and re-the unto the said party of the second part, his heirs dod assigns, forever. IN WITNESS WHEREOFF the said parties of the first part, have berounto set their hands and IN WITHESS THEREOF the said parties of the first part have bereunto set their manas and seals the day mid year first above written. Rachel E.B. Doyle, formerly Rachel E. B. Meldrum. (Seal) State of California, (Seal) ....(9eal) County of Sonoma s: SS On this 15th day of September in the year one thousand nine hundred and sixteen (1916) before me, William H. Pool a Notary Public in and for sail County ×. of Sonoma, State of California, personally appeared Rachol E. B. Doyle formerly Rachel E.B. ALL PLAN Meldrum, and Fred Doyle, her husband, known to me to be the persons whose names are subsoribed to the within im trument and they acknowledged that they executed the same. No. WITNESS my hand and official seal. ((SEAL)) William R. Pool......Notary Public In and for the County of Sonoma, State of California. 198 - 121) 1 not a set of a \$. W. ų Filed for record at the request of W. J. Meldrum, Sept 25th 1916 at 4 min. past 9-o'clock E 1.11. **i**.... Char Emarst. Recorder of El Dorado County COMPARED \*\*\*\*\*\* No. 1233 Joseph W. G. Shannon 5 F 1 1.4 Walter James Meldrum .÷ 1 费 day of September in the year 1916 between Joseph W. G. Shannon of the County of Sacramento State of California, the party of the first part and Walter James Meldrum of the same place partifior and incooksideration of the sum of ten dollars, gold coin of the United States あるの of America to him in hand paid by the said party of the second part, the receipt othereof. Is hereby stinopledged, does by these presents grants bargel n and sell, convey and confirm. unto the sald party of the second part and to his beirs and assigns forever, an undivided one-fourin (1) off all that certain lot and parcel of land-situate in the County of ElDopado 



TOGETHER with all and singular the tenements, hereditaments and appurtenances thereunto belonging or in anywise appertaiping, and the reversion and reversions, remainder and remainders, rents, issues and profits thereof.

TO BAVE AND TO HOLD all and singular the said premises together with the appurtenances, unto the said party of the second part, his heirs and assigns forever.

IN WITHESS WHEREOF, the said party of the first part has berewnly set his band and seal the day and year first above written.

Meldrum D. Shannon....(Seal)

State of California,

30 38 Bl 86

 $\odot$ 

County of Sacramento r: SS On this 23rd day of September in the year 1916 before ! me, C. H. S. Bidwell, a Notary Public in and for said County, personally appeared Meldrum f D. Shannon known to me to be the person whose name is subscribed to the within instrument and he acknowledged to me that he executed the same.

((SEAL)) C. H. S. Bidwell......Notary Public In and for the County of Sacramenic, State of California.

Filed for record at the request of W. J. Weldrum, Sept 25th, 1918 at 8 min. past 9-o'clock A.W. Recorder of El Dorado County

#### 

Ho:1218

COlicon Ser

Alma C. Tagimeier et. al.

Thei ceriain loi in the said City of Placerville, County of El Dorado, State of Celiforni designated, mark-d, numbered and described on the official map and in the field notes of the official survey of the said City of Placerville now on file and of record in the offici at the official survey of the said City of Placerville now on file and of record in the official

W. B. M. HOLE ..... de Rloy: COMPARES No.486864 No. 100 ... State of California-Controller's Office, Sacramento • ; I HEREBY CERTIFY that I have on this 1.8 th day of December 1919. received from 6. 6 Corper Administrator. a cortificate of the Auditor of the County of El Dorado, data December 17 = 1919 containing said Auditor's estimate of the amount (and specifying the several amounts thereof) necessary to be poid to redeem that certain tract of land rituals in said County, assessed to David Mildurm and described us follows, lo-unit: N. 91. 4 9 Section 29, Inp 10, Ran ale or Dur 10= 2790'2 83 2968 ame 15, for delinquent jaxes for the year. 191. Said land was sold to th and was properly indersed, with the receipt of the Treasurer of said County, for the amount, the person desiring to redeem , specified in said estimate, in accordance with section three thousand eight hundred and 10-wil: \$ 37.25 seventeen of the Political Code. n J. Chumbers. Controller. 1277 Recorded at the request of O. H 111101 al 10 minutes past -/- o'clock P. M., in Record page\_ 1/20 ... Records of the County of El Dorado. 3 En lais Recorder. Bu Deputy Recorder COMPARE 493463 No. 123 No State of California-Controller's Office, Sacramento I HEREBY CERTIFY that I have on this 24th day of January icenoed from a certificate of the Auditor of the County of El Dorado, doled Kinn bry 244 1920 containing said Auditor's estimate of nt (and specifying the several amounts thereof) necharry to be paid to redeem that explain tract of land situate in said necessed to first the several amounts thereof) necharry to be paid to redeem that explain tract of land situate in said 2 8.11'4 g Lev 17 Inp 10, R. 13. 4 54 g Jew 8 JAp 10, R. 13. 4 7 Mily a Jew 8 Jap 10, R. 13. 14 D. 64: 162 g S. 9.14 g Jac 8. Inp 10 R. 10, 14 0. 54: 162 g S. 9.14 g Jac 8. Inp 10 R. 10, 14 0 Japp R. 12 Ez g & Ely g & Ely g A. 18'4 g Jeo 6 nd described as follo N.W'14 15 P.N. W 44 L 1.84 :/Y. h \$ 6 44 N. 64 9 0 70 acres 12, Sed 5 Su 9 (Vange 1.2 Into Jen w 25 - 1917 .... For delinquent taxes for the year. 1916. eold to the Sinte Said certificate was mude on the application of \_\_\_\_\_ (N. Beal. and was properly indersed, with the receipt of the Treasurer of said County, for the amount, the person desiring to rederm. 98.46 In-wil: 8 . specified in said estimate, in with section three thousand eight hundred and seventeen of the Political Code. ( loin ; Controller. N.B. Recorded at the request of. ,1020 ut 20 minutes past - 9 o' clock W.M., in Record 1/20, Records of the County of El Durudo. Recorder. By. Deputy Recorder.

# <u>Exhibit B</u> <u>Resolution 2003-12 concerning</u> <u>Limited Waiver of Immunity from Suit</u>

ć )

)

)

# JUN 18 2004



#### SHINGLE SPRINGS RANCHERIA P.O. BOX 1340; SHINGLE SPRINGS, CA 95682 (530) 676-8010; FAX (530) 676-8033

#### **RESOLUTION 2003-12**

#### SUBJECT: RESOLUTION OF THE SHINGLE SPRINGS BAND OF MIWOK INDIANS REGARDING LIMITED WAIVER OF SOVEREIGN IMMUNITY FOR ECONOMIC DEVELOPMENT PURPOSES IN CONNECTIONWITH DEVELOPMENT AND MANAGEMENT OF A CLASS II AND III GAMING FACILITY/ENTERPRISE

WHEREAS, the Shingle Springs Band of Miwok Indians ("the Tribe") is a federally recognized Indian tribe eligible for the special programs and services provided by the United States to Indians because of their status as Indians and is recognized as possessing powers of self-government.

WHEREAS, the Tribal Council of the Tribe is the duly recognized governing body of the Tribe and is a federally-recognized Indian tribal government.

WHEREAS, the United States government holds lands in the State of California in trust for the benefit of the Tribe ("Tribal Lands") over which the Tribe possesses sovereign governmental powers.

WHEREAS, in compliance with the Indian Gaming Regulatory Act of 1988, P.L. 100-497, 25 U.S.C. §2701 <u>et seq</u>. as it may from time to time be amended ("IGRA"), the Tribal Council of the Tribe has enacted a tribal ordinance regulating the operation of gaming activities on Tribal Lands (hereinafter referred to as the "Tribal Gaming Ordinance"), creating the Shingle Springs Tribal Gaming Commission, and authorizing Class II Gaming and Class III Gaming on its Tribal Lands subject to the provisions of IGRA, the Tribal Gaming Ordinance and a Tribal-State Compact.

WHEREAS, the Tribe is committed to the use of gaming activities to provide employment and improve the social, economic, education, and health needs of its members; to increase the revenues of the Tribe; and to enhance the Tribe's economic self-sufficiency and self-determination.

WHEREAS, the Tribe presently lacks the resources to develop and operate a gaming facility and enterprise on its own and desires to retain the services of a developer and manager with knowledge and experience in the gaming industry to secure financing, develop, manage and operate Class II Gaming and Class III Gaming facilities and related resort facilities on Tribal Lands.

WHEREAS, Lakes KAR-Shingle Springs, LLC (the "LKAR") has represented to the Tribe that it has the managerial and financial capacity to provide and secure financing for the funds necessary to develop and construct the Facility, as defined in

the Amended Memorandum of Agreement between the Tribe and LKAR, and to commence operation of the Enterprise, as defined in the Amended Memorandum Agreement; and LKAR agrees to assist the Tribe in obtaining the capital investment necessary to the development of the Facility, and to provide the management expertise necessary to the conducting of successful tribal gaming operations.

WHEREAS, Kean Argovitz Resorts-Shingle Springs, L.L.C. ("KARSS") entered into Development and Management Agreements with the Tribe dated June 11, 1999 (the "Development and Management Agreements"), pursuant to which KARSS was to develop and manage a gaming facility and certain related facilities to be owned by the Tribe.

WHEREAS, Lakes Gaming, Inc. and KARSS entered into an agreement under which they agreed to form LKAR to assume the rights and obligations of KARSS in connection with the Development and Management Agreements; and KARSS has assigned its rights and obligations under the Development and Management Agreements to LKAR pursuant to the terms of an Assignment and Assumption Agreement.

WHEREAS, pursuant to the May 5, 2000 Memorandum of Agreement entered into between the Tribe and LKAR and by tribal resolution, the Tribe consented to the assignment of the Development and Management Agreements to LKAR by KARSS and to LKAR's assumption of KARSS's rights and responsibilities under the Development and Management Agreements.

WHEREAS, in accordance with LKAR's desire to assume KARSS's rights and responsibilities under the Development and Management Agreements, the Tribe granted to LKAR the exclusive right and obligation to develop, manage, operate and maintain the Facility and Enterprise as described in the May 5, 2000Memorandum of Agreement between the Tribe and LKAR, which was intended by the parties thereto to supercede and replace the Development and Management Agreements.

WHEREAS, the May 5, 2000 Memorandum of Agreement (and related exhibit documents) between the Tribe and LKAR has been submitted for approval to the National Indian Gaming Commission ("NIGC"); NIGC staff have reviewed the May 5, 2000 Memorandum of Agreement (and related exhibit documents) and requested certain modifications to the documents prior to issuing NIGC approval; and the Tribe and LKAR have made such modifications to the documents as they deem necessary, and, accordingly, the Tribe and LKAR desire to enter into a First Amended and Restated Memorandum of Agreement Regarding Gaming Development and Management Agreement ("Amended Memorandum Agreement") and related Transaction Documents in connection with the Project, which is intended by the parties thereto to supercede and replace the May 5, 2000 Memorandum of Agreement.

WHEREAS, the Amended Memorandum Agreement shall become effective when all the necessary approvals listed in Section 3.19 of the Amended Memorandum Agreement are received (the "Effective Date") and shall continue for a term of

from the Commencement Date, or as otherwise provided in the Amended Memorandum Agreement.

WHEREAS, the Amended Memorandum Agreement is entered into pursuant to the Indian Gaming Regulatory Act of 1988, P.L. 100-497, 25 U.S.C. §2701 <u>et seq.</u> (the "IGRA") as that statute may be amended. All gaming conducted at the Facility will at all times comply with the IGRA, applicable tribal law and the Tribal-State Compact.

WHEREAS, the Tribe and LKAR desire that the preliminary Facility design and development work will be done (but not the Facility construction or Enterprise operation) so the Project may proceed prior to receipt of necessary regulatory approvals.

WHEREAS, the Tribe and LKAR desire to take all steps which are lawful and reasonably possible prior to the receipt of the necessary regulatory approvals: (i) to obtain a preliminary commitment for financing of the Facility, (ii) to select and develop the site for the Facility, (iii) to design the Facility, and (iv) to enter into contracts to construct and equip the Facility so that the Facility can be opened to the public as soon as possible after the receipt of all necessary regulatory approvals.

WHEREAS, LKAR desires to advance to the Tribe, subject to the terms and conditions of the Transition Loan described in the Amended Memorandum Agreement, sums sufficient to finance performance of the preliminary development work described immediately above and for other purposes, and the Tribe and LKAR have agreed that all sums previously advanced to the Tribe by KARSS under the superceded Development and Management Agreements, and by LKAR under the superceded May 5, 2000 Memorandum of Agreement, shall constitute advances by LKAR to the Tribe under the terms of the Transition Loan described in the Amended Memorandum Agreement.

WHEREAS, under Section 2.3 of the Amended Memorandum Agreement, the Tribe has agreed to repay the sums advanced to it by the LKAR pursuant to the Transition Loan, as evidenced by a Interim Promissory Note in the form agreed to by the Tribe and LKAR; and has agreed: (i) to repay the Transition Loan

[(ii) subject to the terms of Section 2.3 of the Amended Memorandum Agreement to grant to LKAR(

64

64

ſ

Jand (iv) to enter into a limited, transactional waiver of sovereign immunity and consent to jurisdiction and arbitration as to LKAR and in connection with the Transition Loan. 64

64

WHEREAS, the Tribe has selected LKAR, and the LKAR has agreed, to assist the Tribe in obtaining permanent financing for the Project, subject to the terms and conditions of the Facility Loan described in the Amended Memorandum Agreement, and to furnish technical experience and expertise for the development and design of the Project, and for contracting for the construction, furnishing and equipping of the Project.

WHEREAS, under Section 2.5 of the Amended Memorandum Agreement, the Tribe has agreed to repay the sums advanced to it by a Lender pursuant to the Facility Loan, or made directly to it by LKAR at LKAR's discretion, for the Costs of Construction and Initial Costs of Operations of the Facility and Enterprise, as evidenced by a Facility Note in the form agreed to by the Tribe and the Lender or by the Tribe and LKAR, as applicable; and has agreed:

*j*(iv) to enter into a limited, transactional waiver of sovereign immunity and consent to jurisdiction and arbitration as to Lender or LKAR and in connection with the Facility Loan.

WHEREAS, the Tribe understands that its agreement to adopt an enforceable Resolution of Limited Waiver is a material inducement to LKAR's execution of the Amended Memorandum Agreement, to LKAR making advances of sums under the Transition Loan, and is a condition precedent to any of the respective obligations of the parties under the Amended Memorandum Agreement.

WHEREAS, the Tribe understands that its agreement to adopt an enforceable Resolution of Limited Waiver is a material inducement to a Lender or LKAR in advancing the sums under the Facility Loan for Costs of Construction and Initial Costs of Operation of the Facility and Enterprise and is a condition precedent to any of the respective obligations of the parties under the Facility Loan.

WHEREAS, the Tribe believes that the development contemplated by the Amended Memorandum Agreement is important to the governance and long-term economic development of the Tribe, and desires to fulfill its obligations under the

terms of the Amended Memorandum Agreement to ensure the success of the Project.

#### NOW THEREFORE, BE IT RESOLVED,

1. That the capitalized terms used herein and not defined shall have the meanings given them in the Amended Memorandum Agreement between the Tribe and LKAR.

2. That the Tribe hereby expressly waives subject to the terms and provisions hereof its sovereign immunity from suit in connection with the (i) Amended Memorandum Agreement; (ii) the Facility Loan, Facility Note and related documentation; (iii) the Land Acquisition Loan, Land Acquisition Note and related documentation; (iv) the Transition Loan, Interim Promissory Note and related documentation; (v) the Operating Note, (vi) the Security Agreement and related documentation; (vii) the Dominion Account Agreement and related documentation; (viii) the Dominion Account Agreement and related documentation; (viii) the Dominion Account Agreement and related documentation; (viii) the Indemnity Agreement; (ix) any other Transaction Documents; and (ix) the Access Land Transfer Agreement, Non-Gaming Land Transfer Agreement and related documentation (collectively "the Immunity Agreements"); this waiver of sovereign immunity is for the sole and limited purpose of permitting and compelling arbitration as to any issue or dispute under the Immunity Agreements and enforcing any arbitration award or judgment arising thereunder, as described herein and is further limited by the restrictions described herein.

3. That the Tribe consents to be sued in the United States District Court for the district in which the Facility is located, the United States Court of Appeals having jurisdiction over the applicable District Court, and the United States Supreme Court, for the purpose of compelling arbitration or enforcing any arbitration award, orders or judgment arising out of the Immunity Agreements. If the United States District Court lacks jurisdiction, the Tribe consents to be sued in a court of competent jurisdiction for the same limited purpose. Subject to the provisions of this Resolution, the Tribe waives any requirement of exhaustion of tribal remedies and expressly authorizes any governmental authorities having the right and duty under applicable law to take any action authorized or ordered by a court having jurisdiction hereunder, provided that at all times the liability of the Tribe under any judgment shall be always Limited Recourse as defined herein.

4. That the Tribe agrees and consents that all disputes, controversies or claims arising out of or relating to the Immunity Agreements, or the parties' obligations and rights thereunder, shall be settled by binding arbitration in accordance with the terms set forth in the Immunity Agreements, including without limitation, that such arbitration shall be governed by the Commercial Arbitration Rules of the American Arbitration Association in effect on the date the demand for arbitration is made, and the Federal Arbitration Act; except that: (a) the question whether or not a dispute is arbitrators, such question shall be a matter for binding arbitration by the arbitrators, such question shall not be determined by any court and,

in determining any such arbitration question, all doubts shall be resolved in favor of arbitrability; and (b) discovery shall be permitted in accordance with the Federal Rules of Civil Procedure, subject to supervision as to scope and appropriateness by the arbitrators.

5. That the Tribe's waiver of immunity from suit is specifically limited to the following actions and judicial remedies:

The enforcement of an award of compensatory damages by (a) Damages. arbitration; provided that the award of any arbitrator and/or court must be Limited Recourse, and no arbitrator or court shall have authority or jurisdiction to order execution against any assets or revenues of the Tribe except (i) undistributed and future Net Total Revenues of the Enterprise or any other Tribal Gaming Enterprise; (ii) any security interests or liens in any Furnishings and Equipment purchased with Facility Loan or Transition Loan proceeds or other purchase money agreements; (iii) the security interests in the undistributed and future Net Total Revenues of the Enterprise pursuant to the Dominion Account Agreement; and (iv) as to any mortgages or deeds of trust on the Acquired Tribal Lands prior to their transfer into trust. "Limited Recourse" means that the Facility Loan, Land Acquisition Loan and Transition Loan advances, and all liabilities of the Tribe related to the Immunity Agreements and their applicable documentation, the Facility, or the Enterprise contemplated by the Amended Memorandum Agreement, and any related awards, judgments or decrees, shall be payable solely out of undistributed and future Net Total Revenues of the Enterprise or any other Tribal Gaming Enterprise, and shall be a limited recourse obligation of the Tribe, with no recourse to tribal assets other than such undistributed and future Net Total Revenues and the following: (i) any security interests or liens in any Furnishings and Equipment purchased with Facility Loan or Transition Loan proceeds or other purchase money agreements pursuant to the Security Agreement; (ii) the security interests in the undistributed and future Net Total Revenues of the Enterprise or any other Tribal Gaming Enterprise pursuant to the Dominion Account Agreement; and (iii) as to any mortgages or deeds of trust on the Acquired Tribal Lands prior to their transfer into trust . In no event shall LKAR, or any Lender or other claimant, have recourse to (a) the physical property of the Facility (other than Furnishings and Equipment subject to the security interest securing the Facility Loan, Transition Loan or other purchase money agreements), (b) Net Total Revenue distributions already made to the Tribe in accordance with the Amended Memorandum Agreement (c) assets of the Tribe purchased with its Net Total Revenue distributions, or (d) any other asset of the Tribe (other than the mortgage or deed of trust on the Acquired Tribal Lands prior to its transfer into trust, and such undistributed and future Net Total Revenues of the Enterprise or any other Tribal Gaming Enterprise). In no instance shall any enforcement of any kind whatsoever be allowed against any assets of the Tribe other than the limited assets of the Tribe specified herein.

(b) <u>Consents and Approvals.</u> The enforcement of a determination by an arbitrator that the Tribe's consent or approval has been unreasonably withheld

contrary to the terms of the Immunity Agreements.

Injunctive Relief and Specific Performance. The enforcement of a (c) determination by an arbitrator that prohibits the Tribe from taking any action that would prevent the other party from performing its obligations pursuant to the terms of the Immunity Agreements, or that requires the Tribe to specifically perform any obligation under the Immunity Agreements; provided, however, that any injunction against the Tribe shall be Limited Recourse; shall not mandate, preclude or affect payment of any funds of the Tribe other than undistributed and future Net Total Revenues of the Enterprise; and shall not relate to any asset of the Tribe other than as described above under Paragraph 5(a). The Tribe specifically acknowledges and agrees that in connection with any arbitration award, order or judgment, the arbitrators shall be empowered to take the actions and enforce the judicial remedies described in this Paragraph 5; provided however, that although the arbitrators may award damages in the event the Tribe or the Tribal Gaming Commission do not comply with the award, the arbitrators may not require the Tribe or Tribal Gaming Commission to take or modify any governmental legislative decision or action which the arbitrators have determined has resulted in the dispute between the parties and is contrary to the parties rights, liabilities or obligations under the Amended Memorandum Agreement, the Facility Note, the Land Acquisition Note, the Interim Promissory Note, the Operating Note, or any other Transaction Document (Specific Performance Restriction"). Provided further, that: (a) should the arbitrators determine that there has been an intentional bad faith violation of a party's rights under any Immunity Agreement by the Tribe or the Tribal Gaming Commission, and if the Tribe or the Tribal Gaming Commission do not reverse such intentional bad faith violation through governmental legislative decision or action within thirty (30) days after being notified by the arbitrators of such determination, then the arbitrators shall

(d) <u>Action to Compel Arbitration and for Enforcement</u>. An action to compel arbitration pursuant to the applicable provisions in the Immunity Agreements; or, to the extent not previously set forth above herein, an action seeking enforcement of the Immunity Agreements, and to the extent applicable, foreclosure of any of the Immunity Agreements.

64

6. That the Tribe hereby appoints the Chairman and Secretary of the Tribal Council individually as its agents for service of process under or relating to the Immunity Agreements, and agrees that service by hand or by certified mail, return receipt requested, shall be effective as service upon the Tribe for all purposes under or relating to the Immunity Agreements if served upon such agents.

7 That the Tribe shall execute any documents and do any other undertaking as shall from time to time be reasonably required to carry out and make effective the terms and provisions of the Immunity Agreements; and that the Chairman and Secretary of the Tribal Council are hereby authorized to execute and deliver the Immunity Agreements and any related documents on terms substantially the same as the draft documents presented to the Tribal Council on this date, together with such changes and modifications thereto as the Chairman and Secretary deem in their discretion to be in the best interests of the Tribe.

8. That the Tribe hereby authorizes: (a) the submission of the Immunity Agreements to the NIGC for review and approval, as necessary under federal law; (b) the Chairman to provide to the NIGC on behalf of the Tribe any documentation requested by the NIGC in conjunction with its process of reviewing and approving, as necessary, the Immunity Agreements, and any other information needed to expedite review and approval, and (c) the Chairman to act on behalf of the Tribe concerning the submission to the NIGC.

9. That this Resolution of Limited Waiver shall take effect immediately, and shall supercede and replace the January 10, 2000 resolution adopted by the Tribe in connection with the same matter, which resolution is specifically revoked.

10. That the Tribe agrees it will take all lawful steps necessary to preserve the effective terms of this Resolution of Limited Waiver in the event of future changes in its legal status or governance.

#### CERTIFICATION

Chairman

Attest

Tribal Secretary

# <u>Exhibit C</u> <u>Resolution 2004-23 concerning</u> <u>the June 16, 2004 Amendment and June 16, 2004 Consent Agreement</u>

Ì



# Shingle Springs Rancheria

P.O. Box 1340; Shingle Springs, CA 95682 (530) 676-8010; Fax (530) 676-8033

#### **RESOLUTION 2004 – 23**

# <u>SUBJECT</u>: RESOLUTION OF THE SHINGLE SPRINGS BAND OF MIWOK INDIANS REGARDING LIMITED WAIVER OF SOVEREIGN IMMUNITY RELATING TO, AND APPROVAL OF, THE JUNE 16, 2004 AMENDMENT TO THE BAND'S AMENDED MEMORANDUM AGREEMENT, AND JUNE 16, 2004 CONSENT AGREEMENT WITH LKAR IN CONNECTION WITH ITS FOOTHILLS OAKS GAMING FACILITY/ENTERPRISE

WHEREAS, the Shingle Springs Band of Miwok Indians ("the Tribe") is a federally recognized Indian tribe eligible for the special programs and services provided by the United States to Indians because of their status as Indians and is recognized as possessing powers of self-government.

WHEREAS, the Tribal Council of the Tribe is the duly recognized governing body of the Tribe and is a federally-recognized Indian tribal government.

WHEREAS, in compliance with the Indian Gaming Regulatory Act of 1988 ("IGRA"), P.L. 100-497, 25 U.S.C. 2701 <u>et seq.</u> as it may from time to time be amended, the Tribal Council of the Tribe has enacted a tribal ordinance regulating the operation of gaming activities on Tribal Lands (hereinafter referred to as the "Tribal Gaming Ordinance"), creating the Shingle Springs Tribal Gaming Commission, and authorizing Class II Gaming and Class III Gaming on its Tribal Lands subject to the provisions of the Tribal Gaming Ordinance and a Tribal-State Compact.

WHEREAS, the Tribe is committed to the use of gaming activities to provide employment and improve the social, economic, education, and health needs of its members; to increase the revenues of the Tribe; and to enhance the Tribe's economic self-sufficiency and self-determination.

WHEREAS, the Tribe presently lacks the resources to develop and operate a gaming facility and enterprise on its own and desires to retain the services of a developer and manager with knowledge and experience in the gaming industry to secure financing, develop, manage and operate a Class II Gaming and Class III Gaming facility and related resort facilities located on its Indian lands in accordance with the Indian Gaming Regulatory Act of 1988, as amended.

WHEREAS, to assist with the financing, development, management and operations of its planned gaming facility and enterprise (the "Project"), the Tribe and LKAR agreed to enter into their First Amended and Restated Memorandum of Agreement Regarding Gaming Development and Management Agreement ("October 13,

2003 Amended Memorandum Agreement") and related Transaction Documents all dated October 13, 2003 in connection with the Project.

WHEREAS, the October 13, 2003 Amended Memorandum Agreement (and related exhibits and Transaction Documents) between the Tribe and LKAR has been submitted for approval to the National Indian Gaming Commission ("NIGC"); NIGC staff have reviewed the October 13, 2003 Amended Memorandum Agreement (and related exhibits and Transaction Documents) and have requested certain modifications to the documents prior to issuing NIGC approval; the Tribe and LKAR have made such modifications to the documents as they have deemed necessary, including those made necessary by changes in circumstances related to increased financing needs of the Project; and, in this regard, the Tribe and LKAR have negotiated and desire to enter into the June 16, 2004 Amendment to the October 13, 2003 Amended Memorandum Agreement.

WHEREAS, under Section 10.1 of the October 13, 2003 Amended Memorandum Agreement, as amended, the Tribe reserved the right to assign its rights and responsibilities under the agreement to a tribally chartered entity or an IRA section 17 corporation that it wholly owns and controls.

WHEREAS, since signing the October 13, 2003 Amended Memorandum Agreement, the Tribe has adopted tribal Resolution 2004-18 establishing the Foothills Oaks Tribal Gaming Authority ("Authority") and related Ordinance whereby the Tribe has delegated to the Authority all governmental and proprietary powers and rights of the Tribe over the development, construction, operation, promotion and financing of the Project.

WHEREAS, the NIGC has requested that the Tribe and LKAR each consent to the delegation of the Tribe's governmental and proprietary powers and rights to the Authority in connection with the Project as described in tribal Resolution 2004-18 and related Ordinance before approving the October 13, 2003 Amended Memorandum Agreement, as amended; both the Tribe and LKAR desire to make such consent in order to expedite NIGC's approval, and have, accordingly, prepared and desire to enter into a Consent Agreement in connection with the matter.

WHEREAS, the Tribe and LKAR desire to update the Interim Promissory Note to reflect changes to the terms of and advances made to the Tribe by LKAR under the Transition Loan since the parties executed the October 13, 2003 Amended Memorandum Agreement and related Transaction Documents.

WHEREAS, the Tribe understands that its agreement to adopt an enforceable Resolution of Limited Waiver is a material inducement to LKAR's execution of the June 16, 2004 Amendment to Amended Memorandum Agreement, the updated Interim Promissory Note, and the June 16, 2004 Consent Agreement, and to LKAR making

()

advances of sums under the Transition Loan, and is a condition precedent to any of the respective obligations of the parties under the October 13, 2004 Amended Memorandum Agreement.

WHEREAS, the Tribe believes that the development contemplated by the October 13, 2004 Amended Memorandum Agreement is important to the governance and long-term economic development of the Tribe, and desires to fulfill its obligations under the terms of the October 13, 2003 Amended Memorandum Agreement and under IGRA to ensure the success of the Project.

#### NOW THEREFORE, BE IT RESOLVED,

1. That the capitalized terms used herein and not defined shall have the meanings given them in the October 13, 2003 Amended Memorandum Agreement between the Tribe and LKAR.

2. That, subject to the terms and provisions hereof, the Tribe hereby expressly approves, and agrees to waive its sovereign immunity from suit in connection with, the following: (i) Amended Memorandum Agreement dated October 13, 2003 (10/13/03 revision); (ii) Scope of Project Description Sheet (10/13/03 revision); (iii) the Facility Loan, Facility Note dated October 13, 2003 (10/13/03 revision) and related documentation; (iv) the Land Acquisition Loan, Land Acquisition Note dated October 13, 2003 (10/13/03 revision) and related documentation; (v) the Transition Loan, Interim Promissory Note dated October 13, 2003 (10/13/03 revision) and related documentation; (vi) the Operating Note dated October 13, 2003 (10/13/03 revision), (vii) the Security Agreement dated October 13, 2003 (10/13/03 revision) and related documentation; (viii) the Dominion Account Agreement dated October 13, 2003 (10/13/03 revision) and related documentation; (ix) any other Transaction Documents; (x) the Access Land Transfer Agreement dated October 13, 2003 (10/13/03 revision) and related documentation; (xi) Non-Gaming Land Transfer Agreement dated October 13, 2003 (10/13/03 revision) and related documentation; (xii) the June 16, 2004 Amendment to Amended Memorandum Agreement (06/16/04 revision); (xiii) the updated Interim Promissory Note dated June 16, 2004 (06/16/04 revision) in the principal amount of (hereinafter referred to as the "Interim Promissory Note"), and (xiv) the June 16, 2004 Consent Agreement (06/16/04#2) (collectively herein "the Amended Immunity Agreements") and confirms its express approval of the waiver of its sovereign immunity in connection with the October 13, 2003 Amended Memorandum Agreement and the other Immunity Agreements as described in tribal Resolution 2003-12. The Tribe's waiver of sovereign immunity is for the sole and limited purpose of permitting and compelling arbitration as to any issue or dispute under the Amended Immunity Agreements and enforcing any arbitration award or judgment arising thereunder, as described herein and is further limited by the restrictions described herein.

64

3. That the Tribe consents to be sued in the United States District Court for the district in which the Facility is located, the United States Court of Appeals having jurisdiction over the applicable District Court, and the United States Supreme Court, for the purpose of compelling arbitration or enforcing any arbitration award, orders or judgment arising out of the Amended Immunity Agreements. If the United States District Court lacks jurisdiction, the Tribe consents to be sued in a court of competent jurisdiction for the same limited purpose. Subject to the provisions of this Resolution, the Tribe waives any requirement of exhaustion of tribal remedies and expressly authorizes any governmental authorities having the right and duty under applicable law to take any action authorized or ordered by a court having jurisdiction hereunder, provided that at all times the liability of the Tribe under any judgment shall be always Limited Recourse as defined herein.

4. That the Tribe agrees and consents that all disputes, controversies or claims arising out of or relating to the Amended Immunity Agreements, or the parties' obligations and rights thereunder, shall be settled by binding arbitration in accordance with the terms set forth in the Amended Immunity Agreements, including without limitation, that such arbitration shall be governed by the Commercial Arbitration Rules of the American Arbitration Association in effect on the date the demand for arbitration is made, and the Federal Arbitration Act; except that: (a) the question whether or not a dispute is arbitrable under the Amended Immunity Agreements shall be a matter for binding arbitration by the arbitrators, such question shall not be determined by any court and, in determining any such arbitration question, all doubts shall be resolved in favor of arbitrability; and (b) discovery shall be permitted in accordance with the Federal Rules of Civil Procedure, subject to supervision as to scope and appropriateness by the arbitrators.

5. That the Tribe's waiver of immunity from suit is specifically limited to the following actions and judicial remedies:

(a) <u>Damages.</u> The enforcement of an award of compensatory damages by arbitration; provided that the award of any arbitrator and/or court must be Limited Recourse, and no arbitrator or court shall have authority or jurisdiction to order execution against any assets or revenues of the Tribe except (i) undistributed and future Net Total Revenues of the Enterprise or any other Tribal Gaming Enterprise; (ii) any security interests or liens in any Furnishings and Equipment purchased with Facility Loan or Transition Loan proceeds or other purchase money agreements; (iii) the security interests in the undistributed and future Net Total Revenues of the Enterprise pursuant to the Dominion Account Agreement; and (iv) as to any mortgages or deeds of trust on the Acquired Tribal Lands prior to their transfer into trust. "Limited Recourse" means that the Facility Loan, Land Acquisition Loan and Transition Loan advances, and all liabilities of the Tribe related to the Amended

Single Springs Resolution 06/16/04 revision #2

+

Immunity Agreements and their applicable documentation, the Facility, or the Enterprise contemplated by the Amended Memorandum Agreement, and any related awards, judgments or decrees, shall be payable solely out of undistributed and future Net Total Revenues of the Enterprise or any other Tribal Gaming Enterprise, and shall be a limited recourse obligation of the Tribe, with no recourse to tribal assets other than such undistributed and future Net Total Revenues and the following: (i) any security interests or liens in any Furnishings and Equipment purchased with Facility Loan or Transition Loan proceeds or other purchase money agreements pursuant to the Security Agreement; (ii) the security interests in the undistributed and future Net Total Revenues of the Enterprise or any other Tribal Gaming Enterprise pursuant to the Dominion Account Agreement; and (iii) as to any mortgages or deeds of trust on the Acquired Tribal Lands prior to their transfer into trust . In no event shall LKAR, or any Lender or other claimant, have recourse to (a) the physical property of the Facility (other than Furnishings and Equipment subject to the security interest securing the Facility Loan, Transition Loan or other purchase money agreements), (b) Net Total Revenue distributions already made to the Tribe in accordance with the Amended Memorandum Agreement (c) assets of the Tribe purchased with its Net Total Revenue distributions, or (d) any other asset of the Tribe (other than the mortgage or deed of trust on the Acquired Tribal Lands prior to its transfer into trust, and such undistributed and future Net Total Revenues of the Enterprise or any other Tribal Gaming Enterprise). In no instance shall any enforcement of any kind whatsoever be allowed against any assets of the Tribe other than the limited assets of the Tribe specified herein.

(b) <u>Consents and Approvals.</u> The enforcement of a determination by an arbitrator that the Tribe's consent or approval has been unreasonably withheld contrary to the terms of the Amended Immunity Agreements.

Injunctive Relief and Specific Performance. The enforcement of a (c) determination by an arbitrator that prohibits the Tribe from taking any action that would prevent the other party from performing its obligations pursuant to the terms of the Amended Immunity Agreements, or that requires the Tribe to specifically perform any obligation under the Amended Immunity Agreements; provided, however, that any injunction against the Tribe shall be Limited Recourse; shall not mandate, preclude or affect payment of any funds of the Tribe other than undistributed and future Net Total Revenues of the Enterprise; and shall not relate to any asset of the Tribe other than as described above under Paragraph 5(a). The Tribe specifically acknowledges and agrees that in connection with any arbitration award, order or judgment, the arbitrators shall be empowered to take the actions and enforce the judicial remedies described in this Paragraph 5; provided however, that although the arbitrators may award damages in the event the Tribe or the Tribal Gaming Commission do not comply with the award, the arbitrators may not require the Tribe or Tribal Gaming Commission to take or modify any governmental legislative

decision or action which the arbitrators have determined has resulted in the dispute between the parties and is contrary to the parties rights, liabilities or obligations under the Amended Memorandum Agreement, as amended, the Facility Note, the Land Acquisition Note, the Interim Promissory Note, the Operating Note, or any other Transaction Document (Specific Performance Restriction"). Provided further, that: (a) should the arbitrators determine that there has been an intentional bad faith violation of a party's rights under any Amended Immunity Agreement by the Tribe or the Tribal Gaming Commission, and if the Tribe or the Tribal Gaming Commission do not reverse such intentional bad faith violation through governmental legislative decision or action within thirty (30) days after the being notified by the arbitrators of such determination, then the arbitrators shall/

64

(d) <u>Action to Compel Arbitration and for Enforcement</u>. An action to compel arbitration pursuant to the applicable provisions in the Amended Immunity Agreements; or, to the extent not previously set forth above herein, an action seeking enforcement of the Amended Immunity Agreements, and to the extent applicable, foreclosure of any of the Amended Immunity Agreements.

6. That the Tribe hereby appoints the Chairman and Secretary of the Tribal Council individually as its agents for service of process under or relating to the Amended Immunity Agreements, and agrees that service by hand or by certified mail, return receipt requested, shall be effective as service upon the Tribe for all purposes under or relating to the Amended Immunity Agreements if served upon such agents.

7. That the Tribe shall execute any documents and do any other undertaking as shall from time to time be reasonably required to carry out and make effective the terms and provisions of the Amended Immunity Agreements; and that the Chairman and one other official of the Tribal Council are hereby authorized to execute the Amended Immunity Agreements described herein, and the Chairman is authorized to deliver the Amended Immunity Agreements and any related documents as necessary; and the Tribal Council approves, adopts and ratifies as properly executed the October 13, 2003 Amended Memorandum Agreement and related Transaction Documents of same date

8. That the Tribe hereby authorizes: (a) the submission of the Amended Immunity Agreements to the NIGC for review and approval, as necessary under

federal law; (b) the Chairman to provide to the NIGC on behalf of the Tribe any documentation requested by the NIGC in conjunction with its process of reviewing and approving, as necessary, the Amended Immunity Agreements, and any other information needed to expedite review and approval, and (c) the Chairman to act on behalf of the Tribe concerning the submission to the NIGC of the Amended Immunity Agreements.

8. That this Resolution of Limited Waiver shall take effect immediately.

9. That the Tribe agrees it will take all lawful steps necessary to preserve the effective terms of this Resolution of Limited Waiver in the event of future changes in its legal status or governance.

#### **CERTIFICATION**

The foregoing resolution was enacted by the Tribal Council of the Shingle Springs Band of Miwok Indians on the 16 day of 1000, 2004, by a vote of 4 Favor, D Opposed and Abstaining; at a duly called meeting at which a quorum of the Shingle Springs Tribal Council was present.

Chairman

Tribal Secr

# June 16, 2004 Amendment to

# FIRST AMENDED AND RESTATED MEMORANDUM OF AGREEMENT

# REGARDING

# **GAMING DEVELOPMENT**

# AND

# MANAGEMENT

### AGREEMENT

#### between

# SHINGLE SPRINGS BAND OF MIWOK INDIANS a federally recognized tribe

and

# LAKES KAR-SHINGLE SPRINGS, LLC a Delaware limited liability company

Dated: October 13, 2003

1

June 16, 2004 Amendment to October 13, 2003 Amended Memorandum Agreement

06/16/04 revision

THIS AMENDMENT ("AMENDMENT") to the October 13, 2003 FIRST AMENDED AND RESTATED MEMORANDUM OF AGREEMENT, by and between the Shingle Springs Band of Miwok Indians, a federally recognized Indian tribe (hereinafter referred to as ("the Tribe"), located in the State of California with tribal offices located at P.O. Box 1340, Shingle Springs, California 95682 and Lakes KAR-Shingle Springs, LLC, a Delaware limited liability company (hereinafter referred to as "LKAR"), whose business office is located at 130 Cheshire Lane, Minnetonka, MN 55305, is made and entered into this 16th day of June, 2004 by the parties.

#### RECITALS

A. The Tribe is a federally recognized Indian tribe eligible for the special programs and services provided by the United States to Indians because of their status as Indians and is recognized as possessing powers of self-government.

B. In compliance with the Indian Gaming Regulatory Act of 1988, P.L. 100-497, 25 U.S.C. 2701 <u>et seq.</u> as it may from time to time be amended, the Tribal Council of the Tribe has enacted a tribal ordinance regulating the operation of gaming activities on Tribal Lands (hereinafter referred to as the "Tribal Gaming Ordinance"), creating the Shingle Springs Tribal Gaming Commission, and authorizing Class II Gaming and Class III Gaming on its Tribal Lands subject to the provisions of the Tribal Gaming Ordinance and a Tribal-State Compact.

C. The Tribe is committed to the use of gaming activities to provide employment and improve the social, economic, education, and health needs of its members; to increase the revenues of the Tribe; and to enhance the Tribe's economic self-sufficiency and self-determination.

D. The Tribe presently lacks the resources to develop and operate a gaming facility and enterprise on its own and desires to retain the services of a developer and manager with knowledge and experience in the gaming industry to secure financing, develop, manage and operate a Class II Gaming and Class III Gaming facility and related resort facilities located on its Indian lands in accordance with the Indian Gaming Regulatory Act of 1988, as amended.

E. To assist with the financing, development, management and operations of its planned gaming facility and enterprise (the "Project"), the Tribe and LKAR agreed to enter into their First Amended and Restated Memorandum of Agreement Regarding Gaming Development and Management Agreement ("Amended Memorandum Agreement") and related Transaction Documents dated October 13, 2003 in connection with the Project.

F. The October 13, 2003 Amended Memorandum Agreement (and related exhibits and Transaction Documents) between the Tribe and LKAR has been submitted for approval to the National Indian Gaming Commission ("NIGC"); NIGC staff have reviewed the October 13, 2003 Amended Memorandum Agreement (and related exhibits and Transaction Documents) and have requested certain modifications to the documents prior to issuing NIGC approval; the Tribe and

2

06/16/04 revision

ì

į

LKAR have made such modifications to the documents as they have deemed necessary, including those made necessary by changes in circumstances related to increased financing needs of the Project; such modifications are incorporated herein and, accordingly, the Tribe and LKAR agree to enter into this June 16, 2004 Amendment to the October 13, 2003 First Amended and Restated Memorandum of Agreement Regarding Gaming Development and Management Agreement in connection with the Project.

G. This June 16, 2004 Amendment to the October 13, 2003 Amended Memorandum Agreement shall become effective when all the necessary approvals listed in Section 3.19 of the Amended Memorandum Agreement are received (the "Effective Date") and shall continue for a term of \_\_\_\_\_\_\_\_ from the Commencement Date, or as otherwise provided in the October 13, 2003 Amended Memorandum Agreement.

64

64

.64

H. Any dispute regarding this June 16, 2004 Amendment to the October 13, 2003 Amended Memorandum Agreement between the parties is to be subject to the dispute resolution and governing law provisions contained in the October 13, 2003 Amended Memorandum Agreement, as well as the Resolution of Limited Waiver attached as Exhibit B thereto.

NOW, THEREFORE, in consideration of the above circumstances and the hereinafter mutual promises and covenants, and for other good and valuable consideration as set forth herein, the receipt and sufficiency of which are expressly acknowledged, the Tribe and LKAR agree as follows:

1. Section 2.3(b)(i) of the October 13, 2003 Amended Memorandum Agreement is amended to state as follows:

(b)(i) The total amount of funds advanced to the Tribe directly from LKAR pursuant to Section 2.3(a)(i) shall equal

\_\_\_\_\_\_Principal and interest due under the Transition Loan shall be paid as provided under Section 2.3(c) below.

2. Section 2.3(b)(ii) of the October 13, 2003 Amended Memorandum Agreement is amended to state as follows:

(b)(ii) The total amount of funds advanced to the Tribe by LKAR pursuant to Section 2.3(a)(ii) shall equal

June 16, 2004 Amendment to October 13, 2003 Amended Memorandum Agreement

06/16/04 revision

1

and interest due under the Land Acquisition Loan shall be paid as provided under Section 2.3(c) below.

.../Principal

3. Section 2.14 of the October 13, 2003 Amended Memorandum Agreement is amended to state as follows:

<u>Term of Development Agreement</u>. Unless sooner terminated as provided in this Amended Memorandum Agreement, the term of the Development Agreement shall run until the earlier of either (i) the Commencement Date; or (ii)  $\int \int provided$  however, that the Notes and Security Provisions shall continue until all amounts owing to LKAR with respect thereto have been paid in full.

4. Section 6.5(b) of the October 13, 2003 Amended Memorandum Agreement is amended to state as follows:

(b) As compensation for LKAR's services, LKAR shall receive \_\_\_\_\_\_ of Net Total Revenues for the prior calendar month, for \_\_\_\_\_\_

as provided in this Amended Memorandum Agreement. Any amounts owing to LKAR hereunder shall be Limited Recourse obligations of the Tribe and shall be subject to the security provisions described in Section 6.3(c) hereof, including the Dominion Account Agreement and Security Agreement.

5. Section 2.11 of the October 13, 2003 Amended Memorandum Agreement is amended to state as follows:

Limited Waiver of Sovereign Immunity. By this Amended Memorandum Agreement, the Tribe does not waive, limit, or modify its sovereign immunity from unconsented suit except as provided in tribal Resolution 2003-12 dated October 13, 2003 attached hereto as <u>Exhibit B</u> and in tribal Resolution 2004 attached June 16, 2004 attached hereto as <u>Exhibit C</u>. The Tribe understands that its agreement to adopt enforceable resolutions of limited waiver is a material inducement to the LKAR's execution of this Amended Memorandum Agreement and is a condition precedent to any of the respective obligations of the parties under this Amended Memorandum Agreement. The Tribe further agrees that it will not amend or alter or in any way lessen the rights of the Lender or LKAR as set forth in the resolutions of limited waiver, which are attached hereto as <u>Exhibit B</u> and <u>Exhibit C</u>, both of which are incorporated here by reference. This Section 2.11 shall survive the termination of this Amended Memorandum Agreement, regardless of the reason for the termination. June 16, 2004 Amendment to

October 13, 2003 Amended Memorandum Agreement

4

06/16/04 revision

Ì

6. Section 10.10 of the October 13, 2003 Amended Memorandum Agreement is amended to state as follows:

<u>Sovereign Immunity</u>. Except as described in the resolutions of limited waiver attached hereto as <u>Exhibit B</u> and <u>Exhibit C</u>, both of which are incorporated herein by reference, nothing in this Amended Memorandum Agreement shall be deemed or construed to constitute a waiver of sovereign immunity of the Tribe and the only applicable waivers of sovereign immunity shall be those expressly provided and executed by the Tribe's duly authorized representative and substantially conforming to the form as approved by the parties. The parties agree that they will not amend or alter the resolution of limited waiver attached hereto as <u>Exhibit B</u> and <u>Exhibit C</u> in any way which will lessen the rights of any party as set forth in the resolutions of limited waiver. This Section 10.10 shall survive termination of this Amended Memorandum Agreement, regardless of the reason for the termination.

7. Section 11.3(b) of the October 13, 2003 Amended Memorandum agreement is amended to state as follows:

(b) If the dispute is not resolved to the satisfaction of the parties within thirty (30) calendar days after the first meeting in Section 11.3(a) above, then any claim, controversy or dispute arising out of or relating to this Amended Memorandum Agreement, Facility Note, Land Acquisition Note, the Interim Promissory Note, the Operating Note, or any other Transaction Document, or any alleged default hereunder or breach of any provisions thereof, shall be submitted to binding arbitration in accordance with the Commercial Arbitration Rules of the American Arbitration Association in effect at the time of submission; except that: (a) the question whether or not a dispute is arbitrable under this Amended Memorandum Agreement or any other Transaction Document shall be a matter for binding arbitration by the arbitrators, such question shall not be determined by any court and, in determining any such question, all doubts shall be resolved in favor of arbitrability; and (b) discovery shall be permitted in accordance with the Federal Rules of Civil Procedure, subject to supervision as to scope and appropriateness by the arbitrators. Judgment on any arbitration award may be entered in any court having jurisdiction over the parties pursuant to the resolutions of limited waiver attached hereto as Exhibit B and Exhibit C, both of which are incorporated herein by reference.

8. Section 11.3(f) of the October 13, 2003 Amended Memorandum agreement is amended to state as follows:

(f) The arbitration award shall be in writing signed by each of the arbitrators, and shall state the basis for the award. The arbitration award shall be set forth in reasonable detail as to its findings of fact and law, and basis of determination of award form and amount. In connection with any arbitration award, the arbitrators shall be empowered to take the actions and enforce the judicial remedies described in Paragraph 5 of the resolutions of limited waiver attached hereto as <u>Exhibit B</u>

June 16, 2004 Amendment to

October 13, 2003 Amended Memorandum Agreement

5

06/16/04 revision

}

and Exhibit C, both of which are incorporated herein by reference; provided however, that although -the arbitrators may award damages in the event the Tribe or the Tribal Gaming Commission do not to comply with the award, the arbitrators may not require the Tribe or the Tribal Gaming Commission to take or modify any governmental legislative decision or action which the arbitrators have determined has resulted in the dispute between the parties and is contrary to the parties rights, liabilities or obligations under this Amended Memorandum Agreement, the Facility Note, the Land Aequisition Note, the Interim Promissory Note, the Operating Note, or any other Transaction Document ("Specific Performance Restriction"). Provided further, that: (a) should the arbitrators determine that there has been an intentional bad faith violation of a party's rights under this Amended Memorandum Agreement or any other transaction Document by the Tribe or Tribal Gaming Commission, and if the Tribe or the Tribal Gaming Commission do not reverse such intentional bad faith violation through governmental legislative decision or action within thirty (30) days after the being notified by the arbitrators of such determination, then the arbitrators shall

9. The List of Exhibits page of the October 13, 2003 Amended Memorandum Agreement is amended to state as follows:

#### LIST OF EXHIBITS

Exhibit A Legal Description of Gaming Site

Exhibit B	Resolution 2003-12 concerning Limited Waiver of Immunity from Suit	$\hat{O}$
Exhibit $C^-$	Resolution 2003-12 concerning Limited Waiver of Immunity from Suit Resolution 2004- $\frac{23}{24}$ concerning Limited Waiver of Immunity from Suit	Ę,
	6-18-04	८(19) ज
	W Ø IS OG	

64

10. Section 10.14 of the October 13, 2003 Amended Memorandum Agreement is amended to state as follows:

Entire Agreement. This Amended Memorandum Agreement (including all its exhibits and related Transaction Documents, along with tribal Resolution 2004-18 approving the establishment of the Foothills Oaks Tribal Gaming Authority and its related Ordinance, the June 16, 2004 Amendment to the October 13, 2003 Amended Memorandum Agreement, the June 16, 2004 Consent Agreement June 16, 2004 Amendment to 6

October 13, 2003 Amended Memorandum Agreement

06/16/04 revision

ì

ţ

Echolog 6-18-04 21: 6-16-04

entered into by the parties, and tribal Resolution 2004 ar concerning the June 16, 2004 Amendment and Consent Agreement), represents the entire agreement between the parties and supersedes all prior agreements relating to the subject matter of Class II Gaming and Class III Gaming to be developed and conducted by the Tribe at the Facility and operations of the Enterprise. The parties hereto intend that this Amended Memorandum Agreement (and related Transaction Documents) is to supercede and replace the May 5, 2000 Memorandum of Agreement and any other prior agreements between the parties in connection with the Project.

11. The Tribe and LKAR agree that any dispute in connection with this June 16, 2004 Amendment to the October 13, 2003 Amended Memorandum Agreement shall be subject to the dispute resolution procedures and limited waiver of sovereign immunity contained in the October 13, 2003 Amended Memorandum Agreement and the Resolution of Limited Waiver attached thereto as Exhibit B, the terms of which are incorporated by reference herein.

12. The Tribe and LKAR agree that capitalized terms used herein and not defined shall have the meanings given them in the October 13, 2003Amended Memorandum Agreement.

13. The Tribe and LKAR agree that this Amendment shall be construed in accordance with and governed by the internal laws and decisions of the State of California, without giving effect to its choice of law principles.

14. The Tribe and LKAR agree that no modification, amendment or change to this Amendment shall be valid unless the same is in writing and signed by the party against which the enforcement of such modification, amendment or change is sought.

15. Except as amended above, all other provisions of the October 13, 2003 Amended Memorandum Agreement shall remain in full force and effect as originally stated and are equally applicable hereto.

June 16, 2004 Amendment to October 13, 2003 Amended Memorandum Agreement

06/16/04 revision

)

IN WITNESS WHEREOF, the parties hereto have executed this June 16, 2004 Amendment to the October 13, 2003 Amended Memorandum Agreement of the parties.

Shingle Springs Band of Miwok Indian By: Its: Tribal hairman By: Its: l Ú.S.C. § Approved pursuant to 2 National Indian Gaming Commission 19 By: Print Name: Philip N. Hogen Its Chairman

Lakes KAR-Shingle Springs, LLC

Bv: Its:

June 16, 2004 Amendment to October 13, 2003 Amended Memorandum Agreement

06/16/04 revision

7



MAR 2 0 2007

Nicholas C. Fonseca, Chairperson Shingle Springs Rancheria 5281 Honpie Road Placerville, CA 95667 Fax (530) 676-8033

Timothy Cope, President Lakes KAR-Shingle Springs, LLC c/o Lakes Entertainment, Inc. 130 Cheshire Lane Minnetonka, MN 55305 Fax (952) 449-9353

Dear Chairperson Fonseca and Mr. Cope:

On November 9, 2006, the National Indian Gaming Commission (the "NIGC") received the Second Amendment (the "Modification") to a NIGC approved Class II and III gaming management contract (the "Contract") between the Shingle Springs Band of Miwok Indians (the "Tribe") and Lakes KAR-Shingle Springs, LLC ("Lakes", "LKAR", or the "Manager"). I am pleased to inform you that I have approved the Modification.

The Indian Gaming Regulatory Act and the regulations of the NIGC require that the NIGC Chairman approve modifications to management contracts for gaming operations on Indian lands. Accordingly, you submitted the Modification as a modification for approval as required by 25 U.S.C. § 2711 and 25 C.F.R. Part 535. We have reviewed the Modification and other information submitted and have determined that the standards of 25 C.F.R. Parts 531 and 535 have been met. This letter and my signature on the Modification constitute such approval.

Please note that if we learn of any actions or conditions that violate the standards contained in 25 C.F.R. Parts 531, 533, 535, or 537, we may require modifications of, or may void, the Contract and/or the Modification, after providing the parties with an opportunity for a hearing and a subsequent appeal to the NIGC as set forth in 25 C.F.R. Part 577.

Should you have any questions concerning this approval, please call us/at (202)7632-7003.

Sincerel hilip N. Hogen Chairman

NATIONAL HEADQUARTERS 1441 L St. NW, Suite 9100, Washington, DC 20005 Tel: 202.632.7003 Fax: 202.632.7066 www.NIGC.COV

REGIONAL OFFICES Portland, OR; Sacramento, CA; Phoenix, AZ; St. Paul, MN; Tulsa, OK

# Second Amendment dated January 23, 2007 to

# FIRST AMENDED AND RESTATED MEMORANDUM OF AGREEMENT

# REGARDING

#### GAMING DEVELOPMENT

# AND

# MANAGEMENT

#### AGREEMENT

### between

# SHINGLE SPRINGS BAND OF MIWOK INDIANS a federally recognized tribe

and

# LAKES KAR-SHINGLE SPRINGS, LLC a Delaware limited liability company

Dated: October 13, 2003

Second Amendment to October 13, 2003 Amended Memorandum Agreement

1

01/23/07 execution version

THIS SECOND AMENDMENT ("Second Amendment") to the October 13, 2003 FIRST AMENDED AND RESTATED MEMORANDUM OF AGREEMENT, as amended by the parties' June 16, 2004 Amendment, is made and entered into this 23rd day of January, 2007 by and between the Shingle Springs Band of Miwok Indians, a federally recognized Indian tribe ("Tribe"), located in the State of California with tribal offices located at P.O. Box 1340, Shingle Springs, California 95682 and Lakes KAR-Shingle Springs, LLC, a Delaware limited liability company (hereinafter referred to as "LKAR"), whose business office is located at 130 Cheshire Lane, Minnetonka, MN 55305.

#### RECITALS

A. The Tribe is a federally recognized Indian tribe eligible for the special programs and services provided by the United States to Indians because of their status as Indians and is recognized as possessing powers of self-government.

B. In compliance with the Indian Gaming Regulatory Act of 1988 ("IGRA"), P.L. 100-497, 25 U.S.C. 2701 <u>et seq.</u> as it may from time to time be amended, the Tribal Council of the Tribe has enacted a tribal ordinance regulating the operation of gaming activities on Tribal Lands (hereinafter referred to as the "Tribal Gaming Ordinance"), creating the Shingle Springs Tribal Gaming Commission, and authorizing Class II Gaming and Class III Gaming on its Indian lands subject to the provisions of the Tribal Gaming Ordinance and a Tribal-State Compact.

C. The Tribe is committed to the use of gaming activities to provide employment and improve the social, economic, education, and health needs of its members; to increase the revenues of the Tribe; and to enhance the Tribe's economic self-sufficiency and self-determination.

D. The Tribe presently lacks the resources to develop and operate a gaming facility and enterprise on its own and desires to retain the services of a developer and manager with knowledge and experience in the gaming industry to secure financing, develop, manage and operate a Class II Gaming and Class III Gaming facility and related resort facilities located on its Indian lands in accordance with the Indian Gaming Regulatory Act of 1988, as amended.

E. To assist with the financing, development, management and operations of its planned gaming facility (to be known as the "Foothill Oaks Casino") and related ancillary facilities (the "Project"), the Tribe and LKAR agreed to enter into their First Amended and Restated Memorandum of Agreement Regarding Gaming Development and Management Agreement ("Amended Memorandum Agreement") and related Transaction Documents dated October 13, 2003 in connection with the Project.

F. The October 13, 2003 Amended Memorandum Agreement (and related exhibits and Transaction Documents), as amended by the June 16, 2004 Amendment between the Tribe and LKAR, was approved by the Chairman of the National Indian Gaming Commission ("NIGC") on July 19, 2004.

Second Amendment to October 13, 2003 Amended Memorandum Agreement 01/23/07 execution version G. Since signing the October 13, 2003 Amended Memorandum Agreement, the Tribe has adopted tribal Resolution 2004-18 and related Ordinance (hereinafter referred to as the "Ordinance 2004-18") whereby the Tribe has established the Shingle Springs Tribal Gaming Authority ("Authority") and, subject to certain limitations, delegated to the Authority all governmental and proprietary powers and rights of the Tribe over the development, construction, operation, promotion and financing of the Project; and LKAR has consented to the delegation of the Tribe's governmental and proprietary powers and rights to the Authority in connection with the Project.

H. Due to indefinite delays in the financing and construction of the Project arising from litigation commenced by El Dorado County in connection with the dedicated interchange from US 50 providing direct access to the Shingle Springs Rancheria, the Tribe has amended its Ordinance 2004-18 to provide that its delegation to the Authority of certain governmental and proprietary powers and rights of the Tribe over the development, construction, operation, promotion and financing of the Project is suspended until such date that the Tribal Council activates the Authority by formally appointing the Management Board of the Authority; until such time the Tribe shall retain and reassume all of the Tribe's rights and responsibilities under the October 13, 2003 Amended Memorandum Agreement (and related exhibits and Transaction Documents), as amended by the June 16, 2004 Amendment between the Tribe and LKAR.

I. The Tribe and LKAR desire to further amend the October 13, 2003 Amended Memorandum Agreement (and related exhibits and Transaction Documents), as amended by the June 16, 2004 Amendment, deemed necessary by changes in circumstances related to increased financing needs of the Project and to provide for the implementation of Tribe's assignment described in Recital G above; such modifications are incorporated herein and. accordingly, the Tribe and LKAR agree to enter into this Second Amendment to the October 13, 2003 First Amended and Restated Memorandum of Agreement Regarding Gaming Development and Management Agreement, as amended by the June 16, 2004 Amendment, in connection with the Project.

J. This Second Amendment shall supersede and replace the agreement entitled Second Amendment to the October 13, 2003 First Amended and Restated Memorandum of Agreement Regarding Gaming Development and Management Agreement dated July 19, 2006 and shall become effective when approved by the Chairman of the NIGC under IGRA.

K. Any dispute regarding this Second Amendment between the parties is to be subject to the dispute resolution and governing law provisions contained in the October 13, 2003 Amended Memorandum Agreement, as well as the Resolutions of Limited Waiver attached as Exhibit B, C and D thereto.

NOW, THEREFORE, in consideration of the above circumstances and the hereinafter mutual promises and covenants, and for other good and valuable consideration as set forth herein, the receipt and sufficiency of which are expressly acknowledged, the Tribe and LKAR agree as follows:

Second Amendment to October 13, 2003 Amended Memorandum Agreement 01/23/07 execution version
Section 2.3(a)(i) of the October 13, 2003 Amended Memorandum Agreement is amended to state as follows:

(a)(i) LKAR agrees to make the following pre-construction advances to the Tribe: (1)

64

The Tribe and LKAR agree that all sums previously advanced to the Tribe by KARSS under the superseded Development and Management Agreements, and by LKAR under the superseded May 5, 2000 Memorandum of Agreement (excluding advances for acquisition of the Acquired Tribal Lands), shall constitute advances by LKAR to the Tribe hereunder, be credited to LKAR's obligations hereunder, and shall be subject to the terms of the Transition Loan herein.

2. Section 2.3(b)(i) of the October 13, 2003 Amended Memorandum Agreement, as amended by the June 16, 2004 Amendment, is amended to state as follows:

(b)(i) The total amount of funds advanced to the Tribe directly from LKAR pursuant to Section 2.3(a)(i) shall equal

Principal and interest due under the Transition Loan shall be paid as provided under Section 2.3(c) below.

3. Section 2.3(b)(ii) of the October 13, 2003 Amended Memorandum Agreement, as amended by the June 16, 2004 Amendment, is amended to state as follows:

Second Amendment to October 13, 2003 Amended Memorandum Agreement 01/23/07 execution version

¢

(b)(ii) The total amount of funds advanced to the Tribe by LKAR pursuant to Section 2.3(a)(ii) shall equal

64

due under the Land Acquisition Loan shall be paid as provided under Section 2.3(c) below.

 Section 2.5(c) of the October 13, 2003 Amended Memorandum Agreement, as amended by the June 16, 2004 Amendment, is amended to state as follows:

(c) The Costs of Construction and Initial Costs of Operation shall equal

 Section 2.11 of the October 13, 2003 Amended Memorandum Agreement, as amended by the June 16, 2004 Amendment, is amended to state as follows:

Limited Waiver of Sovereign Immunity. By this Amended Memorandum Agreement, the Tribe does not waive, limit, or modify its sovereign immunity from unconsented suit except as provided in tribal Resolution 2003-12 dated October 13, 2003 attached hereto as Exhibit B, in tribal Resolution 2004-23 dated June 16, 2004 attached hereto as Exhibit C, and in tribal Resolution 2007-xx dated January xx, 2007 attached hereto as Exhibit D. The Tribe understands that its agreement to adopt enforceable resolutions of limited waiver is a material inducement to LKAR's execution of this Amended Memorandum Agreement and is a condition precedent to any of the respective obligations of the parties under this Amended Memorandum Agreement. The Tribe further agrees that it will not amend or alter or in any way lessen the rights of the Lender or LKAR as set forth in the resolutions of limited waiver, which are attached hereto as Exhibit C, and Exhibit D, each of which are incorporated here by reference.

Second Amendment to

This Section 2.11 shall survive the termination of this Amended Memorandum Agreement, regardless of the reason for the termination.

Exhibit D referenced above and attached to this Second Amendment shall constitute "Exhibit D" of the Amended Memorandum Agreement.

6. Section 10.10 of the October 13, 2003 Amended Memorandum Agreement, as amended by the June 16, 2004 Amendment, is amended to state as follows:

<u>Sovereign Immunity</u>. Except as described in the resolutions of limited waiver attached hereto as <u>Exhibit B</u>, <u>Exhibit C</u>, and <u>Exhibit D</u>, each of which are incorporated herein by reference, nothing in this Amended Memorandum Agreement shall be deemed or construed to constitute a waiver of sovereign immunity of the Tribe and the only applicable waivers of sovereign immunity shall be those expressly provided and executed by the Tribe's duly authorized representative and substantially conforming to the form as approved by the parties. The parties agree that they will not amend or alter the resolution of limited waiver attached hereto as <u>Exhibit B</u>, <u>Exhibit C</u> and <u>Exhibit D</u>, in any way which will lessen the rights of any party as set forth in the resolutions of limited waiver. This Section 10.10 shall survive termination of this Amended Memorandum Agreement, regardless of the reason for the termination.

 Section 11.3(b) of the October 13, 2003 Amended Memorandum Agreement, as amended by the June 16, 2004 Amendment, is amended to state as follows:

(b) If the dispute is not resolved to the satisfaction of the parties within thirty (30) calendar days after the first meeting in Section 11.3(a) above, then any claim, controversy or dispute arising out of or relating to this Amended Memorandum Agreement, Facility Note, Land Acquisition Note, the Interim Promissory Note, the Operating Note, or any other Transaction Document, or any alleged default hereunder or breach of any provisions thereof, shall be submitted to binding arbitration in accordance with the Commercial Arbitration Rules of the American Arbitration Association in effect at the time of submission: except that: (a) the question whether or not a dispute is arbitrable under this Amended Memorandum Agreement or any other Transaction Document shall be a matter for binding arbitration by the arbitrators, such question shall not be determined by any court and, in determining any such question, all doubts shall be resolved in favor of arbitrability; and (b) discovery shall be permitted in accordance with the Federal Rules of Civil Procedure, subject to supervision as to scope and appropriateness by the arbitrators. Judgment on any arbitration award may be entered in any court having jurisdiction over the parties pursuant to the resolutions of limited waiver attached hereto as Exhibit B, Exhibit C, and Exhibit D, each of which are incorporated herein by reference.

Second Amendment to October 13, 2003 Amended Memorandum Agreement 01/23/07 execution version  Section 11.3(f) of the October 13, 2003 Amended Memorandum Agreement, as amended by the June 16, 2004 Amendment, is amended to state as follows:

The arbitration award shall be in writing signed by each of the arbitrators, (f) and shall state the basis for the award. The arbitration award shall be set forth in reasonable detail as to its findings of fact and law, and basis of determination of award form and amount. In connection with any arbitration award, the arbitrators shall be empowered to take the actions and enforce the judicial remedies described in Paragraph 5 of the resolutions of limited waiver attached hereto as Exhibit B, Exhibit C, and Exhibit D, each of which are incorporated herein by reference; provided however, that although the arbitrators may award damages in the event the Tribe or the Tribal Gaming Commission do not comply with the award, the arbitrators may not require the Tribe or the Tribal Gaming Commission to take or modify any governmental legislative decision or action which the arbitrators have determined has resulted in the dispute between the parties and is contrary to the parties' rights, liabilities or obligations under this Amended Memorandum Agreement, the Facility Note, the Land Acquisition Note, the Interim Promissory Note, the Operating Note, or any other Transaction Document ("Specific Performance Restriction"). Provided further, that: (a) should the arbitrators determine that there has been an intentional bad faith violation of a party's rights under this Amended Memorandum Agreement or any other Transaction Document by the Tribe or Tribal Gaming Commission, and if the Tribe or the Tribal Gaming Commission do not reverse such intentional bad faith violation through governmental legislative decision or action within thirty (30) days after being notified by the arbitrators of such determination, then the arbitrators shall

64

9. The List of Exhibits page of the October 13, 2003 Amended Memorandum Agreement, as amended by the June 16, 2004 Amendment, is amended to state as follows:

#### LIST OF EXHIBITS

Exhibit A	Legal Description of Gaming Site
Exhibit B	Resolution 2003-12 concerning Limited Waiver of Immunity from Suit
Exhibit C	Resolution 2004-23 concerning Limited Waiver of Immunity from Suit

Second Amendment to October 13, 2003 Amended Memorandum Agreement 01/23/07 execution version Exhibit D Resolution 2007-04 concerning Limited Waiver of Immunity from Suit

10. Section 10.14 of the October 13, 2003 Amended Memorandum Agreement, as amended by the June 16, 2004 Amendment, is amended to state as follows:

Entire Agreement. This Amended Memorandum Agreement (including all its exhibits and related Transaction Documents, along with tribal Resolution 2004-18 approving the establishment of the Foothills Oaks Tribal Gaming Authority and its related Ordinance. the June 16, 2004 Amendment to the October 13, 2003 Amended Memorandum Agreement, the June 16, 2004 Consent Agreement entered into by the parties, tribal Resolution 2004-23 concerning the June 16, 2004 Amendment and Consent Agreement), the July 19, 2006 Second Amendment to the October 13, 2003 Amended Memorandum Agreement, and tribal Resolution 2007-04 concerning the January 23, 2007 Second Amendment, represents the entire agreement between the parties and supersedes all prior agreements relating to the subject matter of Class II Gaming and Class III Gaming to be developed and conducted by the Tribe at the Facility and operations of the Enterprise. The parties hereto intend that this Amended Memorandum Agreement (and related Transaction Documents) is to supersede and replace the May 5, 2000 Memorandum of Agreement and any other prior agreements between the parties in connection with the Project.

11. The Tribe and LKAR agree and consent that upon the date the Authority is activated by the Tribal Council formally appointing the Management Board of the Authority, and the Tribe and Authority have completed each of the additional requirements set forth below, all governmental and proprietary powers and rights of the Tribe over the development, construction, operation, promotion and financing of the Foothill Oaks Casino and the Enterprise, as described in Section 1 of Ordinance 2004-18, shall be solely assumed by and be the delegated responsibility of the Authority:

(a) The Tribe and LKAR shall execute and deliver the "Tribal Agreement" (in the form attached hereto as <u>Exhibit 1</u>) whereby the Tribe reaffirms the assignment to the Authority of its rights and responsibilities under the October 13, 2003 Amended Memorandum Agreement (and related exhibits and Transaction Documents), as amended by the June 16, 2004 Amendment between the Tribe and LKAR and this Second Amendment, and agrees to undertake certain rights and obligations in connection with the development, construction, operation, promotion and financing of the Foothill Oaks Casino and the Enterprise.

(b) LKAR, the Tribe and the Authority shall execute and deliver the "Third Amendment" to the October 13, 2003 Amended Memorandum Agreement (in the form attached hereto as <u>Exhibit 2</u> and hereinafter referred to as "Third Amendment"), reflecting the assumption

Second Amendment to

by the Authority of all governmental and proprietary powers and rights of the Tribe over the development, construction, operation, promotion and financing of the Foothill Oaks Casino and the Enterprise as described in Section 1 of Ordinance 2004-18, and the Tribe shall cause the Authority also to duly approve the execution, delivery and performance of the Third Amendment and a related resolution of limited waiver (in the form attached hereto as <u>Exhibit 3</u>) in connection therewith and deliver a duly certified copy of such resolution to LKAR.

(c) The Tribe and LKAR agree that, following execution of the Third Amendment and Tribal Agreement, these documents shall be submitted to the NIGC for approval under applicable Legal Requirements, and the Tribe agrees to cause the Authority to cooperate with such submission.

(d) The Authority has provided LKAR with written confirmation that the NIGC has approved and executed the Third Amendment and no Material Breach by the Tribe under the Transaction Documents shall have occurred and be continuing as of the effective date of the Third Amendment.

64

13. The Tribe and LKAR agree that any dispute in connection with this Second Amendment shall be subject to the dispute resolution procedures and limited waiver of sovereign immunity contained in the October 13, 2003 Amended Memorandum Agreement and the Resolution of Limited Waiver attached thereto as <u>Exhibit B</u>, the terms of which are incorporated by reference herein.

 The Tribe and LKAR agree that capitalized terms used herein and not defined shall have the meanings given them in the October 13, 2003 Amended Memorandum Agreement.

15. The Tribe and LKAR agree that this Second Amendment shall be construed in accordance with and governed by the internal laws and decisions of the State of California, without giving effect to its choice of law principles; and may be executed in any number of counterparts and by facsimile, each of which shall be considered an original but together shall constitute one and the same instrument.

16. The Tribe and LKAR agree that no modification, amendment or change to this Amendment shall be valid unless the same is in writing and signed by the party against which the enforcement of such modification, amendment or change is sought, and is approved by the Chairman of the NIGC.

Second Amendment to October 13, 2003 Amended Memorandum Agreement 01/23/07 execution version 17. The Tribe and LKAR agree that this Second Amendment shall supersede and replace the agreement entitled Second Amendment to the October 13, 2003 First Amended and Restated Memorandum of Agreement Regarding Gaming Development and Management Agreement dated July 19, 2006, and, except as amended above, all other provisions of the October 13, 2003 Amended Memorandum Agreement, as amended by the June 16, 2004 Amendment, shall remain in full force and effect as originally stated and are equally applicable hereto.

IN WITNESS WHEREOF, the parties hereto have as of January 23, 2007 executed this Second Amendment to the October 13, 2003 Amended Memorandum Agreement, as amended by the June 16, 2004 Amendment, between the parties.

Shingle Springs Band of Miwok Indians By: Its: Tribal Chairman By: Its: Approved pursuant to 25 U.S.C. §2 National Indian Gaming Commission By: Print Name: Philip N. Hogen Its Chairman

MAR 2 0 2007

Second Amendment to October 13, 2003 Amended Memorandum Agreement 01/23/07 execution version Lakes KAR-Shingle Springs, LLC

B Its: TRE

10



June 8, 2007

Nicholas C. Fonseca, Chairperson Shingle Springs Rancheria 5281 Honpie Road Placerville, CA 95667 Fax (530) 676-8033

Timothy Cope, President Lakes KAR-Shingle Springs, LLC c/o Lakes Entertainment, Inc. 130 Cheshire Lane Minnetonka, MN 55305 Fax (952) 449-9353

Dear Chairperson Fonseca and Mr. Cope:

On May 22, 2007, the National Indian Gaming Commission ("NIGC") received a Third Amendment (the "Modification"), dated as of May 17, 2007, to the First Amended and Restated Memorandum of Agreement regarding Gaming Development and Management Agreement (the "Contract") between the Shingle Springs Band of Miwok Indians (the "Tribe") and Lakes KAR-Shingle Springs, LLC (the "Manager"). I am pleased to inform you that I have approved the Modification.

The Indian Gaming Regulatory Act and the regulations of the NIGC require that the NIGC Chairman approve modifications to management contracts for gaming operations on Indian lands. Accordingly, you submitted the Modification as a modification for approval as required by 25 U.S.C. § 2711 and 25 C.F.R. Part 535. We have reviewed the Modification and other information submitted and have determined that the standards of 25 C.F.R. Parts 531 and 535 have been met. This letter and my signature on the Modification constitute such approval.

Please note that if we learn of any actions or conditions that violate the standards contained in 25 C.F.R. Parts 531, 533, 535, or 537, we may require modifications of, or may void, the Contract and/or the Modification, after providing the parties with an opportunity for a hearing and a subsequent appeal to the NIGC as set forth in 25 C.F.R. Part 577.

Should you have any questions concerning this approval, please call us at (202) 632-7003.

Sincerely, Chair/nan

## **Third Amendment** dated as of May 17, 2007 to

# FIRST AMENDED AND RESTATED **MEMORANDUM OF AGREEMENT**

## REGARDING

## **GAMING DEVELOPMENT**

## AND

## MANAGEMENT

## AGREEMENT

#### between

## SHINGLE SPRINGS BAND OF MIWOK INDIANS a federally recognized tribe

and

## LAKES KAR-SHINGLE SPRINGS, LLC a Delaware limited liability company

Dated: October 13, 2003

Third Amendment to October 13, 2003 Amended Memorandum Agreement 05/17/07 execution version

-1-

THIS THIRD AMENDMENT ("Third Amendment") to the October 13, 2003 FIRST AMENDED AND RESTATED MEMORANDUM OF AGREEMENT, as amended by the parties' June 16, 2004 Amendment and January 23, 2007 Amendment, is made by and between the Shingle Springs Tribal Gaming Authority (hereinafter referred to as "Authority"), an instrumentality of the Shingle Springs Band of Miwok Indians, a federally-recognized Indian tribe (hereinafter referred to as "Tribe"), authorized by tribal Resolution 2004-18 and its related Ordinance, as amended, to exercise the Tribe's proprietary rights and powers in connection with a gaming facility to be developed on behalf of the Tribe by the Authority, whose offices are located at P.O. Box 1340, Shingle Springs, California 95682, and Lakes KAR-Shingle Springs, LLC, a Delaware limited liability company (hereinafter referred to as "LKAR"), whose business office is located at 130 Cheshire Lane, Minnetonka, MN 55305, and entered into as of the 17th day of May, 2007.

#### RECITALS

A. The Tribe is a federally recognized Indian tribe eligible for the special programs and services provided by the United States to Indians because of their status as Indians and is recognized as possessing powers of self-government.

B. In compliance with the Indian Gaming Regulatory Act of 1988, P.L. 100-497, 25 U.S.C. 2701 <u>et seq.</u> as it may from time to time be amended, the Tribal Council of the Tribe has enacted a tribal ordinance regulating the operation of gaming activities on Tribal Lands (hereinafter referred to as the "Tribal Gaming Ordinance"), creating the Shingle Springs Tribal Gaming Commission, and authorizing Class II Gaming and Class III Gaming on its Indian lands subject to the provisions of the Tribal Gaming Ordinance and a Tribal-State Compact.

C. The Tribe is committed to the use of gaming activities to provide employment and address the social, economic, education, and health needs of its members; to increase the revenues of the Tribe; and to enhance the Tribe's economic self-sufficiency and self-determination.

D. Because the Tribe lacks the resources to develop and operate a gaming facility and enterprise on its own, it desired to retain the services of a developer and manager with knowledge and experience in the gaming industry to secure financing, develop, manage and operate a Class II Gaming and Class III Gaming facility and related resort facilities located on its Indian lands in accordance with the Indian Gaming Regulatory Act of 1988, as amended.

E. To assist with the financing, development, management and operations of its planned gaming facility and enterprise to be known as the Foothill Oaks Casino (the "Project"), the Tribe and LKAR entered into the First Amended and Restated Memorandum of Agreement Regarding Gaming Development and Management Agreement (as amended by an Amendment dated June 16, 2004 and approved by the Chairman of the National Indian Gaming Commission ("NIGC") on July 19, 2004 and as further amended by a Second Amendment dated January 23, 2007 approved by the NIGC on March 20, 2007, hereinafter collectively referred to as the "October 13, 2003 Amended Memorandum

Third Amendment to

Agreement") and other Transaction Documents dated October 13, 2003 in connection with the Project; capitalized terms used but not otherwise defined herein shall have the meanings set forth in the October 13, 2003 Amended Memorandum Agreement.

F. On April 21, 2007, the Tribal Council of the Tribe activated the Shingle Springs Tribal Gaming Authority (the "Authority"), previously established by tribal Resolution 2004-18 as an instrumentality of the Tribe, by appointing the Management Board of the Authority; and the governmental and proprietary powers and rights of the Tribe over the development, construction, operation, promotion and financing of the Foothill Oaks Casino and the Enterprise, as described in Section 1 of Ordinance 2004-18, thereby became the delegated responsibility of the Authority (such activation and delegation being referred to as the "Authority Activation").

G. The Tribe, the Authority and LKAR have contemporaneously herewith executed an Assignment and Assumption Agreement dated as of April 20, 2007 by which, in connection with the Authority Activation, the Tribe assigned to the Authority, and the Authority assumed, the Tribe's rights and responsibilities under the October 13, 2003 Amended Memorandum Agreement and other Transaction Documents.

H. Pursuant to the terms of the January 23, 2007 Second Amendment, the Tribe agreed that upon the Authority becoming activated the Tribe would cause the Authority (i) to approve and execute this Third Amendment, (ii) enter into an agreement reflecting the assignment to and assumption by the Authority of the Tribe's rights and responsibilities under the October 13, 2003 Amended Memorandum Agreement and other Transaction Documents, in accordance with Section 1 of Ordinance 2004-18; and (iii) adopt a resolution of limited waiver related to approval of this Third Amendment.

I. Financing and construction of the Project and the dedicated interchange from US 50 providing direct access to the Shingle Springs Rancheria was delayed because of litigation commenced by El Dorado County. The Tribe has entered into an agreement with El Dorado County and received judicial decisions in connection with the litigation, and has also received a permit from the California Department of Transportation ("CALTRANS"), which will now allow construction of the US 50 dedicated interchange to begin.

J. Consistent with the January 23, 2002 FONSI issued by the NIGC in connection with its approval of the October 13, 2003 Amended Memorandum Agreement, including the requirement that the Tribe obtain approval of the US 50 dedicated interchange by CALTRANS and the BIA "prior to construction and operation" of the Project, and also the Tribe's July 7, 2004 assurance that "it will not open and operate its gaming facility... until the US 50 dedicated interchange is completed and open to traffic," construction of the US 50 dedicated interchange and the Project is scheduled to commence in May 2007, and the Tribe expects to close on the Facility Loan financing in June 2007.

K. In connection with the financing and construction of the Project and the US 50

Third Amendment to

dedicated interchange, the Authority and LKAR desired to further amend the October 13, 2003 Amended Memorandum Agreement as deemed necessary by the Authority and LKAR; accordingly, the Authority and LKAR entered into a Third Amendment to the October 13, 2003 Amended Memorandum Agreement dated as of April 20, 2007; since signing the Third Amendment dated as of April 20, 2007, the Authority and LKAR desire to make certain technical revisions to that agreement before approval by the NIGC and such necessary modifications are incorporated herein; and, accordingly, the Authority and LKAR agree to enter into this Third Amendment in connection with the Project. Therefore, the parties hereto intend that this Third Amendment is to supersede and replace in all respects the Third Amendment dated as of April 20, 2007.

L. This Third Amendment shall become effective when approved and executed by the Chairman of the NIGC.

M. Any dispute regarding this Third Amendment is to be subject to the dispute resolution and governing law provisions contained in the October 13, 2003 Amended Memorandum Agreement, as well as the Authority's resolution of limited waiver to be attached as <u>Exhibit E</u> thereto.

NOW, THEREFORE, in consideration of the above circumstances and the hereinafter mutual promises and covenants, and for other good and valuable consideration as set forth herein, the receipt and sufficiency of which are expressly acknowledged, the Authority and LKAR agree as follows:

 This Third Amendment shall become effective when approved and executed by the Chairman of the NIGC.

2. The Authority acknowledges and agrees that, through and subject to the Assignment and Assumption Agreement dated as of April 20, 2007, the Tribe has duly assigned and delegated to the Authority all of its right, title, interest and obligations in and to the Project, the Enterprise, the October 13, 2003 Amended Memorandum Agreement, and the Authority has accepted the assignment and assumption of each of the same. Without limiting the generality of the foregoing, and for avoidance of doubt, the Authority and LKAR agree that any and all Minimum Guaranteed Monthly Payments and other disbursements or advances which were to be made to the Tribe under Sections 2.3, 2.5, 6.3, 6.5, or 6.7 of the October 13, 2003 Amended Memorandum Agreement shall hereafter be made by LKAR to the Authority rather than the Tribe.

3. New definitions shall be added to Article 1 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

"Affiliate" means, with respect to any specified Person, any other Person that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with the specified Person. For the purposes of this definition, "control" (including the terms controlling, controlled by, or under common control with) means the possession, direct or indirect, of the power to direct or cause the direction of the management

Third Amendment to

and policies of a person, whether through the ownership of voting securities, partnership or member interests, by contract or otherwise.

"Assignment and Assumption Agreement" means the Assignment and Assumption Agreement dated as of April 20, 2007 between the Tribe, LKAR and the Authority, as amended from time to time, whereby the Tribe has duly assigned and delegated to the Authority effective April 20, 2007, all of its right, title, interest and obligations in and to this Amended Memorandum Agreement and other Transaction Documents, the Authority has accepted the assignment and assumption of each of the same, and LKAR has consented to such assignment and assumption.

"Authority" means the Shingle Springs Tribal Gaming Authority, the tribal agency established pursuant to tribal Resolution 2004-18, as amended from time to time, as an instrumentality of the Tribe to own and operate the Project and Enterprise.

"Person" means any person or entity, whether an individual, trustee, corporation, general partnership, limited partnership, limited liability company, limited liability partnership, joint stock company, trust, estate, unincorporated organization, business association, Indian tribe, commission, instrumentality, firm, joint venture, Governmental Authority, or otherwise.

"Tribal Agreement" means that certain Tribal Agreement dated as of April 20, 2007, executed by and between the Tribe and LKAR, as amended from time to time.

 Section 1.12 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

1.12 "Costs of Construction" means all costs incurred by the Tribe or LKAR pursuant to this Amended Memorandum Agreement in the aggregate to develop, construct and complete the Facility (and the commercial access road/interchange providing access to the Gaming Site), including, without limitation,

64

but excluding Initial Costs of Operation. The final amount of costs to be included in the Costs of Construction shall be determined by mutual agreement of the parties and shall be documented in the Approved Construction Budget.

 Section 1.13 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows: 1.13 "Costs of Gaming Operations" means all expenses for the operation of the Enterprise's Class II Gaming and III Gaming activities pursuant to Generally Accepted Accounting Principles ("GAAP"), including but not limited to the following:

64

6. Section 1.14 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

Third Amendment to October 13, 2003 Amended Memorandum Agreement 05/17/07 execution version 1.14 "Costs of Incidental Operations" means all expenses pursuant to Generally Accepted Accounting Principles incurred in

comprising the Facility in which the Enterprise conducts neither Class II Gaming nor Class III Gaming, including, without limitation:

7. Section 1.17 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

1.17 "Development Agreement" shall mean those provisions of this Amended Memorandum Agreement that deal with the development and construction of the Facility, as the same

Third Amendment to

may be amended or modified. The Development Agreement shall continue until the earlier of either the Commencement Date or provided however, that

8. Section 1.19 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

1.19 "Dominion Account Agreement" shall mean any agreement granting and governing the security interest in favor of LKAR in the Dominion Account, together with all amendments, substitutions and renewals thereof.

9. Section 1.21 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

1.21 "Enterprise" means the business of the Authority to be commonly known as the Foothill Oaks Casino which will conduct Class II Gaming and Class III Gaming at the Facility, and which shall include any other lawful commercial activity allowed in or near the Facility including, but not limited to,

10. Section 1.24 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

1.24 "Facility Loan" means, collectively, the loans, notes, bonds, equipment leases and/or other debt obligations arranged by Developer for the Authority, as borrower (excluding the loans, advances or other indebtedness evidenced by the Interim Promissory Note, the Operating Note or the Land Acquisition Note), in an aggregate principal amount not to exceed for financing Initial Costs of Operation

and Costs of Construction (including without limitation the Furnishings and Equipment), which Facility Loan shall be further evidenced by the Facility Note and other loan documentation as further defined herein.

 Section 1.25 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

1.25 "Facility Note" means the promissory notes, bonds and other documentation evidencing the Facility Loan in a form or forms to be agreed to by the parties to the Facility Loan, together with all amendments, substitutions and renewals thereof.

Third Amendment to October 13, 2003 Amended Memorandum Agreement 05/17/07 execution version

Sugar Administration and a

12. Section 1.33 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

1.33 "Gross Incidental Revenues" means the Enterprise's total receipts from

64

64

comprising the Facility in which the Enterprise conducts neither Class II Gaming nor Class III Gaming, including without limitation any sales, hotel, use or other pass-through taxes collected on Project patrons which are to be consistent with those collected from patrons of other similar businesses in El Dorado County (but excluding any insurance proceeds received other than business interruption insurance proceeds and insurance proceeds received to reimburse the Enterprise for any claims included, or to be included, as Costs of Incidental Operations).

13. Section 1.42 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

1.42 "Limited Recourse" means

 Section 1.48 of the October 13, 2003 Amended Memorandum Agreement, is amended to state as follows: Third Amendment to

1.48 "Net Incidental Revenues" means Gross Incidental Revenues less Costs of Incidental Operations (excluding as set forth in Section 6.5 herein).

64

64

15. Section 1.59 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

1.59 "Resolution of Limited Waiver" refers to the limited waivers of sovereign immunity adopted by the Tribe and the Authority as <u>Exhibit B</u>, <u>Exhibit C</u>, <u>Exhibit D</u> and <u>Exhibit E</u>, evidencing all approvals required pursuant to the Tribe's or the Authority's governing documents and applicable law.

16. Section 2.3(b)(i) of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

(b)(i) The total amount of funds advanced directly from LKAR pursuant to Section 2.3(a)(i) shall equal

Principal and accrued interest due under the Transition Loan shall be paid as provided under Section 2.3(c) below.

17. Section 2.3(a)(i) of the October 13, 2003 Amended Memorandum Agreement is amended to state as follows:

(a)(i) LKAR agrees to make the following pre-construction advances to the Tribe: (1)

Third Amendment to

October 13, 2003 Amended Memorandum Agreement 05/17/07 execution version 10

The Tribe and LKAR agree that all sums previously advanced to the Tribe by KARSS under the superseded Development and Management Agreements, and by LKAR under the superseded May 5, 2000 Memorandum of Agreement (excluding advances for acquisition of the Acquired Tribal Lands), shall constitute advances by LKAR to the Tribe hereunder, be credited to LKAR's obligations hereunder, and shall be subject to the terms of the Transition Loan herein. 64

64

14

18. Section 2.3(b)(ii) of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

(b)(ii) The total amount of funds advanced directly from LKAR pursuant to Section 2.3(a)(ii) shall equal

Principal and accrued interest due under the Land Acquisition Loan shall be paid as provided under Section 2.3(c) below.

19. The first sentence of Section 2.3(c) of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

(c) The Land Acquisition Loan and the Transition Loan shall each (i) be subject to all the terms and conditions of this Amended Memorandum Agreement;

Each payment under the Transition Loan shall be applied first to the payment of

Third Amendment to

accrued interest on the principal balance thereunder, and the remainder shall be applied to principal.

20. The first Sentence of Section 2.4(c) of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

(c) The Construction Contract shall contain such provisions for the protection of the Authority and LKAR as the parties deem appropriate, and shall provide that construction of the Facility shall commence within such time as the parties agree after receiving NIGC Approval, and any necessary Tribal Gaming Commission approvals (not to be unreasonably withheld), following and subject to the granting of all approvals necessary to commence construction and obtaining the Facility Loan; and shall also provide that the General Contractor, and all its subcontractors, shall exert its best efforts to complete construction within such time as the Authority and LKAR agree.

21. Section 2.4(c) of the October 13, 2003 Amended Memorandum Agreement is amended by adding the following sentence to the section:

Notwithstanding the foregoing, the terms of the Construction Contract may be modified as mutually agreed by the Authority and LKAR.

22. Section 2.4(g) of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

(g) LKAR, with the assistance of the Architect, shall submit to the Authority, for its review and approval, the specifications for Furnishings and Equipment. Thereafter, LKAR shall select and procure vendors for purchase by the Authority of Furnishings and Equipment required to operate the Enterprise in conformity with such specifications. The cost of Furnishings and Equipment shall be financed through the Facility Loan. Alternatively, in the sole discretion of the Authority, LKAR may arrange for the procurement of Furnishings and Equipment on lease terms consistent with the terms provided as to the Facility Loan.

23. The first sentence of Section 2.5(c) of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

(c) The Costs of Construction (including Furnishings and Equipment) and Initial Costs of Operation shall equal

64

Third Amendment to

October 13, 2003 Amended Memorandum Agreement 05/17/07 execution version 12

24. Section 2.5(d) of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

(d) The Facility Loan, if made directly by LKAR, shall (i) be subject to all the terms and conditions of this Amended Memorandum Agreement; (ii) be evidenced by the Facility Note (in a form consistent with the terms of this Development Agreement if LKAR makes the advances); and (iii)

64

The Authority agrees to enter into a limited, transactional waiver of sovereign immunity and consent to jurisdiction and arbitration with respect to the Facility Loan as to LKAR, if LKAR directly makes the Facility Loan, as provided in the Resolution of Limited Waiver.

25. Section 2.5(e) of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

(e) The Authority shall retain the right to prepay the Facility Loan, if made directly by LKAR, in whole or in part, without imposition of any prepayment penalty.

 Section 2.5(f) of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

Third Amendment to

(f) It is the understanding of the parties that the Facility Loan will be the sole responsibility of the Authority and, if made directly by LKAR, will be a Limited Recourse obligation of the Authority and will not be subject to any other guarantee or obligation on the part of the Authority except the security interests and liens described herein.

27. Section 2.11 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

2.11 Limited Waiver of Sovereign Immunity. By this Amended Memorandum Agreement, neither the Tribe nor the Authority waives, limits, or modifies its sovereign immunity from unconsented suit except as provided in the Resolution of Limited Waiver. The Authority understands that the agreement of the Tribe and the Authority to adopt an enforceable Resolution of Limited Waiver is a material inducement to the LKAR's execution of this Amended Memorandum Agreement and is a condition precedent to any of the respective obligations of the parties under this Amended Memorandum Agreement. The Authority further agrees that it will not amend or alter or in any way lessen the rights of the LKAR as set forth in the Resolution of Limited Waiver, which is attached hereto and incorporated here by reference. This Section 2.11 shall survive the termination of this Amended Memorandum Agreement, regardless of the reason for the termination.

28. Section 2.14 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

2.14 <u>Term of Development Agreement</u>. Unless sooner terminated as provided in this Amended Memorandum Agreement, the term of the Development Agreement shall run until the earlier of either (i) the Commencement Date; or (ii) provided however, that the Notes and Security Provisions shall continue until all amounts owing to LKAR with respect thereto have been paid in full.

64

64

29. Section 5.6 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

5.6 <u>Property Insurance</u>. LKAR, acting as agent for the Authority, shall procure replacement value all-risk casualty and extended hazard insurance in appropriate coverage amounts which shall insure the Facility and any fixtures, improvements and contents located therein against lost or damage by fire, theft and vandalism. Such casualty insurance policy or policies shall name the Authority and LKAR, and the Facility Loan lenders as insureds. Unless otherwise required by the terms of the Facility Loan, all such casualty insurance proceeds shall be applied to the immediate replacement of the applicable Facility part or fixture, improvements or contents therein unless the parties agree otherwise.

Third Amendment to

30. Section 5.10 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

5.10 <u>Insurance Proceeds</u>. Subject to the terms of Sections 7.4 and 7.6 hereof, and unless otherwise required by the terms of the Facility Loan, any insurance proceeds received with respect to the Enterprise, except as provided in Section 5.6 hereof, shall be deposited into the Dominion Account and disbursed in accordance with the same terms and provisions applicable to Gross Total Revenues, provided, however, that if there is any insurance recovery for a claim related to the operation of the Enterprise for which either the Authority or LKAR has previously paid from its own separate funds, then, to the extent of amounts paid by either of such parties, the insurance proceeds will be paid over to them and the balance shall be deposited into the Dominion Account as above.

31. Section 7.1(a) of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

Subject to the provisions of Section 9.2, either party may terminate this (a) Amended Memorandum Agreement if the other party (or its Affiliates identified below) commits or allows to be committed a Material Breach (as hereinafter defined) of this Amended Memorandum Agreement and fails to cure or to take steps to substantially cure such breach within thirty (30) calendar days after receipt of a written notice from the non-breaching party identifying the nature of the Material Breach in specific detail and its intention to terminate this Amended Memorandum Agreement. Termination is not an exclusive remedy for breach, and the non-breaching party shall be entitled to other rights and remedies as may be available. For purposes of this Amended Memorandum Agreement, a "Material Breach" is any of the following circumstances: (i) failure of LKAR to provide the Authority with the Minimum Guaranteed Monthly Payments pursuant to Section 6.3, (ii) material failure of either party to perform in accordance with this Amended Memorandum Agreement for reasons not excused under Section 10.6 (Force Majeure), (iii) if any of LKAR's employees commits theft, embezzlement or crime of moral turpitude and if, after knowledge of such act or, if disputed, after determination by arbitration under Article 11, LKAR does not remove such employee from connection with Class II Gaming or Class III Gaming operations of the Enterprise within thirty (30) days after receipt of written notice, (iv) default under the Facility Note, the Land Acquisition Note, the Interim Promissory Note, the Operating Note, any other Transaction Document or any document or agreement related thereto by the Authority, and any default or Third Amendment to

misrepresentation by the Tribe under the Tribal Agreement or any other document or agreement executed by the Tribe in favor of LKAR or its Affiliates; or (v) any representation or warranty made pursuant to Section 10.11 or 10.12 proves to be knowingly false or erroneous in any material way when made. Any final notice of termination hereunder shall be in writing detailing the reason the party considers the Material Breach not to be cured and must be delivered to the other party before such termination becomes effective

32. Section 6.8 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

6.8 <u>Development and Construction Cost Repayment</u>. The maximum dollar amount for repayment of development and construction costs for the Facility and Enterprise shall be Subject to any applicable Legal Requirements, the parties may increase the maximum repayment amount by mutual written agreement.

64

33. Section 10.3 of the October 13, 2003 Amended Memorandum Agreement is amended to substitute certain of the notice address, as follows:

(a) In place of the Notice to the Tribe, the following notice:

Shingle Springs Tribal Gaming Authority P.O. Box 1660 El Dorado, CA 95623-1660 Attn: Chairman

(b) In place of the notice to Anthony Cohen, Esq., the following notice:

Barbara E. Karshmer Karshmer & Associates 2150 Shattuck Avenue, Suite 725. Berkeley, CA 94704 Fax: 510-841-6167

(c) Adding the following notice after notice to Lakes LKAR-Shingle-Springs, LLC:

With a Copy to: Damon E. Schramm General Counsel Lakes Entertainment, Inc.

Third Amendment to

130 Cheshire Lane, Suite 101 Minnetonka, Minnesota 55305 Fax: (952) 449-7068

(d) In place of the notice to Brian Klein, Esq., the following notice:

Daniel R. Tenenbaum, Esq. Gray Plant Mooty 500 IDS Center 80 South Eighth Street Minneapolis, MN 55402-3796 (612) 632-3050 Fax: (612) 632-4050

34. Section 10.10 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

10.10 <u>Sovereign Immunity</u>. Except as described in the Resolution of Limited Waiver, each attached hereto as exhibits which are incorporated herein by reference, nothing in this Amended Memorandum Agreement, as amended, shall be deemed or construed to constitute a waiver of sovereign immunity of the Authority and the only applicable waivers of sovereign immunity shall be those expressly provided and executed by the Authority's duly authorized representative and substantially conforming to the form as approved by the parties. The parties agree that they will not amend or alter the Resolution of Limited Waiver in any way which will lessen the rights of any party as set forth in the Resolution of Limited Waiver. This Section 10.10 shall survive termination of this Amended Memorandum Agreement, regardless of the reason for the termination.

35. Section 10.12 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

10.12 <u>Representations and Warranties of the Authority</u>. The Authority hereby represents and warrants as follows:

(a) The Tribe is a duly organized Indian tribe under the laws of the Tribe and the United States, and the Authority is duly organized under the laws of the Tribe.

(b) The Authority has full legal right, power and authority under the laws for the Tribe and has taken all official Management Board actions necessary (i) to assume and enter into this Amended Memorandum Agreement and related Transaction Documents (as defined below) and authorize the Authority to assume, execute and deliver this Amended Third Amendment to
17

Memorandum Agreement, including any amendment thereto, the Operating Note, Dominion Account Agreement, Security Agreement, the Facility Note and other Facility Loan documentation (if the Facility Loan is made directly by LKAR), Land Acquisition Note and Interim Promissory Note, and any and all other documents and agreements executed by the Authority or the Tribe related thereto (collectively and as amended, renewed, or extended from time to time, the "Transaction Documents"), (ii) to perform its obligations hereunder and thereunder, and (iii) to consummate all other transactions contemplated by this Amended Memorandum Agreement and other Transaction Documents.

(c) This Amended Memorandum Agreement (and the other Transaction Documents), as amended, constitute valid and binding obligations, enforceable against the Authority in accordance with their terms, executed by the Authority or the Tribe in connection therewith, and when approved by the NIGC, will constitute, together with the other Transaction Documents, valid and binding obligations, enforceable against the Authority in accordance with their terms.

(d) Subject to the provisions of Section 10.11(d), the execution and delivery of this Amended Memorandum Agreement (and other Transaction Documents), as amended, the performance by the Authority of its obligations hereunder and thereunder, and the consummation by the Authority of the transactions contemplated hereby will not violate any contract or agreement to which the Tribe or the Authority is a party, law, regulation, rule or ordinance, or any order, judgment or decree of any federal, state, tribal or local court, or require any approval by Governmental Authorities beyond those contemplated herein.

(e) Neither LKAR, the Project, the Facility, the Enterprise nor the transaction(s) between the parties contemplated by this Amended Memorandum Agreement, the Operating Note, and any related security documents and instruments described herein as amended, are now, or at any time during the term of this Amended Memorandum Agreement will be, subject to any tribal tax of any sort that will put the Facility or the Enterprise, or any portion thereof, nor the transaction(s) between the parties contemplated by the Transaction Documents, at a competitive disadvantage with businesses in the same or like industries.

(f) The Tribe (and the Authority under delegated powers from the Tribe) is legally permitted to conduct Class II Gaming and Class III Gaming (pursuant to the Tribal-State Compact) activities in the State under all Legal Requirements, and the Gaming Site for the Project constitutes "Indian lands" upon which the Tribe may legally conduct gaming under IGRA.

(g) Neither the Tribe nor any of its Affiliates has enacted any law, ordinance, resolution, rule or regulation impairing the rights or obligations of the Authority or LKAR

Third Amendment to

under this Amended Memorandum Agreement or under any Transaction Documents, as amended, contemplated hereby.

(h) The Tribal Agreement constitutes a valid and binding obligation, enforceable against the Tribe in accordance with its terms.

36. Section 10.13 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

## 10.13 Governing Law.

(a) This Amended Memorandum Agreement and (except as the law of another State may be specifically chosen as the governing law therein) the other Transaction Documents have been negotiated, made and executed at the Authority's office located in the State of California and shall be construed in accordance with the applicable laws of the State of California, without regard to its conflict of laws provisions, and applicable Tribe and federal laws.

(b) Use of the laws of the State of California for the foregoing limited purpose of interpretation and construction is not intended to and shall not otherwise (i) incorporate substantive laws or regulations of the State of California, including usury laws or any present or future provisions of the State of California that would restrict the rate of interest upon any loan, advance or other financial accommodation contemplated by any Transaction Document, or (ii) grant any jurisdiction to the State of California or any political subdivision thereof over the Gaming Site or the Property.

(c) Any other provision of this Amended Memorandum Agreement or any other Transaction Document to the contrary notwithstanding, (i) in no event shall the rate of interest payable in connection with the indebtedness incurred pursuant to any Transaction Document or in connection with the transactions contemplated by any Transaction Document exceed the maximum nonusurious interest rate, if any, at any time or from time to time permitted to be contracted for, taken, reserved, charged or received under applicable law ("Maximum Rate"), and (ii) if at any time the rate of interest that would otherwise be payable on any such indebtedness would exceed any such Maximum Rate, then the rate of interest that would otherwise be contracted for, taken, reserved, charged or received under such indebtedness shall be reduced to such Maximum Rate.

37. Section 10.14 of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

Third Amendment to

10.14 Entire Agreement. This Amended Memorandum Agreement (including all its exhibits and related Transaction Documents, along with tribal Resolution 2004-18 approving the establishment of the Shingle Springs Tribal Gaming Authority and its related Ordinance(and any amendments thereto as of April 20, 2007), the June 16, 2004 Amendment to this Amended Memorandum Agreement, the June 16, 2004 Consent Agreement entered into by the parties, tribal Resolution 2004-23 concerning the June 16, 2004 Amendment and Consent Agreement, the January 23, 2007 Second Amendment to this Amended Memorandum Agreement, tribal Resolution 2007-04 concerning the January 23, 2007 Second Amendment, the April 20, 2007 Assignment and Assumption Agreement, the April 20, 2007 Tribal Agreement and tribal Resolution 2007-18 concerning the Tribal Agreement and the April 20, 2007 Assignment and Assumption Agreement, the May 17, 2007 Third Amendment to this Amended Memorandum Agreement, and Authority Resolution 2007-05 concerning the May 17, 2007 Third Amendment), represents the entire agreement between the parties and supersedes all prior agreements relating to the subject matter of Class II Gaming and Class III Gaming to be developed and conducted by the Authority at the Facility and operations of the Enterprise. The parties hereto intend that this Amended Memorandum Agreement (and related Transaction Documents) is to supersede and replace the May 5, 2000 Memorandum of Agreement and any other prior agreements between the Tribe and LKAR in connection with the Project.

38. Section 11.3(b) of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

(b) If the dispute is not resolved to the satisfaction of the parties within thirty (30) calendar days after the first meeting in Section 11.3(a) above, then any claim, controversy or dispute arising out of or relating to this Amended Memorandum Agreement, Facility Note, Land Acquisition Note, the Interim Promissory Note, the Operating Note, or any other Transaction Document, or any alleged default hereunder or breach of any provisions thereof, shall be submitted to binding arbitration in accordance with the Commercial Arbitration Rules of the American Arbitration Association in effect at the time of submission: except that: (a) the question whether or not a dispute is arbitrable under this Amended Memorandum Agreement or any other Transaction Document shall be a matter for binding arbitration by the arbitrators, such question shall not be determined by any court and, in determining any such question, all doubts shall be resolved in favor of arbitrability; and (b) discovery shall be permitted in accordance with the Federal Rules of Civil Procedure, subject to supervision as to scope and appropriateness by the arbitrators. Judgment on any arbitration award may be entered in any court having jurisdiction over the parties pursuant to the Resolution of Limited Waiver. The Authority, on behalf of itself and each of its Affiliates, agrees that any arbitration proceeding hereunder may be consolidated with any other arbitration proceeding that any of LKAR or its respective Affiliates may bring against the Authority, the Tribe or any of their respective Affiliates.

39. Section 11.3(f) of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

The arbitration award shall be in writing signed by each of the arbitrators, and (f) shall state the basis for the award. The arbitration award shall be set forth in reasonable detail as to its findings of fact and law, and basis of determination of award form and amount. In connection with any arbitration award, the arbitrators shall be empowered to take the actions and enforce the judicial remedies described in Paragraph 5 of the Resolution of Limited Waiver; provided however, that although the arbitrators may award damages in the event the Tribe, the Authority or the Tribal Gaming Commission do not comply with the award, the arbitrators may not require the Tribe, the Authority or the Tribal Gaming Commission to take or modify any governmental legislative decision or action which the arbitrators have determined has resulted in the dispute between the parties and is contrary to the parties' rights, liabilities or obligations under this Amended Memorandum Agreement, the Facility Note, if the Facility Loan is made directly by LKAR, the Land Acquisition Note, the Interim Promissory Note, the Operating Note, or any other Transaction Document, as amended, ("Specific Performance Restriction"). Provided further, that: (a) should the arbitrators determine that there has been an intentional bad faith violation of a party's rights under this Amended Memorandum Agreement or any other Transaction Document by the Tribe, the Authority or Tribal Gaming Commission, and if the Tribe, the Authority or the Tribal Gaming Commission do not reverse such intentional bad faith violation through governmental legislative decision or action within thirty (30) days after being notified by the arbitrators of such determination, then the arbitrators shall

64

40. The reference to "Depository" in Section 11.3(j) of the October 13, 2003 Amended Memorandum Agreement is amended to read "depository holding the Dominion Account".

41. The List of Exhibits page of the October 13, 2003 Amended Memorandum Agreement is amended to read as follows:

## LIST OF EXHIBITS

Exhibit A Legal Description of Gaming Site

Third Amendment to

October 13, 2003 Amended Memorandum Agreement 05/17/07 execution version 21

Exhibit B	Resolution 2003-12 concerning Limited Waiver of Immunity from Suit
Exhibit C	Resolution 2004-23 concerning Limited Waiver of Immunity from Suit
Exhibit D	Resolution 2007-04 concerning Limited Waiver of Immunity from Suit
– Exhibit E	Authority Resolution 2007-05 concerning Limited Waiver of Immunity from Suit

42. The Authority and LKAR agree that any dispute between them in connection with this Third Amendment shall be subject to the dispute resolution procedures and limited waiver of sovereign immunity contained in the October 13, 2003 Amended Memorandum Agreement (as heretofore and hereafter amended) and Authority Resolution 2007-05 concerning this Third Amendment, attached to the October 13, 2003 Amended Memorandum Agreement as Exhibit E.

43. The Authority and LKAR agree that capitalized terms used herein and not defined shall have the meanings given them in the October 13, 2003Amended Memorandum Agreement.

44. The Authority and LKAR agree that this Third Amendment shall be construed in accordance with and governed by the internal laws and decisions of the State of California, without giving effect to its choice of law principles; and may be executed in any number of counterparts and by facsimile, each of which shall be considered an original but together shall constitute one and the same instrument.

45. The Authority and LKAR agree that no modification, amendment or change to this Third Amendment shall be valid unless the same is in writing and signed by the party against which the enforcement of such modification, amendment or change is sought. The parties hereto specifically intend and agree that this Third Amendment is to supersede and replace in all respects the Third Amendment dated as of April 20, 2007.

46. Except as amended above, all other provisions of the October 13, 2003 Amended Memorandum Agreement shall remain in full force and effect as originally stated and are equally applicable hereto.

[Signature Page Follows]

Third Amendment to

IN WITNESS WHEREOF, the parties hereto have executed this May 17, 2007 Third Amendment to the October 13, 2003 Amended Memorandum Agreement.

Shingle Springs Tribal Gaming Authority

By: augon Its: Management Board Chairman

By: Its: Secvi

## Lakes KAR-Shingle Springs, LLC

By: \_\_\_\_\_\_ Its: \_\_\_\_\_

Approved pursuant to 25 U.S.C. §2711

National Indian Gaming Commission

By: \_\_\_\_\_ Print Name: Its: Chairman

Third Amendment to October 13, 2003 Amended Memorandum Agreement 05/17/07 execution version

23

IN WITNESS WHEREOF, the parties hereto have executed this May 17, 2007 Third Amendment to the October 13, 2003 Amended Memorandum Agreement.

Shingle Springs Tribal Gaming Authority Lakes KAR-Shingle Springs, LLC

By: c Fo RELIAST Its:

By: \_\_\_\_\_\_ Its: Management Board Chairman

By: \_\_\_\_\_\_ Its: \_\_\_\_\_\_

Approved pursuant to 25 U.S.C. §2711

National Indian Gaming Commission

By: \_\_\_\_\_ Print Name: Its: Chairman

Third Amendment to October 13, 2003 Amended Memorandum Agreement 05/17/07 execution version

23

Third Amendment dated May 17, 2007 To the October 13, 2003 Amended Memorandum Agreement

Approved pursuant to 25 U.S.C.§2711 National Indian Gaming Commission

Philip Hogen, Chairman

June 8, 2007



Shingle Springs Tribal Gaming Authority P.O. Box 1660; El Dorado, CA 95623-1660 (530) 676-8010; Fax (530) 676-8033

## GA RESOLUTION 2007 - 05

MAY 2 2 2007

<u>SUBJECT</u>: RESOLUTION OF THE SHINGLE SPRINGS TRIBAL GAMING AUTHORITY REGARDING LIMITED WAIVER OF SOVEREIGN IMMUNITY RELATING TO, AND APPROVAL OF, THE APRIL 20, 2007 THIRD AMENDMENT TO THE OCTOBER 13, 2003 AMENDED MEMORANDUM AGREEMENT, AS AMENDED, AND THE ASSIGNMENT AND ASSUMPTION AGREEMENT DATED AS OF APRIL 20, 2007 WITH LKAR IN CONNECTION WITH THE FOOTHILL OAKS GAMING FACILITY/ENTERPRISE

WHEREAS, the Shingle Springs Band of Miwok Indians ("Tribe") is a federally recognized Indian tribe eligible for the special programs and services provided by the United States to Indians because of their status as Indians and is recognized as possessing powers of self-government; and

WHEREAS, the Tribal Council of the Tribe is the duly recognized governing body of the Tribe and is a federally-recognized Indian tribal government; and

WHEREAS, in compliance with the Indian Gaming Regulatory Act of 1988 ("IGRA"), P.L. 100-497, 25 U.S.C. 2701 <u>et seq.</u> as it may from time to time be amended, the Tribal Council of the Tribe has enacted a tribal ordinance regulating the operation of gaming activities on Tribal Lands (hereinafter referred to as the "Tribal Gaming Ordinance"), creating the Shingle Springs Tribal Gaming Commission, and authorizing Class II Gaming and Class III Gaming on its Indian lands subject to the provisions of the Tribal Gaming Ordinance and a Tribal-State Compact; and

WHEREAS, the Tribe is committed to the use of gaming activities to provide employment and improve the social, economic, education, and health needs of its members; to increase the revenues of the Tribe; and to enhance the Tribe's economic self-sufficiency and self-determination; and

WHEREAS, because the Tribe lacks the resources to develop and operate a gaming facility and enterprise on its own, it desired to retain the services of a developer and manager with knowledge and experience in the gaming industry to secure financing, develop, manage and operate a Class II Gaming and Class III Gaming facility and related resort facilities located on its Indian lands in accordance with the Indian Gaming Regulatory Act of 1988, as amended; and

WHEREAS, to assist with the financing, development, management and operations of its planned gaming facility and enterprise to be known as the Foothill Oaks Casino (the "Project"), the Tribe and LKAR entered into the First Amended and Restated Memorandum of Agreement Regarding Gaming Development and Management Agreement (as amended by an Amendment dated June 16, 2004 and approved by the Chairman of the National Indian Gaming Commission ("NIGC") on July 19, 2004 and as further amended by a Second Amendment dated January 23, 2007 approved by the NIGC on March 20, 2007, hereinafter collectively referred to as the "October 13, 2003 Amended Memorandum Agreement") and other Transaction Documents in connection with the Project; and

WHEREAS, after originally signing the October 13, 2003 Amended Memorandum Agreement, the Tribe adopted tribal Resolution 2004-18 establishing the Shingle Springs Tribal Gaming Authority ("Authority") and related Ordinance whereby the Tribe delegated to the Authority, subject to certain limitations, all governmental and proprietary powers and rights of the Tribe over the development, construction, operation, promotion and financing of the Project; and, through a Consent Agreement dated June 16, 2004, LKAR consented to the Tribe's assignment to and the Authority's assumption of all of the Tribe's rights and responsibilities under the October 13, 2003 Amended Memorandum Agreement and the other Transaction Documents; and

WHEREAS, due to indefinite delays in the financing and construction of the Project arising from litigation commenced by El Dorado County in connection with the dedicated interchange from US 50 providing direct access to the Shingle Springs Rancheria, the Tribe amended its Ordinance 2004-18 to provide that its delegation to the Authority of certain governmental and proprietary powers and rights of the Tribe over the development, construction, operation, promotion and financing of the Project would be suspended until such date that the Tribal Council activated the Authority by formally appointing the Management Board of the Authority; and to provide further that until such time the Tribe would retain and reassume all of the Tribe's rights and responsibilities under the October 13, 2003 Amended Memorandum Agreement (and other Transaction Documents); and

WHEREAS, on April 20, 2007, the Tribal Council activated the Authority by appointing the Management Board of the Authority; and the governmental and proprietary powers and rights of the Tribe over the development, construction, operation, promotion and financing of the Foothill Oaks Casino and the Enterprise, as described in Section 1 of Ordinance 2004-18, thereby became the delegated responsibility of the Authority; and

WHEREAS, the Management Board of the Authority is the duly recognized governing body of the Authority; which possesses the necessary powers to make binding and enforceable agreements with third parties in connection with the Project; and

WHEREAS, pursuant to the terms of the January 23, 2007 Second Amendment, the Tribe agreed that, upon the Authority becoming activated, the Tribe would cause the

Authority to (i) approve and execute an amendment to the October 13, 2003 Amended Memorandum Agreement (hereinafter referred to as the "Third Amendment"), (ii) enter into an agreement with LKAR reflecting the Tribe's assignment to and assumption by the Authority of the Tribe's rights and responsibilities under the October 13, 2003 Amended Memorandum Agreement and other Transaction Documents, in accordance with Section 1 of Ordinance 2004-18 (hereinafter referred to as the "Assignment and Assumption Agreement"); and (ii) adopt a resolution of limited waiver related to approval of this Third Amendment; and

WHEREAS, the Authority understands that its agreement to adopt an enforceable resolution of limited waiver is a material inducement to LKAR's execution of the Third Amendment and the Assignment and Assumption Agreement, and to LKAR continuing to make advances of sums under the Transition Loan, and is a condition precedent to any of the respective obligations of the parties under the October 13, 2003 Amended Memorandum Agreement; and

WHEREAS, the Authority believes that the development contemplated by the October 13, 2003 Amended Memorandum Agreement, is important to the governance and long-term economic development of the Tribe, and desires to fulfill its obligations under the terms of the October 13, 2003 Amended Memorandum Agreement, and under IGRA to ensure the success of the Project.

WHEREAS, in connection with the financing and construction of the Project and the US 50 dedicated interchange, the Authority and LKAR entered into the Assignment and Assumption Agreement and a Third Amendment to the October 13, 2003 Amended Memorandum Agreement, each dated as of April 20, 2007; since signing the Third Amendment dated as of April 20, 2007, the Authority and LKAR desire to make certain technical revisions to that agreement before approval by the NIGC and such necessary modifications are incorporated into a revised Third Amendment dated May 17, 2007 which the parties intend to supersede and replace in all respects the Third Amendment dated as of April 20, 2007.

## NOW THEREFORE, BE IT RESOLVED,

1. That the capitalized terms used herein and not defined shall have the meanings given them in the October 13, 2003 Amended Memorandum Agreement, as amended.

2. That the Authority hereby approves and ratifies, and subject to the terms and provisions hereof expressly agrees to irrevocably waive its sovereign immunity from suit in connection with, the following documents relating to the Project: (i) the Third Amendment to the October 13, 2003 Amended Memorandum Agreement dated as of May 17, 2007 (05/17/07 execution version), (ii) the Assignment and Assumption Agreement dated as of April 20, 2007 (04/20/07 execution version), and (iii) confirms its express approval and ratification of the Tribe's waiver of sovereign immunity in connection with the October 13, 2003 Amended Memorandum Agreement and the

other Immunity Agreements as described and defined in tribal Resolution 2003-12, tribal Resolution 2004-23, and tribal Resolution 2007-04 (collectively, the "Amended Immunity Agreements"). The Authority's waiver of sovereign immunity is for the sole and limited purpose of permitting and compelling arbitration as to any issue or dispute under the Amended Immunity Agreements and enforcing any arbitration award or judgment arising thereunder, as described herein and as further limited by the restrictions described herein.

That the Authority consents to be sued in the United States District Court 3 for the district in which the Facility is located, the United States Court of Appeals having jurisdiction over the applicable District Court, and the United States Supreme Court, solely for the purpose of compelling arbitration or enforcing any arbitration award, orders or judgment arising out of the Amended Immunity Agreements. If the United States District Court lacks jurisdiction, the Authority consents to be sued in a court of competent jurisdiction for the same limited purpose. Subject to the provisions of this Resolution, the Authority waives any requirement of exhaustion of tribal remedies or abstention doctrine and any other law, rule, regulation, resolution or interpretation that might otherwise require, as a matter of law or comity, that any suit or proceeding be heard first in a tribal forum and expressly authorizes any governmental authorities having the right and duty under applicable law to take any action authorized or ordered by a court having jurisdiction hereunder, provided that at all times the liability of the Authority under any judgment shall be always Limited Recourse as defined herein.

4 That the Authority agrees and consents that all disputes, controversies or claims arising out of or relating to the Amended Immunity Agreements, or the parties' obligations and rights thereunder, shall be settled by binding arbitration in accordance with the terms set forth in the Amended Immunity Agreements, including without limitation, that such arbitration shall be governed by the Commercial Arbitration Rules of the American Arbitration Association in effect on the date the demand for arbitration is made, and the Federal Arbitration Act; except that: (a) the question whether or not a dispute is arbitrable under the Amended Immunity Agreements shall be a matter for binding arbitration by the arbitrators, such question shall not be determined by any court and, in determining any such arbitration question, all doubts shall be resolved in favor of arbitrability; and (b) discovery shall be permitted in accordance with the Federal Rules of Civil Procedure, subject to supervision as to scope and appropriateness by the arbitrators. The Authority further agrees that any arbitration proceeding held in connection with the Amended Immunity Agreements may be consolidated with any other arbitration proceeding involving LKAR or its Affiliates and any of the Authority's Affiliates.

5. That the Authority's waiver of immunity from suit is specifically limited to the following actions and judicial remedies:

(a) <u>Damages.</u> The enforcement of an award of compensatory damages by arbitration; provided that the award of any arbitrator and/or court must be Limited Single Springs Authority Resolution 4
 Third Amendment 05/17/07 execution version

Recourse, and no arbitrator or court shall have authority or jurisdiction to order execution against any assets or revenues of the Authority except (i) undistributed and future Net Total Revenues of the Enterprise or any other Tribal Gaming Enterprise; (ii) any Furnishings and Equipment purchased with Facility Loan or Transition Loan proceeds or other purchase money agreements; (iii) undistributed and future Gross Total Revenues of the Enterprise (as contemplated by the Dominion Account Agreement); and (iv) any Acquired Tribal Lands prior to their transfer into trust. "Limited Recourse" means that repayment of any advances made under the Facility Loan, if made directly by LKAR, the Land Acquisition Loan and Transition Loan, and all liabilities of the Authority related to the Amended Immunity Agreements and their applicable documentation, the Facility, or the Enterprise contemplated by the October 13, 2003 Amended Memorandum Agreement, and any related awards, judgments or decrees, shall be payable solely out of undistributed and future Net Total Revenues of the Enterprise or any other Tribal Gaming Enterprise, and shall be a limited recourse obligation of the Authority, with no recourse to Authority assets other than such undistributed and future Net Total Revenues and the following: (i) any Furnishings and Equipment purchased with Facility Loan or Transition Loan proceeds or other purchase money agreements pursuant to the Security Agreement; (ii) the undistributed and future Gross Total Revenues of the Enterprise or any other Tribal Gaming Enterprise pursuant to the Dominion Account Agreement; and (iii) any Acquired Tribal Lands prior to their transfer into trust. In no event shall LKAR, or its Affiliates, have recourse to (a) the physical property of the Facility (other than Furnishings and Equipment subject to the security interest securing the Facility Loan, if made directly by LKAR, the Transition Loan or other purchase money agreements); (b) Net Total Revenue distributions already made to the Authority or the Tribe in accordance with the October 13, 2003 Amended Memorandum Agreement and/or the Dominion Account Agreement; (c) assets of the Authority or the Tribe purchased with its Net Total Revenue distributions; or (d) any other asset of the Authority or the Tribe (other than the mortgage or deed of trust on the Acquired Tribal Lands prior to its transfer into trust, and such undistributed and future Net Total Revenues of the Enterprise or any other Tribal Gaming Enterprise). In no instance shall any enforcement of any kind whatsoever be allowed against any assets of the Authority other than the limited assets of the Authority specified herein.

(b) <u>Consents and Approvals.</u> The enforcement of a determination by an arbitrator that the Authority's consent or approval has been unreasonably withheld contrary to the terms of the Amended Immunity Agreements.

(c) <u>Injunctive Relief and Specific Performance.</u> The enforcement of a determination by an arbitrator that prohibits the Authority from taking any action that would prevent the other party from performing its obligations pursuant to the terms of the Amended Immunity Agreements, or that requires the Authority to specifically perform any obligation under the Amended Immunity Agreements; provided, however, that any injunction against the Authority shall be Limited Recourse; shall not mandate, preclude or affect payment of any funds of the Authority other than undistributed and future Net Total Revenues of the Enterprise or any other Tribal Single Springs Authority Resolution 5

Gaming Enterprise; and shall not relate to any asset of the Authority other than as described above under Paragraph 5(a). The Authority specifically acknowledges and agrees that in connection with any arbitration award, order or judgment, the arbitrators shall be empowered to take the actions and enforce the judicial remedies described in this Paragraph 5; provided however, that although the arbitrators may award damages in the event the Tribe, the Authority or the Tribal Gaming Commission do not comply with the award, the arbitrators may not require the Tribe, the Authority or Tribal Gaming Commission to take or modify any governmental legislative decision or action which the arbitrators have determined has resulted in the dispute between the parties and is contrary to a party's rights, liabilities or obligations under the October 13. 2003 Amended Memorandum Agreement, the Facility Note (if the Facility Loan is made directly by LKAR), the Land Acquisition Note, the Interim Promissory Note, the Operating Note, or any other Transaction Document ("Specific Performance Restriction"). Provided further, that: (a) should the arbitrators determine that there has been an intentional bad faith violation of a party's rights under any Amended Immunity Agreement by the Tribe, the Authority or the Tribal Gaming Commission, and if the Tribe, the Authority or the Tribal Gaming Commission do not reverse such intentional bad faith violation through governmental legislative decision or action within thirty (30) days after the being notified by the arbitrators of such determination. then the arbitrators shall

(d) <u>Action to Compel Arbitration and for Enforcement.</u> An action to compel arbitration pursuant to the applicable provisions in the Amended Immunity Agreements; or, to the extent not previously set forth above herein, an action seeking enforcement of the Amended Immunity Agreements, and to the extent applicable, foreclosure of any of the Amended Immunity Agreements.

6. That the Authority hereby appoints the Chairman and Secretary of its Management Board as its agents for service of process under or relating to the Amended Immunity Agreements, and agrees that service by hand or by certified mail, return receipt requested, shall be effective as service upon the Authority for all purposes under or relating to the Amended Immunity Agreements if served upon such agents.

7. That the Authority shall execute any documents and do any other undertaking as shall from time to time be reasonably required to carry out and make effective the terms and provisions of the Amended Immunity Agreements; and that the Single Springs Authority Resolution 6 Third Amendment 05/17/07 execution version Chairman and one other official of the Management Board are hereby authorized to execute the Amended Immunity Agreements described herein, and the Chairman is authorized to deliver the Amended Immunity Agreements and any related documents as necessary; and the Management Board approves, adopts and ratifies as properly executed the October 13, 2003 Amended Memorandum Agreement and related Transaction Documents of same date, as amended.

8. That the Authority hereby authorizes: (a) the submission of the Amended Immunity Agreements to the NIGC for review and approval, as necessary under federal law; (b) the Chairman of the Management Board to provide to the NIGC on behalf of the Authority any documentation requested by the NIGC in conjunction with its process of reviewing and approving, as necessary, the Amended Immunity Agreements, and any other information needed to expedite review and approval, and (c) the Chairman of the Management Board to act on behalf of the Authority concerning the submission to the NIGC of the Amended Immunity Agreements.

8. That this Resolution of Limited Waiver shall take effect immediately and shall supersede and replace the Authority Resolution 2007-01 adopted by the Authority on May 11, 2007 in connection with the same matter, which resolution is specifically revoked.

9. That the Authority agrees it will take all lawful steps necessary to preserve the effective terms of this Resolution of Limited Waiver in the event of future changes in its legal status or governance.

10. That the Authority shall not adopt any resolution or take any other action that has the effect of impairing any contractual rights of LKAR under any of the Amended Immunity Agreements, and any such resolution or action to the contrary shall be void and of no effect; and the Authority agrees and affirms that upon execution and delivery of any Amended Immunity Agreements as herein authorized, the document shall become a valid and binding obligation of the Authority, enforceable in accordance with its terms.

## CERTIFICATION

The foregoing resolution was enacted by the Management Board of the Shingle Springs Tribal Gaming Authority on the <u>18</u> day of <u>1000</u> Authority a vote of <u>4</u> In Favor, <u>6</u> Opposed and <u>6</u> Abstaining; at a duly called meeting at which a quorum of the Management Board was present.

Chairman

518107 Attest: Secretary